CMS Manual System

Pub 100-1, Medicare General Information, Eligibility, and Entitlement

Department of Health Human Services (DHHS)

Centers for Medicare & Medicaid Services (CMS)

Transmittal 1 Date: SEPTEMBER 11, 2002

CHAPTERS	REVISED SECTIONS	NEW SECTIONS	DELETED SECTIONS
1		10 - 70	
2		10 - 60	
3		10 - 40	
4		10 - 80	
5		10 - 90.4	
6		10 - 180	
Exhibits		A - C	

CLARIFICATION/MANUALIZATION--EFFECTIVE DATE: Not Applicable

This is the initial issuance of the CMS Manual System, Publication 100-1, which replaces current instructions in nine manuals about general provisions of the Medicare program. Prior to this revision these instructions were found in Chapter 1 of the various Medicare provider manuals and in Part 3, Chapter I, of the Medicare Carriers Manual and the Medicare Intermediary Manual.

Related instructions in those manuals are being retired in separate revisions to each manual. A crosswalk from this publication to the respective sections of those manuals begins on page 2 of this transmittal.

These instructions should be implemented within your current operating budget.

NOTE: Normally red italicized font identifies new material. However, because this release is a new manual, normal text font is used for the initial release.

Crosswalk to Deleted Material In Other Manuals

New	New	МІМ	МСМ	ННА	НО	HSP	ОРТ	RDF	RHC	SNF
Chapter	Section			Pub 11	Pub 10	Pub 21	Pub09	Pub 29	Pub 27	Pub 12
Chap 1	10			100, 155-165	100, 155-165	100, 144- 144.3	100	100	100, 104	100, 155- 165
Chap 1	10.1			155-155.1	155-155.2	144-144.3	104	102	104	155-155.1
Chap 1	10.2			155.2	155.3	144.3			104	155.2
Chap 1	10.3			160-160.1	160-160.1		106	104	106	160-160.1
Chap 1	10.4			160.2	160.2					160.2
Chap 1	20			100	100	100	100	100	100	100
Chap 1	20.1			102	102	102	102		102	102
Chap 1	20.2			104	104	104	116	204	118	104
Chap 1	20.3			106	106	106		128-128.3	120	106
Chap 1	30, 30.1			108	108	108	108- 108.2	118	108, 108.1, 108.2	108
Chap 1	30.2			108	108	108.2		118	108.3	108.2
Chap 1	30.3			112	112	110	114	120	108.8	112
Chap 1	40			120	120	120	110	122	108.5	120
Chap 1	40.1			140	140	140	122		108.5	140
Chap 1	50			125	125	125	112	122	108.6	125
Chap 1	60				195				108.7	127
Chap 1	60.1				195.1					
Chap 1	60.2				195.2					
Chap 1	60.3				195.2,					
Chap 1	60.4				195.3					
Chap 1	60.5				195.4					
Chap 1	70			146	146					146
Chap 2	10	3001	1001-20							
Chap 2	10.1			155, 165-168	155, 165-8	144-148		105, 102,	104	155,
										165-168
Chap 2	10.2			165	165			105.1		165
Chap 2	10.3		1015	166	166			105.4		166
Chap 2	10.4		1020	167	167			105.3, 200		167

New Chapter	New Section	MIM	мсм	ННА	но	HSP	ОРТ	RDF	RHC	SNF
Chapter	Section			Pub 11	Pub 10	Pub 21	Pub09	Pub 29	Pub 27	Pub 12
Chap 2	20		1008- 1008.2	168, 168.1	168, 168.1			104		168
Chap 2	30	3001.4	1008.3							
Chap 2	40	3003	1050	160, 175, 180	160, 175			106	106	160, 175,
Chap 2	40.1	3003.1	1050					106.1		
Chap 2	40.2	3003.2						106.2		
Chap 2	40.3	3003.3						106.3		
Chap 2	40.4		1060					106.4		
Chap 2	40.5		1008.3, 1050.3	180	180			106.5		180
Chap 2	40.6	3003.6						106.6		
Chap 2	40.7	3003.7						106.7		
Chap 2	40.8	3004		185	185					
Chap 2	50	3501	3200		304	301		300 - 304.3	600-604.3	
Chap 2	60	Not in cu	ırrent manua	ls			•			
Chap 3	10				220			108, 110,		249
Chap 3	10.1	3201			220			108		
Chap 3	10.2	3203			225			110		246
Chap 3	10.3	3206			226					247
Chap 3	10.4	3035			215					240
Chap 3	20 20.4	3185- 3185.1	2450- 2462	160, 160.1, 160.3, 160.5	160, 160.1, 160.3, 160.5	410		112, 114, 306, 308	606-607	160, 160.1, 160.3, 160.5,
Chap 3	20.5- 20.5.4	3235			249, 222, 220-227.1			116		210210
Chap 3	30-30.3		2470, 2480							
Chap 3	40		2480							
Chap 4	10-60	3315- 3339		224, 240	273-282					220

New	New	МІМ	мсм	нна	но	HSP	ОРТ	RDF	RHC	SNF
Chapter	Section			Pub 11	Pub 10	Pub 21	Pub09	Pub 29	Pub 27	Pub 12
Chap 5	10- 10.1.8	3005		130	130	130	120		122	130,205
Chap 5	10.1.9			132.1	132.1					
Chap 5	10.1.10				IM 103					
Chap 5	10.2	3006		134	134					134
Chap 5	10.3	3007			207					206
Chap 5	10.4	3005- 3005.3		132, 134	132	132				132, 142
Chap 5	10.5	3005.1								
Chap 5	10.6	3008- 3008.4		142	142	142	124			142
Chap 5	10.6.1	3008.1		142.1	142.1	142.1	124.1			142.1
Chap 5	10.6.2	3008.2		142.2	142.2	142.2	124.2			142.2
Chap 5	10.6.4	3008.4		142.3	142.3	142.4	124.3			142.4
Chap 5	10.6.5	3009		145	145	142.3			331D	145
Chap 5	20	3010		157, 157.1	200					203
Chap 5	20.1	3012			202					
Chap 5	20.2	3012			202.1					
Chap 5	20.3	3016			204					
Chap 5	20.4	3017			205					
Chap 5	20.5	3017.1			205.1					
Chap 5	20.6	3017.2			205.2					
Chap 5	20.7	3017.3			205.3					
Chap 5	30	3020		157.2						201
Chap 5	40	3023			206					202
Chap 5	50	3025		157, 200						
Chap 5	60	3027				144.4				
Chap 5	70	3030	2020.1- 2020.5		208				123	
Chap 5	70.1	3030.1	2020.2							
Chap 5	70.2	3030.2	2020.3							
Chap 5	70.3	3030.3	2020.4							
Chap 5	70.4	3030.4	2020.5							
Chap 5	70.5		2020.25							

New	New	MIM	мсм	нна	НО	HSP	ОРТ	RDF	RHC	SNF
Chapter	Section			Pub 11	Pub 10	Pub 21	Pub09	Pub 29	Pub 27	Pub 12
Chap 5	70.6		2020.26							
Chap 5	70.7		2020.8							
Chap 5	80								126	126
Chap 5	90.1								124	
Chap 5	90.2		2070.1							
Chap 5	90.4	42CFR 46135 1								
Chap 6	10		10020	190		155	130		130	190
Chap 6	10.1-4	3760- 3760.3	10020	100	190	100	100	125	100	130
Chap 6	21-21.3	3761- 3761.3								
Chap 6	30-30.2	3762 - 3762.2	10005							
Chap 6	40-40.2	3763- 3763.2	10010					125		
Chap 6	50	3764	10013							
Chap 6	60.1-4	3765A -D								
Chap 6	60	3765	10021	146						
Chap 6	70	3767	10022							
Chap 6 to	80	3768	10025							
Chap 6	90.1-3	3769B -D								
Chap 6	90	3769- 3769A	10028							
Chap 6	100	3770	10031							
Chap 6	100.1	3770A								
Chap 6	100.2	3770B	10031B							
Chap 6	100.3		10031C							
Chap 6	100.4		10031D							
Chap 6	100.5	3770C								
Chap 6	110	3770.1								
Chap 6	120	3771	10034							

New	New Section	МІМ	мсм	ННА	но	HSP	ОРТ	RDF	RHC	SNF
Chapter	Section			Pub 11	Pub 10	Pub 21	Pub09	Pub 29	Pub 27	Pub 12
Chap 6	130	3772	10037							
Chap 6	130.1-3	3772A -C								
Chap 6	140	3773A	10040	192	192	157		126	134	192
Chap 6	140.1	3773B	10040B							
Chap 6	140.2		10040C	192.1	192.1	157.1		126.1	134.1	192.1
Chap 6	140.3		10040							
Chap 6	140.4	3773E	10040	192.2,.3	192.2,3	157.2		126.2,.3		192.2,.3
Chap 6	150.1-2	3774A -B								
Chap 6	150	3774	10043							
Chap 6	160- 160.2	3775- 3775B	10046							
Chap 6	170 - 180	3776	10099	190-193	190-193	156-158	130- 132	125-126	136	190-193

Hospital Manual section **196** is not in this manual but can be found in section 4000 of the PRO manual on the Internet.

Hospital Manual section **197** is not in this manual but can be found in section 9000 of the PRO manual on the Internet.

Hospital Manual section **198** is not in this manual but can be found in section 5000 of the PRO manual on the Internet.

Hospital Manual section **199** is not in this manual but can be found in section 7000 of the PRO manual on the Internet.

Medicare General Information, Eligibility, and Entitlement

Summary Table of Contents

<u>Cha</u>	pter	1 -	<u>General</u>	<u> </u>)verv	view

<u>Chapter 2 - Hospital Insurance and Supplementary Medical Insurance</u>

Chapter 3 - Deductibles, Coinsurance Amounts, and Payment Limitations

Chapter 4 - Physician Certification and Recertification of Services

<u>Chapter 5 - Definitions</u>

Chapter 6 - Disclosure of Information

Medicare General Information, Eligibility, and Entitlement

Chapter 1 - General Overview

Table of Contents

- 10 General Program Benefits
 - 10.1 Hospital Insurance (Part A) for Inpatient Hospital, Hospice, and Skilled Nursing Facility (SNF) Services A Brief Description
 - 10.3 Supplementary Medical Insurance (Part B) A Brief Description
 - 10.4 Basis for Payment
- 20 Administration of the Medicare Program Introduction
 - 20.1 Financing the Program
 - 20.2 Discrimination Prohibited
 - 20.3 Fraud and Abuse General
 - 20.3.1 Definition and Examples of Fraud
 - 20.3.2 Definition and Examples of Abuse
- 30 Federal Government Administration of the Health Insurance Program
 - 30.1 CMS Responsibilities
 - 30.2 Public Health Service Responsibilities
 - 30.3 State Agencies
 - 30.3.1 Certification by State Agencies
 - 30.3.2 Consultation by State Agencies
 - 30.3.3 Coordination by State Agencies
- 40 Role of Part A Intermediaries
 - 40.1 Election of Intermediary
 - 40.2 Intermediary Service to Home Health Agencies

- 50 Role of Part B Carriers
- 60 Background and Responsibilities of the Peer Review Organization (PRO)
 - 60.1 Purpose of PRO Review for the Individual Medicare Beneficiaries
 - 60.2 PRO Responsibility for the Overall Approach to Health Care
 - 60.3 Other PRO Responsibilities
 - 60.4 Statutory Obligations of Practitioners and Other Persons
 - 60.5 Responsibilities of Designated Agents Working With PROs; Organizations Subcontracted for Review
- 70 Institutional Planning and Budgeting

10 - General Program Benefits

(Rev. 1, 09-11-02)

The Health Insurance for the Aged and Disabled Act (title XVIII of the Social Security Act), known as "Medicare," has made available to nearly every American 65 years of age and older a broad program of health insurance designed to assist the nation's elderly to meet hospital, medical, and other health costs. Health insurance coverage has also been extended to persons under age 65 qualifying as disabled and those having end stage renal disease (ESRD) or Lou Gehrig's disease. The program includes two related health insurance programs - hospital insurance (HI) (Part A) and supplementary medical insurance (SMI) (Part B).

10.1 - Hospital Insurance (Part A) for Inpatient Hospital, Hospice, and Skilled Nursing Facility (SNF) Services - A Brief Description

(Rev. 1, 09-11-02)

Hospital insurance is designed to help patients defray the expenses incurred by hospitalization and related care. In addition to inpatient hospital benefits, hospital insurance covers posthospital extended care in SNFs and posthospital care furnished by a home health agency in the patient's home. Blood clotting factors, for hemophilia patients competent to use such factors to control bleeding without medical or other supervision, and items related to the administration of such factors, are also a Part A benefit for beneficiaries in a covered Part A stay. The purpose of these additional benefits is to provide continued treatment after hospitalization and to encourage the appropriate use of more economical alternatives to inpatient hospital care. Program payments for services rendered to beneficiaries by providers (i.e., hospitals, SNFs, and home health agencies) are generally made to the provider.

In each benefit period, payment may be made for up to 90 inpatient hospital days, and 100 days of posthospital extended care services. Under the latter benefit, the beneficiary must have been in a hospital receiving inpatient hospital services for at least 3 consecutive days (counting the

day of admission but not the day of discharge) and be admitted to a SNF or to the SNF level of care in a swing bed hospital within 30 days after the date of hospital discharge. (Under certain circumstances, the 30 days may be extended.)

Where the person became entitled to HI at or after age 65, the hospital discharge must have occurred on or after the first day of the month in which he or she attained age 65. If his/her current entitlement began before age 65; i.e., he or she became entitled to HI under the disability or chronic renal disease provisions of the law, the hospital discharge must have occurred while he or she was so entitled. The 3 consecutive calendar days requirement can be met by stays totaling 3 consecutive days in one or more hospitals.

A SNF provides skilled nursing care and related services for patients who require medical or nursing care, or rehabilitation services. A SNF may be either a separate institution (e.g., a nursing home) or a part of an institution (e.g., a convalescent wing of a hospital). It must be licensed or approved under State or local law, meet the health and safety conditions prescribed by the Secretary of the Department of Health and Human Services (DHHS), and have a written transfer agreement with one or more participating hospitals providing for the transfer of patients between the hospital and the facility, and for the interchange of medical and other information. If an otherwise qualified SNF has attempted in good faith but without success to enter into a transfer agreement, this requirement may be waived by the State agency.

For Medicare purposes, the term SNF does not include any institution which is primarily for the care and treatment of mental diseases. Extended care services include room and board; skilled nursing care by or under the supervision of a registered nurse; physical, occupational, or speech therapy; medical social services, drugs, biologicals, supplies, appliances, and equipment; and other services ordinarily furnished by or under arrangements made by the facility. No payment may be made for items or services which would not be covered in a hospital, or for custodial care when that is the only type of care that the beneficiary needs.

The services of residents and interns of a hospital with which the facility has a swing bed "transfer" agreement and other diagnostic and therapeutic services furnished by such a hospital are covered, but only if billed through the SNF.

Under §4005(b)(2) of the Omnibus Budget Reconciliation Act of 1987, effective for swing bed agreements entered into after March 31, 1988, hospitals with more than 49 beds (but less than 100 beds) are subject to the following conditions. However, these conditions were eliminated by section 408 of the Balanced Budget Refinement Act of 1999 (BBRA), effective with the start of the facility's fourth cost reporting period that begins on or after July 1, 1998. (For those facilities that received no Medicare payment prior to October 1, 1995, this change is effective as of the date of BBRA's enactment, November 29, 1999.)

If there is an available SNF bed in the geographic region, the hospital must transfer the extended care patient within 5 days of the availability date (excluding weekends and holidays) unless the patient's physician certifies within that 5-day period, that transfer of that patient to that facility is not medically appropriate on the availability date. In order to do this, the hospital must identify all SNFs in the geographic region and enter into agreements with them for the transfer of extended care patients. The agreement must call for the SNF to notify the hospital of the availability of beds and the dates these beds will be available for extended care patients.

For each cost reporting period, payment may not be made for patient days of extended care services that exceed 15 percent of the total number of available patient days (except that such payment shall continue to be made for those patients who are receiving extended care services at the time the hospital reaches the 15 percent limit). The limit is calculated by multiplying the average number of licensed beds by the total number of days in the cost reporting period.

Hospitals having fewer than 50 beds and rural hospitals which entered into agreements before March 31, 1988 (i.e., those which were licensed for more than 49 beds but who were operating as a 50 or less bed facility), are not subject to the 5-weekday transfer requirement or the payment limitation for extended care days. (See § 2230.10 of the Provider Reimbursement Manual, Part 1, for the explanation of the payment limitation.)

"Geographic region" is an area which includes the SNFs with which a hospital has traditionally arranged transfers and all other SNFs within the same proximity to the hospital. In the case of a hospital without existing transfer practices upon which to base a determination, the geographic region is an area which includes all the SNFs within 50 miles of the hospital unless the hospital can demonstrate that the SNFs are inaccessible to its patients. In the event of a dispute as to whether a SNF is within this region or the SNF is inaccessible to hospital patients, the CMS regional office shall make a determination.

Hospices also provide Part A hospital insurance services such as short term inpatient care. In order to be eligible to elect hospice care under Medicare, an individual must be entitled to Part A of Medicare and be certified as being terminally ill. An individual is considered to be terminally ill if the individual has a medical prognosis that his or her life expectancy is 6 months or less if the illness runs its normal course.

10.2 - Home Health Services

(Rev. 1, 09-11-02)

To qualify for home health benefits under either Part A or Part B of the program, a beneficiary must be confined to his/her home, under the care of a physician, and in need of skilled nursing services on an intermittent basis, physical therapy, or speech therapy. Being "confined to the home" does not mean a beneficiary can never leave the home. See Benefit Policy, Chapter 7, for the complete definition of homebound. A beneficiary who requires one or more of these services in the treatment of his/her illness or injury and otherwise qualifies for home health benefits is eligible to have payment made on his/her behalf for the skilled nursing, physical or speech therapy he or she needs, as well as for any of the other home health services specified in the law. These services include occupational therapy, medical social services, the use of medical supplies and medical appliances, and the part-time or intermittent services of home health aides. Conversely, a patient who does not require intermittent skilled nursing or physical or speech therapy cannot qualify to have payment made under the program for any home health services furnished him. Excluded as home health services are the costs of housekeepers, food service arrangements, and transportation to outpatient facilities.

To be covered, the home health services must be needed for a condition for which the patient required inpatient hospital services or extended care services. See Benefit Policy, Chapter 7, for a description of services covered. Discharge from the hospital must have occurred in a month in

which the patient has attained age 65 or was entitled to health insurance benefits under the disability or chronic renal disease provisions of the law.

Home health services are services provided by a home health agency or by others under arrangements with such an agency. A home health agency is a public agency or private organization which is primarily engaged in providing skilled nursing and other therapeutic services. Where applicable the agency must be licensed under State or local law, or be approved by the State or local licensing agency as meeting the licensing standards. Examples of home health agencies are visiting nurse associations, official health agencies, and hospital-based home care programs. To participate in the health insurance program, a home health agency must meet certain other requirements included in the law as well as health and safety conditions prescribed by the Secretary of the Department of Health and Human Services. It may not qualify under hospital insurance, however, if it is primarily engaged in the treatment of mental diseases; such an agency may qualify only under supplementary medical insurance.

Home health services are usually furnished on a visiting basis in a place of residence used as the individual's home. However, outpatient services in a hospital, SNF, or rehabilitation center are covered home health services, if arranged for by a home health agency, when equipment is required that cannot be made available in the patient's home.

The services of an intern or resident-in-training are covered if the agency has an affiliation with or is under common control of a hospital providing such medical services and the agency bills for such services.

Prior to July 1, 1981, home health services under hospital insurance included up to 100 home health visits, after the beginning of one benefit period and before the beginning of the next. The visits must have been furnished to a patient within 1 year of his/her most recent discharge from a hospital where he or she was an inpatient for at least 3 consecutive calendar days (counting the day of admission, but not the day of discharge). If, after his/her hospitalization, he or she had a covered stay in a SNF, the 1 year during which the patient may receive home health services began with the discharge from the SNF. A plan of treatment must have been established within 14 days after the hospital or SNF discharge. Home health services were also provided under supplementary medical insurance where the 100-visit limit under Part A was exceeded.

Effective July 1, 1981, the 100-visit limitation under Parts A and B, and the prior inpatient stay requirement under Part A were eliminated. In addition, a person could qualify for home health services based on his or her need for skilled nursing services on an intermittent basis, physical therapy, speech therapy, or occupational therapy. Effective December 1, 1981, occupational therapy was eliminated as a basis for entitlement to home health services. However, if a person has otherwise qualified for home health services because of the need for skilled nursing care, physical or speech therapy, the patient's eligibility for home health services may be extended solely on the basis of the continuing need for occupational therapy.

Effective January 1, 1998, the first 100 visits must be paid under Part A if the beneficiary is entitled under Part A, and the remainder of the visits may be paid under Part B.

10.3 - Supplementary Medical Insurance (Part B) - A Brief Description (Rev. 1, 09-11-02)

To obtain SMI, an eligible individual must enroll during an enrollment period and pay the required premiums. An individual is eligible to enroll if they are entitled to HI or are 65 years of age and a citizen or resident alien who meets certain residence requirements. SMI provides for payment to participating providers for furnishing covered services after a yearly cash deductible is met. The voluntary medical insurance plan is designed to supplement the basic hospital insurance coverage. It provides coverage for home health visits not available under hospital insurance (e.g., no Part A entitlement or visits after the first 100 visits) and for medical and other health services. Payment may not be made under Part B for any service that may be paid under Part A. However, where payment is not possible under Part A (e.g., no Part A entitlement or benefits are exhausted) payment may be made under Part B if the service is covered.

Subject to coverage and limitations described in the Benefit Policy Publication, the following services are covered under Part B.

- Physicians' services;
- Services and supplies (including drugs and biologicals which are not usually selfadministered by the patient) furnished as an incident to a physician's professional service, of kinds which are commonly furnished in physicians' offices and are commonly either rendered without charge or included in the physicians' bills;
- Hospital services (including drugs and biologicals which are not usually self-administered by the patient) incident to physicians' services rendered to outpatients and partial hospitalization services incident to such services;
- Diagnostic services which are-- (1) furnished to an individual as an outpatient by a hospital or by others under arrangements with them made by a hospital, and (2) ordinarily furnished by such hospital (or by others under such arrangements) to its outpatients for the purpose of diagnostic study;
- Outpatient physical therapy services and outpatient occupational therapy services;
- Rural health clinic services and Federally qualified health center services;
- Home dialysis supplies and equipment, self-care home dialysis support services, and institutional dialysis services and supplies;
- Antigens (subject to quantity limitations prescribed in regulations by the Secretary) prepared by a physician, as defined in section 1861(r)(1) of the Act, for a particular patient, including antigens so prepared which are forwarded to another qualified person (including a rural health clinic) for administration to such patient, from time to time, by or under the supervision of another such physician;
- Services furnished pursuant to a contract under section 1876 of the Act to a member of an eligible organization by a physician assistant or by a nurse practitioner and such services and

supplies furnished as an incident to his/her service to such a member as would otherwise be covered under this part if furnished by a physician or as an incident to a physician's service; and, services furnished pursuant to a risk-sharing contract under section 1876(g) of the Act to a member of an eligible organization by a clinical psychologist (as defined by the Secretary) or by a clinical social worker, and such services and supplies furnished as an incident to such clinical psychologist's services or clinical social worker's services to such a member as would otherwise be covered under this part if furnished by a physician or as an incident to a physician's service;

- Blood clotting factors, for hemophilia patients competent to use such factors to control
 bleeding without medical or other supervision, and items related to the administration of such
 factors, subject to utilization controls deemed necessary by the Secretary for the efficient use
 of such factors;
- Prescription drugs used in immunosuppressive therapy furnished to an individual who receives an organ transplant for which payment is made under this title;
- Services which would be physicians' services if furnished by a physician and which are performed by a physician assistant under the supervision of a physician and which the physician assistant is legally authorized to perform by the State in which the services are performed, and such services and supplies furnished as incident to such services as would be covered if furnished incident to a physician's professional service; and but only if no facility or other provider charges or is paid any amounts with respect to the furnishing of such services;
- Services which would be physicians' services if furnished by a physician and which are performed by a nurse practitioner or clinical nurse specialist working in collaboration with a physician which the nurse practitioner or clinical nurse specialist is legally authorized to perform by the State in which the services are performed, and such services and supplies furnished as an incident to such services as would be covered if furnished incident to a physician's professional service, but only if no facility or other provider charges or is paid any amounts with respect to the furnishing of such services;
- Certified nurse-midwife services;
- Qualified psychologist services;
- Clinical social worker services;
- Erythropoietin for dialysis patients competent to use such drug without medical or other supervision with respect to the administration of such drug, subject to methods and standards established by the Secretary by regulation for the safe and effective use of such drug, and items related to the administration of such drug;
- Prostate cancer screening tests;
- An oral drug (which is approved by the Federal Food and Drug Administration) prescribed for use as an anticancer chemotherapeutic agent for a given indication, and containing an active ingredient (or ingredients), which is the same indication and active ingredient (or

ingredients) as a drug which the carrier determines would be covered if the drug could not be self-administered;

- Colorectal cancer screening tests;
- Diabetes outpatient self-management training services;
- An oral drug (which is approved by the Federal Food and Drug Administration) prescribed for use as an acute anti-emetic used as part of an anticancer chemotherapeutic regimen if the drug is administered by a physician (or as prescribed by a physician)--(1) for use immediately before, at, or within 48 hours after the time of the administration of the anticancer chemotherapeutic agent; and (2) as a full replacement for the anti-emetic therapy which would otherwise be administered intravenously;
- Screening for glaucoma for individuals determined to be at high risk for glaucoma, individuals with a family history of glaucoma and individuals with diabetes;
- Medical nutrition therapy services in the case of a beneficiary with diabetes or a renal disease
 who--(1) has not received diabetes outpatient self-management training services within a
 time period determined by the Secretary; (2) is not receiving maintenance dialysis for which
 payment is made under section 1881 of the Act; and (3) meets such other criteria determined
 by the Secretary after consideration of protocols established by dietitian or nutrition
 professional organizations;
- Diagnostic X-ray tests (including tests under the supervision of a physician, furnished in a place of residence used as the patient's home, if the performance of such tests meets such conditions relating to health and safety as the Secretary may find necessary and including diagnostic mammography if conducted by a facility that has a certificate (or provisional certificate) issued under section 354 of the Public Health Service Act), diagnostic laboratory tests, and other diagnostic tests;
- X-ray, radium, and radioactive isotope therapy, including materials and services of technicians;
- Surgical dressings, and splints, casts, and other devices used for reduction of fractures and dislocations;
- Durable medical equipment;
- Ambulance service where the use of other methods of transportation is contraindicated by the individual's condition, but only to the extent provided in regulations;
- Prosthetic and orthotic devices (other than dental) which replace all or part of an internal body organ (including colostomy bags and supplies directly related to colostomy care), including replacement of such devices, and including one pair of conventional eyeglasses or contact lenses furnished subsequent to each cataract surgery with insertion of an intraocular lens:
- Leg, arm, back, and neck braces, and artificial legs, arms, and eyes, including replacements if

required because of a change in the patient's physical condition;

• Vaccines: (1) pneumococcal vaccine and its administration and, subject to section 4071(b) of the Omnibus Budget Reconciliation Act of 1987, (2) influenza vaccine and its administration; and (3) hepatitis B vaccine and its administration, furnished to an individual who is at high or intermediate risk of contracting hepatitis B;

NOTE: A charge separate from the ESRD composite rate will be recognized and paid for administration of the vaccine to ESRD patients.

NOTE: For Medicare program purposes, the hepatitis B vaccine may be administered upon the order of a doctor of medicine or osteopathy by home health agencies, SNFs, renal dialysis facilities (RDFs), hospital outpatient departments, persons recognized under the "incident to physicians' services" provision of law, and, of course, doctors of medicine and osteopathy.

- Services of a certified registered nurse anesthetist;
- Subject to section 4072(e) of the Omnibus Budget Reconciliation Act of 1987, extra-depth shoes with inserts or custom molded shoes with inserts for an individual with diabetes, if-(1) the physician who is managing the individual's diabetic condition (a) documents that the individual has peripheral neuropathy with evidence of callus formation, a history of preulcerative calluses, a history of previous ulceration, foot deformity, or previous amputation, or poor circulation, and (b) certifies that the individual needs such shoes under a comprehensive plan of care related to the individual's diabetic condition; (2) the particular type of shoes are prescribed by a podiatrist or other qualified physician (as established by the Secretary); and (3) the shoes are fitted and furnished by a podiatrist or other qualified individual (such as a pedorthist or orthotist, as established by the Secretary) who is not the physician described in (1) above (unless the Secretary finds that the physician is the only such qualified individual in the area);
- Screening mammography;
- Screening pap smear and screening pelvic exam; and
- Bone mass measurement.

No diagnostic tests performed in any laboratory, including a laboratory that is part of a rural health clinic, or a hospital (which, for purposes of this sentence, means an institution considered a hospital for purposes of section 1814(d)) of the Act shall be included unless such laboratory:

- Is situated in any State in which State or applicable local law provides for licensing of establishments of this nature, (1) is licensed pursuant to such law, or (2) is approved, by the agency of such State or locality responsible for licensing establishments of this nature, as meeting the standards established for such licensing;
- Meets the certification requirements under section 353 of the Public Health Service Act;
 and

• Meets such other conditions relating to the health and safety of individuals with respect to whom such tests are performed as the Secretary may find necessary.

There shall be excluded from the diagnostic services specified any item or service which would not be included if it were furnished to an inpatient of a hospital. None of the items and services referred to in the preceding paragraphs of this subsection which are furnished to a patient of an institution which meets the definition of a hospital for purposes of section 1814(d) of the Act shall be included unless such other conditions are met as the Secretary may find necessary relating to health and safety of individuals with respect to whom such items and services are furnished.

Effective October 1, 1983, in order to be covered under Medicare Part A or Part B, any nonphysician service provided to an inpatient of a hospital must be provided directly by the hospital or under an arrangement.

10.4 - Basis for Payment

(Rev. 1, 09-11-02)

(See Claims Processing, Pub 100-4, for a description of the basis for payment for the various services.)

20 - Administration of the Medicare Program - Introduction

(Rev. 1, 09-11-02)

The conduct of the program has been delegated by the Secretary of the Department of Health and Human Services to the Administrator of the Centers for Medicare & Medicaid Services (CMS). Congress has also provided substantial administrative roles for the States and for voluntary insurance organizations in recognition of their experience in the health care and insurance fields.

The law does not permit the Federal Government to exercise supervision or control over the practice of medicine, the manner in which medical services are provided, and the administration or operation of medical facilities. The patient is free to choose any qualified institution, agency, or person offering him/her services. The responsibility for treatment and the control of care remains with the individual's physician and the hospital or other facility or agency furnishing services. The individual may keep or obtain any other health insurance he/she desires including the choice to enroll in a Medicare+Choice plan. More information about Medicare+Choice plans is in the Managed Care Manual.

20.1 - Financing the Program

(Rev. 1, 09-11-02)

Part A is financed through separate payroll contributions paid by employees, employers, and self-employed persons. The proceeds are deposited to the account of the Federal Hospital Insurance Trust Fund, which is used only for hospital insurance benefits and administrative expenses. Federal employees and State and local employees who do not pay the full FICA tax must pay the HI portion; they are not eligible for monthly Social Security or railroad retirement

benefits. The cost of providing Part A benefits to other persons who are not Social Security or railroad retirement beneficiaries is met by appropriations to the Federal Hospital Insurance Trust Fund from general revenues or through premium payments.

Part B is financed by monthly premiums of those who voluntarily enroll in the program and by the Federal Government which makes contributions from general revenues. All premiums and Government contributions are deposited in a separate account known as the Federal Supplementary Medical Trust Fund. Money from this fund is used only to pay for Part B benefits and administrative expenses.

20.2 - Discrimination Prohibited

(Rev. 1, 09-11-02)

Participating providers of Part A services under the supplementary medical insurance program (e.g., hospitals, SNFs, HHAs, hospices, outpatient physical therapy (OPT), comprehensive outpatient rehabilitation facilities (CORFs), occupational therapy and speech pathology providers, and renal dialysis facilities) must comply with the requirements of title VI of the Civil Rights Act of 1964. Under the provisions of that Act, a participating provider is prohibited from making a distinction on the grounds of race, color, or national origin, in the treatment of patients, the use of equipment, other facilities, and the assignment of personnel to provide services.

DHHS is responsible for investigating complaints of noncompliance.

20.3 - Fraud and Abuse - General

(Rev. 1, 09-11-02)

Providers and suppliers have an obligation, under law, to conform to the requirements of the Medicare program. Fraud and abuse committed against the program may be prosecuted under various provisions of the United States Code and could result in the imposition of restitution, fines, and, in some instances, imprisonment. In addition, there is also a range of administrative sanctions (such as exclusion from participation in the program) and civil monetary penalties that may be imposed when facts and circumstances warrant such action.

Following are definitions and examples of fraud and abuse. These definitions and examples give a better understanding of the types of practices that are forbidden, under law, in the Medicare program.

20.3.1 - Definition and Examples of Fraud

(Rev. 1, 09-11-02)

Fraud is defined as making false statements or representations of material facts in order to obtain some benefit or payment for which no entitlement would otherwise exist. These acts may be committed either for the person's own benefit or for the benefit of some other party. In order to

prove that fraud has been committed against the Government, it is necessary to prove that fraudulent acts were performed knowingly, willfully, and intentionally.

Examples of fraud include, but are not limited to, the following:

- Billing for services that were not furnished and/or supplies not provided. This includes billing Medicare for appointments that the patient failed to keep;
- Altering claims forms and/or receipts in order to receive a higher payment amount;
- Duplicating billings that includes billing both the Medicare program and the beneficiary, Medicaid, or some other insurer in an effort to receive payment greater than allowed;
- Offering, paying, soliciting, or receiving bribes, kickbacks, or rebates, directly or indirectly, in cash or in kind, in order to induce referrals of patients or the purchase of goods or services that may be paid for by the Medicare program;
- Falsely representing the nature of the services furnished. This encompasses describing a noncovered service in a misleading way that makes it appear as if a covered service was actually furnished;
- Billing a person who has Medicare coverage for services provided to another person not eligible for Medicare coverage; and
- Using another person's Medicare card to obtain medical care.

20.3.2 - Definition and Examples of Abuse

(Rev. 1, 09-11-02)

Abuse describes practices that, either directly or indirectly, result in unnecessary costs to the Medicare program. Many times abuse appears quite similar to fraud except that it is not possible to establish that abusive acts were committed knowingly, willfully, and intentionally.

Following are three standards that CMS uses when judging whether abusive acts in billing were committed against the Medicare program:

- Reasonable and necessary;
- Conformance to professionally recognized standards; and
- Provision at a fair price.

Examples of abuse include, but are not limited to, the following:

- Charging in excess for services or supplies;
- Providing medically unnecessary services or services that do not meet professionally recognized standards;
- Billing Medicare based on a higher fee schedule than for non-Medicare patients;

- Submitting bills to Medicare that are the responsibility of other insurers under the Medicare secondary payer (MSP) regulation; and
- Violating the participating physician/supplier agreement.

Although these types of practices may initially be categorized as abusive in nature, under certain circumstances they may develop into fraud if there is evidence that the subject was knowingly and willfully conducting an abusive practice.

30 - Federal Government Administration of the Health Insurance Program

(Rev. 1, 09-11-02)

DHHS has overall responsibility for administering the hospital insurance and voluntary SMI programs. Two major agencies - CMS and the Public Health Service - are involved in specified administrative functions.

30.1 - CMS Responsibilities

(Rev. 1, 09-11-02)

CMS is responsible for policy formulation. The central and regional offices are responsible for the general management and operation of the program. In brief, CMS's responsibilities include the following:

- Determining an individual's entitlement to benefits in consultation with the Social Security Administration (SSA);
- Determining the nature and duration of services for which a beneficiary's benefits may be paid;
- Establishing, maintaining, and administering agreements with State agencies, providers of services, and intermediaries;
- Establishing operational policy for contractors;
- Developing operational instructions and official interpretations of policy for contractors;
- Formulating major policies regarding conditions of participation for providers in consultation with the Public Health Service:
- Developing and maintaining statistical research and actuarial programs;
- Managing general finances of the program; and
- Managing the Medicare Premium Collection Operation.

CMS Regional Offices are responsible for assuring that contractors meet applicable Federal requirements under the provisions of their contracts. They also:

- Provide liaison, direction, and technical assistance to contractors in the day-to-day management of their operations;
- Interpret CMS guidelines, policies, and procedures applicable to contractor activities;
- Analyze contractor budgets and spending patterns to assure that funds are economically and appropriately utilized;
- Conduct assessments of contractor operations;
- Review contractor actions; and
- Provide feedback to each contractor.

30.2 - Public Health Service Responsibilities

(Rev. 1, 09-11-02)

The Public Health Service is responsible for administering the professional health aspects of the program. In brief, its responsibilities include the following:

- Consulting and recommending to CMS matters concerning the development of health and safety standards and other guidelines needed for determining whether providers of services meet the conditions of participation under the program;
- Consulting and advising State agencies concerning the application of standards for providers;
 and
- Coordinating programs and activities necessary in studying the utilization of services under the program.

30.3 - State Agencies

(Rev. 1, 09-11-02)

The States, by agreement with the Secretary, are assigned significant administrative functions to the extent that each is willing and capable of discharging such responsibilities.

30.3.1 - Certification by State Agencies

(Rev. 1, 09-11-02)

Facilities desiring to participate in either the Medicare or Medicaid programs must meet participation conditions for certification. State agencies certify to DHHS whether providers satisfy, and continue to satisfy, their respective conditions of participation in the Medicare and Medicaid programs. The Secretary, DHHS, certifies facilities requesting participation in the Medicare and Medicaid programs. States certify those facilities that request participation in the Medicaid program only.

The State function of making certifications is intended to be a natural adjunct to ongoing State activities (such as the licensing of health care facilities and the setting of standards).

30.3.2 - Consultation by State Agencies

(Rev. 1, 09-11-02)

A State consults with providers of services that need and request participation condition assistance. For Medicare participation, the Secretary, DHHS, must approve the consultation service rendered by the State certifying agency.

30.3.3 - Coordination by State Agencies

(Rev. 1, 09-11-02)

A State coordinates activities with other State programs that involve payment for health care, quality of care, and location of health facilities. Coordinating these activities is essential in assuring effective and economical use of existing State facilities and trained personnel and to prevent duplication of effort.

40 - Role of Part A Intermediaries

(Rev. 1, 09-11-02)

The Part A intermediary is a public or private agency or organization that has entered into an agreement with CMS to enroll legitimate providers into the Medicare program and process Medicare claims under both Part A services and Part B services.

Intermediaries make payments to providers. The amount of payment to a provider is restricted to the lower of the billed charge, the reasonable cost of covered services or the fee schedule amount. Hospices are paid on a per diem amount that is prospectively set. SNFs and HHAs are paid based on a Prospective Payment System (PPS). (See Provider Reimbursement Manual, Part 1, §§2800ff.)

Hospitals are paid based on the PPS. Under this system, Medicare payment is made at a predetermined, specific rate for each hospital discharge. This statement applies to inpatient for acute care hospitals and to inpatient rehabilitation hospitals. Whereas inpatient acute and rehab PPS payment is based on the discharge date, Outpatient PPS (OPPS) payments are based on the Ambulatory Patient Classification (APC) payment for the date of service.

The amount of payment to other types of providers is restricted to the lesser of (a) the reasonable cost of covered services and items; or (b) the billed charges with respect to such services; or (c) the fee schedule amount.

In addition, intermediaries assist in applying safeguards against unnecessary use of covered services, furnish consultative services to serve as a center for communicating with providers, conduct audits of provider records, assist in the beneficiary appeals process, and provide information and advice to institutions and organizations that wish to qualify as providers of services

(See http://www.hcfa.gov/medicare/incardir.htm for a list of intermediaries and service areas.)

40.1 - Election of Intermediary

(Rev. 1, 09-11-02)

Except for HHAs, hospices and freestanding CORFs, providers may elect to be served by an intermediary authorized to serve other providers in its area. Elections are reviewed by the RO and approved unless special or temporary limitations have been placed on the elected intermediary's availability, or an addition to the elected intermediary's workload at the time would be undesirable.

The RO sends the official notice to the contractor of changes in the contractor's list of providers.

40.2 – Intermediary Service to Home Health Agencies

(Rev. 1, 09-11-02)

Under 42 CFR 421.117, CMS is authorized to designate intermediaries to service HHAs and hospices. This provision was implemented through the designation of regional and alternative regional intermediaries to service all HHAs and hospices within the respective intermediary's jurisdictional boundaries. The list of Regional Home Health Intermediaries (RHHIs) and their service areas can be found at the following CMS Web site address:

http://www.hcfa.gov/medicare/incardir.htm#4

In the case of HHAs and hospices based in another Medicare provider (e.g., a hospital or SNF), audit, cost report settlement, and other fiscal functions (such as setting interim payment rates) are performed by the intermediary serving the parent provider.

While RHHIs have been designated to service all HHAs and hospices, CMS has designated an alternative regional intermediary. Where HHAs or hospices can demonstrate that it is in the best interest of the Government for them to not be serviced by the designated RHHI, the provider may request (through the CMS Regional Office) to be serviced by the designated alternative regional intermediary. The following are the designated alternative regional intermediaries and their respective jurisdictions.

Blue Cross and Blue Shield of Alabama.

Alabama, Alaska, Arizona, Arkansas, California, Connecticut, Delaware, District of Columbia, Florida, Georgia, Hawaii, Idaho, Illinois, Indiana, Kentucky, Louisiana, Maine, Maryland, Massachusetts, Michigan, Minnesota, Mississippi, Nevada, New Hampshire, New Jersey, New Mexico, New York, North Carolina, Ohio, Oklahoma, Oregon, Pennsylvania, Puerto Rico, Rhode Island, South Carolina, Tennessee, Texas, Vermont, the Virgin Islands, Virginia, Washington, West Virginia, and Wisconsin.

Blue Cross and Blue Shield United of

Colorado, Iowa, Kansas, Missouri, Montana, Nebraska, North Dakota, South Dakota, Utah, and Wyoming.

Wisconsin

50 - Role of Part B Carriers

(Rev. 1, 09-11-02)

The law requires the Secretary, DHHS, to enter into contracts with carriers to serve in the operation and administration of the non-provider Part B program. Carriers enroll physicians, non-physician health care practitioners and other entities that will submit claims to the carrier, and process Medicare claims and make payments for services and supplies covered by Part B. Other major functions include, for example, controlling over-utilization and communicating with beneficiaries and the health community.

Durable Medical Equipment (DME) Regional Carriers have been given the responsibility of processing durable medical equipment, prosthetic, orthotic, and supply (DMEPOS) claims.

60 - Background and Responsibilities of the Peer Review Organization (PRO) (Rev. 1, 09-11-02)

Section 1153 of the Social Security Act (the Act) requires the Secretary, DHHS, to enter into contracts with physician-approved or physician-access organizations defined as PROs.

PROs are organizations who are responsible for monitoring the quality of care provided to Medicare patients by hospitals, SNFs, home health agencies, Medicare+Choice plans, and other types of health care providers.

PRO review is governed by titles XI and XVIII of the Act as amended, and by regulations contained in:

- 42 CFR 411 Limitation on liability;
- 42 CFR 412 Outlier review, diagnosis related group (DRG) validation, and hospital notices of non coverage;
- 42 CFR 417.605 Immediate PRO review of Health Maintenance Organization (HMO)/Competitive Medical Plan (CMP) notices of discharge;
- 42 CFR 422.622 Immediate review of Medicare+Choice discharge notices;
- 42 CFR 475 Definition of eligible organizations and area designation;
- 42 CFR 476 Assumption and conduct of review;
- 42 CFR 478 PRO reconsideration and appeals;
- 42 CFR 480 Disclosure of information:
- 42 CFR 482 Hospital conditions of participation; and

• 42 CFR 1004 – PRO recommendations of sanctions.

60.1 - Purpose of PRO Review for the Individual Medicare Beneficiaries

(Rev. 1, 09-11-02)

PROs review items or services provided to Medicare beneficiaries to determine:

- Whether services provided or proposed to be provided are reasonable and medically necessary for the diagnosis and treatment of illness or injury, or to improve functioning of a malformed body member, or for prevention of an illness, or for the palliation and management of terminal illness;
- Whether those services furnished or proposed to be furnished on an inpatient basis could be effectively furnished on an outpatient basis, or in an inpatient health care facility of a different type;
- Medical necessity, reasonableness, and appropriateness of inpatient hospital care for which additional payment is sought under the outlier provisions of PPS;
- Whether a hospital has misrepresented admission or discharge information, or has taken an
 action that results in the unnecessary admission of an individual entitled to benefits under
 Part A, unnecessary multiple admissions of an individual, or other inappropriate medical or
 other practices with respect to beneficiaries, or billing for services furnished to beneficiaries;
- The validity of diagnostic and procedural information supplied by the provider to the fiscal intermediary for payment purposes;
- The completeness and adequacy of hospital care provided; and
- Whether the quality of services meets professionally recognized standards of health care.

These activities enable PROs to determine whether Medicare payment may be made for the services claimed and to identify and initiate corrective action where appropriate. PROs have the authority to deny Medicare payment for medically inappropriate and unnecessary admissions. They also investigate individual beneficiary complaints about the quality of care received.

60.2 - PRO Responsibility for the Overall Approach to Health Care

(Rev. 1, 09-11-02)

In addition to individual case review, PROs also help providers improve the overall approach to health care for Medicare beneficiaries.

This function includes the following activities:

• Using clinical and other data bases to examine patterns of care and outcomes, rather than focusing on isolated cases;

- Identifying systemic variations of concern or interest by monitoring patterns of care and outcomes;
- Working with providers to identify provider-specific root causes of systemic variations; and
- Working cooperatively with the health care community to measurably improve the processes and outcomes of care for Medicare beneficiaries.

60.3 - Other PRO Responsibilities

(Rev. 1, 09-11-02)

A. Responsibilities Prior to Review

PROs are responsible for:

- Specifying in their plan and instructions to practitioners and providers the type of evidence they require to document the care ordered or furnished to provide medically necessary, appropriate, and quality health care services; and
- Applying professionally developed criteria for providing care, diagnosis, and treatment based upon typical patterns of practice within their geographic area to evaluate the medical necessity, quality, or appropriateness of services ordered or furnished.

B. Ongoing Review Activities

As a part of their ongoing review activities, PROs must:

- Notify the appropriate agency of the State or Federal Government when they become aware of situations which appear to be improper, but which do not fall within their review responsibilities (e.g., poor quality care in a renal dialysis center);
- Use their authority or influence to enlist the support of other professional or Government agencies to ensure that all providers and practitioners for which they have review responsibilities comply with their obligations (see §1156 of the Act.); and
- Perform beneficiary and physician outreach activities.
- C. Responsibilities as a Result of PRO Review

To act upon information they obtain as a result of their review activities, PROs must:

- Provide information on results of their review (e.g., annual report, periodic meetings with providers/practitioners);
- Identify and seek correction of situations that, if continued, would result in violations under §1156 of the Act;
- Submit reports to the Office of the Inspector General on providers and practitioners found to have substantially violated an obligation in a substantial number of cases, or to have grossly

and flagrantly violated an obligation in one or more instances. This includes referring certain cases to State licensing boards.

D. Additional Activities

PROs perform all other activities specified in the Scope of Work, including any modifications, CMS regulations and instructions, and relevant statutory provisions.

E. Payment Error Prevention Program (PEPP)

In order to reduce inpatient PPS payment errors, PROs must initiate a program of Payment Error Prevention Projects (PEPPs). CMS defines the payment error rate as the number of dollars found to be paid in error out of the total of all dollars paid for inpatient PPS services. CMS provides State-specific error rates to PROs to evaluate performance.

60.4 - Statutory Obligations of Practitioners and Other Persons

(Rev. 1, 09-11-02)

It is the obligation of any health care practitioner or other person who furnishes or orders health care items or services that may be reimbursed under Medicare, to ensure that to the extent of his or her authority, those services are:

- Furnished economically and only when and to the extent medically necessary;
- Of a quality that meets professionally recognized standards of health care; and
- Supported by evidence of the medical necessity and quality of the services in the form and fashion that the reviewing PRO may reasonably require (including copies of the necessary documentation) to ensure that the practitioner or other person is meeting the obligations imposed by Section 1156(a) of the Act

These obligations apply whether payment is made directly to the provider (i.e., assignment) or to the beneficiary, or even if payment is not made.

60.5 - Responsibilities of Designated Agents Working With PROs; Organizations Subcontracted for Review

(Rev. 1, 09-11-02)

The PRO has ultimate responsibility for monitoring the compliance of practitioners and providers with statutory obligations. It is not relieved of any of the responsibility under the sanction regulations in the event of non-performance by an organization with which it has subcontracted for review.

An organization with which the PRO subcontracts to carry out review functions is responsible for:

• Ensuring that practitioners and other persons meet their obligations with respect to the items and services it reviews; and

• Reporting to the PRO (primary contractor) those instances where it appears a violation of an obligation may have occurred or is occurring.

70 - Institutional Planning and Budgeting

(Rev. 1, 09-11-02)

The Social Security Act requires each provider, as a condition of participation under Medicare, to have a written overall plan and budget reflecting an annual operating budget and a capital expenditures plan (that covers at least a 3-year period including the year to which the operating budget is applicable). For this requirement, provider means hospital, critical access hospital, SNF, comprehensive outpatient rehabilitation facility, home health agency, or hospice program.

The annual operating budget will include all anticipated income and expenses related to items which would under generally accepted accounting principles be considered income and expense items. The capital expenditure plan would be expected to include and identify in detail the anticipated sources of finance for, the objectives of, each anticipated expenditure in excess of \$100,000 related to acquisition of land, the improvement of land, buildings, and equipment, and the replacement, modernization, and expansion of the buildings and equipment which would, under generally accepted accounting principles, be considered capital items.

The overall budget and plan will be prepared under the direction of the provider's governing body by a committee consisting of representatives of the governing body, administrative staff and if any, the medical staff. Further, it will be reviewed and updated at least annually. The purpose of the requirement is to assure that providers carry on budgeting and substance by the Government or any of its agents.

Medicare General Information, Eligibility, and Entitlement

Chapter 2 - Hospital Insurance and Supplementary Medical Insurance

Table of Contents

- 10 Hospital Insurance Entitlement
 - 10.1 Insured Status
 - 10.2 Hospital Insurance for the Aged
 - 10.3 Hospital Insurance for Disability Beneficiaries
 - 10.4 Hospital Insurance for Persons Needing Kidney Transplant or Dialysis
 - 10.4.1 Effective Date of Entitlement for Persons on Dialysis
 - 10.4.2 Entitlement Based on Transplant
 - 10.4.3 Effect on Self-dialysis Training on Entitlement
 - 10.4.4 End of Coverage Based on ESRD
 - 10.4.5 Reentitlement for Beneficiaries with ESRD
- 20 Hospital Insurance Obtained by Premium Payment
 - 20.1 Eligibility Requirements
 - 20.2 Premiums for Hospital Insurance
 - 20.3 Beginning of Coverage
 - 20.3.1 Initial Enrollment Period (IEP)
 - 20.3.2 General Enrollment Period (GEP)
- 30 End of Coverage for Hospital Insurance
- 40 Supplementary Medical Insurance (SMI)
 - 40.1 Eligibility for Enrollment
 - 40.2 Automatic Enrollment in SMI
 - 40.3 Enrollment Periods

- 40.3.1 Enrollment During the Individual's Initial Enrollment Period
- 40.3.2 Enrollment During General Enrollment Period
- 40.3.3 Enrollment During the Individual's Special Enrollment Period
- 40.4 Nature and Purpose of State Buy-in
- 40.5 End of Coverage
- 40.6 Termination and Reenrollment
- 40.7 Premiums
 - 40.7.1 Amount of Premiums
 - 40.7.2 Increase in Base Premium Amount
 - 40.7.3 Collection of Premiums
 - 40.7.4 Grace Period for Payment
 - 40.7.5 Payment After Grace Period Applies
 - 40.7.6 Premiums Paid by Other Than the Enrollee
 - 40.7.6.1 Informal Arrangement
 - 40.7.6.2 Premium Payer
 - 40.7.6.3 Formal Group Arrangement
 - 40.7.6.4 Premium Surcharge Payment
 - 40.7.7 Premiums Under Buy-In
- 40.8 Waiver of Enrollment Period Requirements Due to Administrative Error
- 50 Identifying the Patient's Health Insurance Record Using the Health Insurance Card
 - 50.1 Temporary Eligibility Notice
 - 50.2 Health Insurance Claims Numbers (HICNs)
 - 50.3 HICNs Assigned by CMS
 - 50.4 HICNs Assigned by the RRB
 - 50.4.1 Six-Digit Numbers
 - 50.4.2 Valid RRB HICNs
- 60 Medicare Part C, Medicare+Choice

10 - Hospital Insurance Entitlement

(Rev. 1, 09-11-02)

Hospital insurance (HI), as well as supplementary medical insurance (SMI), is available to three basic groups of "insured individuals"- the aged, the disabled, and those with end stage renal disease. Following is an explanation of how an individual becomes "insured" as well as an explanation of the eligibility requirements for each group.

10.1 - Insured Status

(Rev. 1, 09-11-02)

To be eligible for premium-free HI, an individual must be "insured" based on his or her own earnings or those of a spouse, parent, or child. To be insured, the worker must have a specific number of quarters of coverage (QCs); the exact number required is dependent upon whether the person is filing for HI on the basis of age, disability, or end stage renal disease. QCs are earned through payment of payroll taxes under the Federal Insurance Contributions Act (FICA) during the person's working years. QCs earned by an individual who pays the full FICA tax are usable to insure the person for both monthly social security benefits and HI.

Federal employees were exempt from payment of FICA taxes prior to January 1983. However, beginning in January 1983, Federal employees became subject to the HI portion of the FICA tax (those actually employed in January 1983 were also deemed to have earned HI quarters of coverage for their Federal service prior to January 1983). Also, Government employees who pay only the HI portion of the FICA tax are only insured for HI; they are not insured for monthly social security benefits.

State and local Government employees hired after March 31,1986, are eligible for Medicare coverage and must pay the HI portion of the FICA tax. A State may elect to cover employees hired prior to April 1986 for the Medicare portion of the FICA tax by requesting an agreement or modification of its existing agreement under section 218 of the Social Security Act.

10.2 - Hospital Insurance for the Aged

(Rev. 1, 09-11-02)

To be eligible for HI on the basis of age, a person must be age 65 or older and either eligible for monthly social security or railroad retirement cash benefits, or would be eligible for such benefits if the worker's Government QCs were regular social security QCs. An individual who is insured for monthly benefits need not actually file for benefits to receive HI benefits. If such a person continues to work beyond age 65, he or she may instead elect to file an application for HI only.

Premium-free HI for the aged begins with the month in which the individual attains age 65, provided he or she files an application for HI or for cash benefits and HI within 6 months of the month in which he or she attains age 65. If the application is filed later than that, HI entitlement

can be retroactive for only 6 months. An individual attains age 65 on the day before his or her 65th birthday. For example, if an individual is born on August 1, the attainment date is July 31, and HI begins with July 1. Entitlement generally does not end until death.

10.3 - Hospital Insurance for Disability Beneficiaries

(Rev. 1, 09-11-02)

A disabled person who is entitled to social security or railroad retirement benefits on the basis of disability is automatically entitled to HI after 24 months of entitlement to such benefits. Since there is a 5-month disability benefits waiting period, the person actually becomes entitled to HI after being disabled for 29 months.

In addition, disabled persons who are not insured for monthly Social Security disability benefits but would be insured for such benefits if Government QCs were treated as social security QCs, are deemed to be entitled to disability benefits and automatically entitled to HI after being disabled for 29 months.

The months in the Medicare qualifying period need not be consecutive so that months from a previous period of disability benefit entitlement generally may be counted in determining when the qualifying period requirement is met. HI entitlement on the basis of disability is available not only to the worker, but to the widow, widower, or child of a deceased, disabled, or retired worker if any of them become disabled within the meaning of the Social Security or Railroad Retirement Acts.

If an individual recovers from a disability, HI entitlement ends with the month after the month he or she is notified of the disability termination. For example, if notification is November 15, entitlement ends December 31. However, if the individual's disability benefit entitlement ends only because he or she was working, HI entitlement may continue for up to 78 additional months.

10.4 - Hospital Insurance for Persons Needing Kidney Transplant or Dialysis (Rev. 1, 09-11-02)

Individuals of any age with end stage renal disease (ESRD) who receive dialysis on a regular basis or a kidney transplant are eligible for HI (and are deemed enrolled for Supplementary Medical Insurance (SMI) unless such coverage is refused) if they file an application. They must also meet certain work requirements for insured status under the Social Security or railroad retirement programs, or be entitled to monthly Social Security benefits or an annuity under the Railroad Retirement Act, or be the spouse or dependent child of an insured or entitled person.

10.4.1 - Effective Date of Entitlement for Persons on Dialysis

(Rev. 1, 09-11-02)

Entitlement usually begins after a 3-month waiting period has been served, i.e., with the first day of the third month after the month in which a course of regular dialysis begins. Entitlement

begins before the waiting period has expired if the individual receives a transplant or participates in a self-dialysis training program during the waiting period.

10.4.2 - Entitlement Based on Transplant

(Rev. 1, 09-11-02)

Entitlement begins with the month the individual is admitted as an inpatient to a hospital for procedures in preparation for or in anticipation of a kidney transplant, provided the transplant surgery takes place within the following 2 months. If the transplant is delayed more than 2 months after the preparatory hospitalization, entitlement begins with the second month prior to the month of transplant. Under the Medicare, Medicaid, and State Children's Health Insurance Program (SCHIP) Benefits Improvement and Protection Act of 2000, Congress extended immunosuppressive drug benefits to lifetime, as long as the beneficiary is entitled to Medicare and was entitled to Medicare when his/her transplant took place.

10.4.3 - Effect on Self-dialysis Training on Entitlement

(Rev. 1, 09-11-02)

Entitlement begins with the first month of the course of dialysis if the individual is expected to complete the self-dialysis training program and self-dialyze thereafter.

10.4.4 - End of Coverage Based on ESRD

(Rev. 1, 09-11-02)

HI coverage based on ESRD ends with the earliest of the following dates:

- The day an individual dies,
- The last day of the 12th month after the month the course of dialysis is discontinued, unless
 the individual receives a kidney transplant during that period or begins another course of
 dialysis, or
- The last day of the 36th month after the month the person receives a kidney transplant.

10.4.5 - Reentitlement for Beneficiaries with ESRD

(Rev. 1, 09-11-02)

If a person whose entitlement based on ESRD has ended begins a new course of regular dialysis or has a kidney transplant, reentitlement begins without a waiting period.

20 - Hospital Insurance Obtained by Premium Payment

(Rev. 1, 09-11-02)

20.1 - Eligibility Requirements

(Rev. 1, 09-11-02)

Individuals who want hospital insurance coverage but who are not otherwise eligible (i.e., do not qualify under either the regular or deemed insured provisions, in §§10.1, 10.2, 10.3 10.4 above) may obtain such coverage by enrolling timely, paying a monthly premium, and upon meeting the following three requirements:

- Attaining age 65,
- Enrolling or already having enrolled in the SMI program,
- Being a resident of the U.S., and either
 - A citizen, or
 - An alien lawfully admitted for permanent residence who has resided continuously in the U.S. for the 5 years prior to the month of enrollment.

20.2 - Premiums for Hospital Insurance

(Rev. 1, 09-11-02)

Each year, the Secretary of the Department of Health and Human Services announces the amount of the hospital insurance premium payable for each month in the following calendar year. The applicable premium amount is 33/76 (about 43.4 percent) of the hospital insurance deductible for that calendar year. The premium is rounded to the nearest \$1. An individual who is not entitled to free Part A and wishes this type of coverage must enroll him/herself during the appropriate enrollment period. A premium is due for each month of entitlement under Premium HI.

20.3 - Beginning of Coverage

(Rev. 1, 09-11-02)

20.3.1 - Initial Enrollment Period (IEP)

(Rev. 1, 09-11-02)

Persons may enroll for premium hospital insurance by filing a request during the IEP which begins the third month before the month of first eligibility and lasts for 7 months. The individual's IEP for premium hospital insurance is in most cases the same 7-month period as the IEP for SMI.

The individual may enroll for hospital insurance when he or she enrolls for SMI, or later during his or her IEP. The beginning date of an individual's premium hospital insurance coverage period is determined by the rules applicable to SMI coverage based on an SMI enrollment during an IEP. (See chapter 2 §40 below.)

20.3.2 - General Enrollment Period (GEP)

(Rev. 1, 09-11-02)

Eligible persons who have not enrolled for premium hospital insurance during their IEP, or whose premium hospital insurance has been terminated, may enroll during a GEP (January 1 - March 31 of each year) if they are enrolling or have enrolled for SMI. As with SMI, an eligible person who enrolls for hospital insurance during a GEP will have hospital insurance coverage beginning the following July.

The same restrictions on enrollment and reenrollment apply as in the SMI program. Individuals who enroll late may pay a 10 percent premium penalty.

30 - End of Coverage for Hospital Insurance

(Rev. 1, 09-11-02)

A beneficiary's entitlement under the provision terminates:

- With the month of the individual's death,
- With the last day of the third month following the premium billing month (for nonpayment of premiums),
- With the end of the month following the month in which he or she files a voluntary request for termination,
- With the month SMI coverage is terminated, or
- With the month before the month in which he or she becomes entitled to hospital insurance under the regular or ESRD provisions. (See chapter §2, §10.4 of this manual.)

40 - Supplementary Medical Insurance (SMI)

(Rev. 1, 09-11-02)

Unlike the HI benefits program, which is largely financed by compulsory taxes on employers, employees, and the self-employed, the SMI benefits program is a voluntary program financed from premium payments by enrollees, together with contributions from funds appropriated by the Federal Government, and certain deductible and cost-sharing provisions.

40.1 - Eligibility for Enrollment

(Rev. 1, 09-11-02)

To obtain SMI coverage, eligible individuals must enroll in the plan during an enrollment period open to them and pay the required premiums. They are eligible to enroll if they are entitled to premium-free HI, or are age 65 and resident citizens or resident aliens. To qualify as a resident alien, an individual must have been lawfully admitted for permanent residence and have resided

continuously in the United States during the 5 years immediately preceding the month of enrollment

40.2 - Automatic Enrollment in SMI

(Rev. 1, 09-11-02)

Beneficiaries (except those residing in foreign countries or Puerto Rico) are deemed to have enrolled in SMI in the month before the month for which they are entitled to HI so that HI and SMI coverage start in the same month. Thus, monthly beneficiaries other than disability beneficiaries are deemed to enroll in the month before attainment of age 65. Disability beneficiaries are deemed to enroll in the 24th consecutive month of entitlement to disability benefits.

Every potential "deemed" enrollee is given a reasonable opportunity (at least 2 months) to decline SMI enrollment by filing notice to this effect. He or she is deemed not to have enrolled and does not incur any premium liability. A refusal of SMI not timely filed is treated as a request for voluntary termination under existing disenrollment rules once it has been established that the individual does not wish to have SMI.

40.3 - Enrollment Periods

(Rev. 1, 09-11-02)

Enrollment is possible only during specified enrollment periods:

- An individual's initial enrollment period is of 7 months duration. It begins 3 full calendar months before and ends 3 full calendar months after the month in which the individual first meets all the requirements for enrollment.
- A general enrollment period occurs each year from January 1 through March 31. Coverage is effective the following July 1. These periods afford enrollment opportunities to those who failed to enroll during their initial enrollment periods and to those whose enrollment has terminated.
- A special enrollment period (SEP) is available, effective November 1984, for individuals age 65 or over who did not enroll in SMI when first eligible (or who terminated SMI enrollment because of coverage under a group health plan (GHP) based on their own or a spouse's current employment status). These individuals may enroll in SMI anytime while covered under the GHP or during the 8-month period immediately following the last month of GHP coverage based on current employment status.
- An SEP also became available, effective January 1987, for disabled beneficiaries under age 65 who did not enroll in SMI when first eligible (or who terminated SMI enrollment because of coverage under a GHP or a large group heath plan (LGHP) based on their own or a family member's current employment status. These individuals may also enroll in SMI anytime while covered under the GHP/LGHP or during the 8-month period immediately following the last month of GHP/LGHP coverage based on current employment status.

There is no SEP for individuals who have ESRD. This includes individuals who are dually entitled to Medicare based on ESRD and age or disability.

40.3.1 - Enrollment During the Individual's Initial Enrollment Period

(Rev. 1, 09-11-02)

Coverage begins on the first day of:

- The month in which the individual first becomes eligible for SMI if enrollment takes place during the first 3 months of the initial enrollment period,
- The month following the month of enrollment if enrollment occurs during the fourth month of the initial enrollment period,
- The second month following the month of enrollment if enrollment occurs during the fifth month of the initial enrollment period, or
- The third month following the month of enrollment if enrollment occurs during the sixth or seventh month of the initial enrollment period.

40.3.2 - Enrollment During General Enrollment Period

(Rev. 1, 09-11-02)

Coverage begins the following July 1.

40.3.3 - Enrollment During the Individual's Special Enrollment Period

(Rev. 1, 09-11-02)

If an individual enrolls in SMI or premium HI while still covered under a GHP or LGHP or during the first full month when not enrolled in a GHP/LGHP based on current employment status, coverage begins either with:

- The first day of the month of SMI or premium HI enrollment, or
- At the individual's option, with the first day of any of the following 3 months.

If the individual enrolls during any of the 7 remaining months of the special enrollment period, coverage begins with the first day of the month following the month of enrollment;

40.4 - Nature and Purpose of State Buy-in

(Rev. 1, 09-11-02)

Under the buy-in program, States may enroll certain groups of needy people in the supplementary medical insurance program and pay their premiums. The purpose of buy-in is to permit the State, as part of its total assistance plan, to provide medical insurance protection to designated categories of needy individuals who are eligible for Medicaid and also meet the

eligibility requirements for SMI. It has the effect of transferring some medical costs for this population from the title XIX Medicaid program, which is partially State financed, to the title XVIII program, which is financed by the Federal Government. Federal matching money is available through the Medicaid program to assist the States with the premium payments for certain buy-in enrollees.

40.5 - End of Coverage

(Rev. 1, 09-11-02)

An individual may notify CMS in writing, at any time, that he or she no longer wishes to participate in the supplementary medical insurance plan. Termination of coverage takes effect at the close of the calendar month following the calendar month in which the request for termination was filed.

Enrollment under medical insurance is terminated because of nonpayment of premiums. Termination is effective with the end of the grace period provided for payment of premiums. (See chapter 2, §40.7.4 of this manual.)

For individuals under age 65, enrollment is terminated because HI entitlement ended. SMI terminates at the same time as HI

If not terminated sooner, coverage ends with the beneficiary's death.

40.6 - Termination and Reenrollment

(Rev. 1, 09-11-02)

An individual whose enrollment has terminated may reenroll only in a general enrollment period or a special enrollment period if the requirements are met.

40.7 - Premiums

(Rev. 1, 09-11-02)

A premium is due for each month of entitlement under SMI. The SMI premium covers about one-fourth of the cost of the Part B program.

40.7.1 - Amount of Premiums

(Rev. 1, 09-11-02)

In the third quarter of each calendar year, the Secretary of the Department of Health and Human Services announces the base premium for SMI coverage that is payable as of the following January 1. The SMI premium is the same for all enrollees even though the cost to the program is greater for ESRD enrollees and the disabled than it is for the aged. The SMI premium is refigured each year and is based on the lower of one-half of SMI program costs per aged enrollee or the general percentage by which social security benefits were increased during the calendar year in which the announcement occurs. (During the 5-year period, January 1984 through

December 1988, the formula for computing the SMI premium has been modified to provide that the Part B premium reflects exactly one-fourth of the cost of the Part B program. Beginning with the premium announced for the period beginning January 1989, however, the method of computing the SMI premium reverts back to the pre-January 1984 rules.)

40.7.2 - Increase in Base Premium Amount

(Rev. 1, 09-11-02)

The base premium that is announced by the Secretary is paid by most enrollees. However, it is increased for those who enroll late or reenroll after a prior termination of entitlement. The amount of the increase is 10 percent of the base rate for each full 12-month period during which the individual could have been but was not enrolled in the program. However, months in which an individual has GHP coverage based on current employment status are not counted as months in which the individual "could have been, but was not enrolled." This exception does not apply to individuals who are entitled to Medicare on the basis of ESRD.

40.7.3 - Collection of Premiums

(Rev. 1, 09-11-02)

SMI premiums are collected from benefits payable (in order of priority) by the Railroad Retirement Board, SSA, or Office of Personnel Management (Civil Service) unless the individual is covered under a State buy-in agreement. If the individual is entitled to such benefits but the payments are in suspense, e.g., due to work deductions, he or she is billed for the amounts due. Uninsured beneficiaries are also billed directly for the premiums due.

Premium bills are sent every 3 months unless the individual specifically requests a monthly bill or is also entitled to Premium -HI (in which case a monthly combined SMI-Premium-HI bill is sent).

40.7.4 - Grace Period for Payment

(Rev. 1, 09-11-02)

A grace period has been provided for payment of premiums by those who are billed directly. The period extends for 90 days after the month in which the bill is mailed. If the premiums are not received in that prescribed time, entitlement terminates at the end of the grace period. This 90-day grace period for paying overdue SMI premiums and continuing SMI coverage may be extended by CMS for good cause for up to an additional 90 days. Good cause, for example, is found if the enrollee was mentally or physically incapable of paying his or her premiums timely, or had some reasonable basis to believe that payment had been made, or the failure to pay was due to administrative error.

40.7.5 - Payment After Grace Period Applies

(Rev. 1, 09-11-02)

If an extension to the grace period (additional 90 days; a total not to exceed 180 days) has been granted, an enrollee may retain SMI entitlement by paying all past due premiums if there was a good cause for his/her failure to pay premiums within the initial grace period.

40.7.6 - Premiums Paid by Other Than the Enrollee

(Rev. 1, 09-11-02)

The following subsections describe premiums paid by other than the enrollee.

40.7.6.1 - Informal Arrangement

(Rev. 1, 09-11-02)

Enrollees who are being billed directly for Medicare premiums may turn over their premium bills to a friend, relative, employer lodge, union or other organization to pay premiums on their behalf. The third party payer forwards the proper amount of payment for each enrollee to CMS's Premium Collection Center. Enrollees participating in this type of informal arrangement continue to receive their premium notice (Form CMS-500), and remain responsible for assuring that premiums paid on their behalf are paid timely in order to avoid termination of Medicare coverage.

40.7.6.2 - Premium Payer

(Rev. 1, 09-11-02)

Individuals may be designated as the premium payer for an enrollee and receive the enrollee's premium bills, if it is judged to be in the enrollee's best interest (e.g., the enrollee is competent, but too physically ill or infirm to be able to handle such matters). The individual receiving the premium bill must be a relative or friend showing personal interest in the welfare of the enrollee.

40.7.6.3 - Formal Group Arrangement

(Rev. 1, 09-11-02)

Employers, religious orders, lodges, unions or other organizations may enter into a formal agreement with CMS in order to receive a single bill and pay a lump sum for the premiums due from a group of individuals. Group payments under a formal group agreement may be made only on behalf of individuals who are already enrolled and are being billed for direct remittance. This type of arrangement is referred to as a "Formal Group Payer" arrangement. While included in a formal group payer arrangement, enrollees will no longer receive a premium notice (Form CMS-500). This arrangement is available only where the number of enrollees in the group is large enough to make it practicable to send one bill to the group payer and where the following conditions are also met:

• Enrollees included under the formal group arrangement are directly billed for their premium, and are not having his/her premiums deducted from a Social Security, Railroad Retirement, or Civil Service benefit, or, are not having premiums paid by a State Medicaid agency under a State buy-in agreement;

- The enrollee authorizes the formal group payer to pay premiums on his/her behalf;
- The formal group payer has a minimum of 20 enrollees; and
- The formal group payer agrees to submit premium payments via electronic funds transfer.

If the formal group payer agreement is terminated, or if the group payer alerts CMS to remove an individual from the group payer arrangement, CMS will resume collection of premiums from the individual through the direct billing process.

40.7.6.4 - Premium Surcharge Payment

(Rev. 1, 09-11-02)

States or local Governments may pay a lump sum for the total amount of the SMI late enrollment premium surcharge for a group of individuals. In order to pay the premium surcharge, States or local Governmental entities are required to enter into a formal agreement with CMS. An individual is not billed for the premium surcharge portion of his/her SMI premium while under a premium surcharge agreement, however, he/she is billed for the base premium amount. If the enrollee is receiving a Social Security, Railroad Retirement, or Civil Service annuity, the agencies responsible for these programs will continue to withhold the base premium amount from the annuity. Enrollees who are billed directly for premiums will continue to receive a premium bill for the base premium amount.

The ultimate responsibility for paying premiums rests with the enrollees.

40.7.7 - Premiums Under Buy-In

(Rev. 1, 09-11-02)

Although a person who individually enrolls for SMI may be subject to an increase in the premium rate if he or she fails to enroll when first eligible, a State always pays premiums for its enrollees at the base rate. No premium surcharge for late enrollment is imposed.

40.8 - Waiver of Enrollment Period Requirements Due to Administrative Error

(Rev. 1, 09-11-02)

CMS and SSA are authorized to take necessary action to correct an erroneous SMI or premium hospital insurance enrollment or nonenrollment which was based upon the action, inaction, or error of a Government officer, employee, or agent. The action may include designating special individual enrollment periods and premium adjustments.

For the purpose of this provision, an individual's enrollment or nonenrollment in SMI is considered to have been prejudiced and due to the "action, inaction, or error" of an officer, employee, or agent of the Government, if there is an official record or other evidence showing that:

- The individual took reasonable appropriate and timely measures to assert his or her rights, and
- Due to administrative fault or other action, which may or may not have been erroneous at the time taken, his or her rights have been or are likely to be impaired unless relief is given.

This authority applies to all Part B cases which have arisen since the Medicare program began on July 1, 1966. It is applied to cases that come to CMS's or SSA's attention; however, intermediaries and carriers should not search their files.

50 - Identifying the Patient's Health Insurance Record Using the Health Insurance Card (Rev. 1, 09-11-02)

As part of health insurance electronic data processing, health insurance cards are issued by CMS (or by the RRB where railroad retirement beneficiaries are involved) to individuals who have established entitlement to health insurance. An example of the Health Insurance card is found at:

http://www.medicare.gov/Basics/ymc.asp

The HI card is used to identify the individual as being entitled and also serves as a source of information required to process Medicare claims or bills. It displays the beneficiary's name, sex, Health Insurance Claim Number, and effective date of entitlement to hospital insurance and/or medical insurance.

The Social Security Administration's Social Security Office assists in replacing a lost or destroyed HI card.

50.1 - Temporary Eligibility Notice

(Rev. 1, 09-11-02)

A Social Security field office may issue a temporary health insurance eligibility notice if medical services are needed immediately. Sample text for such notices follows:

TEMPORARY NOTICE OF MEDICARE ELIGIBILITY

((If the individual is oi	ly eligible for HI	or SMI, delete the	e inapplicable words

-	. •		• .	_	\	30	•		- 1	1		
1	110	tr	101	. (Н	-+-	ice	Λ	$^{\prime}$	പ	20C	0
		1.1	ıwı	٠.	,,			$\overline{}$	u	ш		

Date:

Patients HICN

Dear

Based on the information given to the Social Security Administration, you are (Mr./Ms. is) eligible for hospital insurance beginning (mo.) (yr.) and for medical insurance beginning (mo.) (yr.) . This notice will serve as evidence of your (his/her) eligibility for these benefits for 60 days from the date shown at the top of this notice, unless you are notified otherwise during the 60-day period.

To obtain medical services (or payment for medical services) before you receive a health insurance card, show this letter to your hospital or doctor, but keep the letter with you. This temporary notice of eligibility is to be used only by the person to whom it is addressed. Misuse is unlawful and will make the offender liable to a penalty.

This letter should be destroyed as soon as you receive a health insurance card or other notice of eligibility.

Sincerely Yours,

Commissioner of Social Security

IMPORTANT: When services are provided on the basis of this notice, all bills or correspondence with a carrier, intermediary, or the Social Security Administration should show the patient's health insurance claim number.

50.2 - Health Insurance Claims Numbers (HICNs)

(Rev. 1, 09-11-02)

All HICNs issued by SSA are 9-digit numbers with at least one letter suffix (called a beneficiary identification code or BIC) in the tenth position. If there is an eleventh position, it may be either a letter or number e.g., 123-45-6789A or 987-65-4321D4. The HICN issued by the RRB, may contain either 6 or 9 digit numbers with up to a 3-position letter prefix e.g., A123456 or MA123-45-6789. If a beneficiary's entitlement changes, it is possible for the 9-digit number, the prefix, the suffix or all three to change. It is also possible to go from an SSA issued HICN to a RRB HICN or vise versa.

The numeric portion of a 9-digit HICN consists of a Social Security Number (SSN). If the BIC is A, T, TA, M, M1, J1, J2, J3, J4 or the RRB prefix is A or H the number is the beneficiary's own SSN. If the BIC or RRB prefix is other than one of the above, the SSN belongs to a number holder and the beneficiary is entitled as an auxiliary or survivor on that SSN.

Currently, the first three digits of the HICN range from 001-772. However this may change as SSA issues more numbers. All numbers except 00 are possible for the fourth and fifth digits and all numbers except 0000 are possible for the last four digits.

The patient's HICN is on his/her HI card, SSA award letter, SSA Benefit Verification letter, an SSA issued Temporary Notice of Eligibility, or Medicare Summary Notice (MSN). Where the patient cannot furnish a HICN, it may be an indication that he/she has not filed an application

with SSA to establish entitlement to health insurance benefits, or that SSA action on a pending application has not been completed.

50.3 - HICNs Assigned by CMS

(Rev. 1, 09-11-02)

(See Section 50.2 for an explanation of the valid 9-digit numbers issued by SSA.)

Α

B, Bl, B2, B3, B4, B5, B6, B7, B8, B9, BA, BD, BG, BH, BJ, BK, BL, BN, BP, BQ, BR, BT, BW, BY

C1, C2, C3, C4, C5, C6, C7, C8, C9, CA, CB, CC, CD, CE, CF, CG, CH, CI, CJ, CK, CL, CM, CN, CO, CP, CQ, CR, CS, CT, CU, CV, CW, CX, CY, CZ

D, Dl, D2, D3, D4, D5, D6, D7, D8, D9, DA, DC, DD, DG, DH, DJ, DK, DL, DM, DN, DP, DQ, DR, DS, DT, DV, DW, DX, DY, DZ

E, El, E2, E3, E4, E5, E6, E7, E8, E9, EA, EB, EC, ED, EF, EG, EH, EJ, EK, EM

F1, F2, F3, F4, F5, F6, F7, F8

J1, J2, J3,J4

K1, K2, K3, K4, K5, K6, K7, K8, K9, KA, KB, KC, KD, KE, KF, KG, KH, KJ, KL, KM

T, TA, TB, TC, TD, TE, TF, TG, TH, TJ, TK, TL, TM, TN, TP, TQ, TR, TS, TT, TU, TV, TW, TX, TY, TZ, and T2

W, Wl, W2, W3, W4, W5, W6, W7, W8, W9, WB, WC, WF, WG, WJ, WR, WT

50.4 - HICNs Assigned by the RRB

(Rev. 1, 09-11-02)

The RRB began using the social security number in their numbering system during calendar year 1964. The HICNs assigned prior to that time were 6-digit numbers assigned in numerical sequence and had no special characteristics. However, both the 6-digit numbers and the 9-digit social security numbers when used as claim numbers by the RRB always have letter prefixes. In rare cases, a qualified railroad retirement beneficiary may have a claim number with less than 6-digits. In this case, sufficient zeros are added between the prefix and other digits to make a 6-digit number, e.g., WD-001234. The current range of valid RRB claim numbers is 000001-994999.

50.4.1 – Six-Digit Numbers

(Rev. 1, 09-11-02)

The basic RRB claim numbers assigned to each type of prefix are shown in this section. Under the RRB system, it is permissible for two beneficiaries to have identical claim numbers. For example, when a widower remarries, the second wife is assigned the same claim number that was assigned to the first wife. Under the Medicare program, however, every individual has a distinctive claim number. Therefore, for Medicare purposes, pseudo numbers are assigned to railroad retirement beneficiaries who would otherwise have a claim number that was assigned to someone else.

The numbers in the series 995000 through 999999 were assigned to these beneficiaries. But, whenever possible, the Board will use the railroad retirement beneficiary's own 9-digit social security number with the appropriate prefix. They will only use the 6-digit number if the railroad retirement beneficiary does not have their own social security number and cannot obtain one because of Social Security Administration limitations on issuing numbers. An example of an individual who cannot get a number is a beneficiary who lives outside the United States and is not a citizen of the U.S.

50.4.2 - Valid RRB HICNs

(Rev. 1, 09-11-02)

A-000000	WD-000000	PD-000000
A-000-00-0000	WD-000-00-0000	PD-000-00-0000
MA-000000	WCD-000000	H-000000
MA-000-00-0000	WCD-000-00-0000	MH-000000
WA-000000	WCA-000000	PH-000000
WA-000-00-0000	WCA-000-00-0000	JA-000000
CA-000000	PA-000000	WH-000000
CA-000-00-0000	PA-000-00-0000	WH-000-00-0000
H-000-00-0000	MH-000-00-0000	WCH-000000
JA-000-00-0000	PH-000-00-0000	WCH-000-00-0000

60 - Medicare Part C, Medicare+Choice

(Rev. 1, 09-11-02)

Medicare Part C, which is also known as Medicare+Choice, is a Medicare program that gives beneficiaries more choices among health plans. Everyone who has Medicare Parts A and B is eligible, except those who have end stage renal disease, and were not in a Medicare+Choice plan at the onset of this condition.

Public Law 105-33, the Balanced Budget Act of 1997, establishes a new authority permitting contracts between CMS and a variety of different Medicare+Choice organizations and fee-for-service entities. The types of entities that may be granted contracts under this new authority include:

- Coordinated care plans, including Health Maintenance Organizations (HMOs), Preferred
 Provider Organizations (PPOs), and Provider-Sponsored Organizations (PSOs). A PSO is
 defined as a public or private entity established by health care providers, which provide a
 substantial proportion of health care items and services directly through affiliated providers
 who share, directly or indirectly, substantial financial risk.
- Religious fraternal benefit society plans which may restrict enrollment to members of the church, convention or group with which the society is affiliated. Payments to such plans may be adjusted, as appropriate to take into account the actuarial characteristics and experience of plan enrollees.
- Private fee-for-service plans which reimburse providers on a fee-for-service basis, and are authorized to charge enrolled beneficiaries up to 115 percent of the plan's payment schedule (which may be different from the Medicare fee schedule).

In addition to the Medicare+Choice contractors, beginning in January, 1999, up to 390,000 beneficiaries will have the choice (on a demonstration basis ending January 1, 2003) of enrolling in a Medical Savings Account (MSA) option. Under this option, beneficiaries would obtain high deductible health policies that pay for at least all Medicare-covered items and services after an enrollee meets the annual deductible of up to \$6,000. The difference between the premiums for such high deductible policies and the applicable Medicare+Choice premium amount would be placed into an account for the beneficiary to use in meeting his or her deductible expenses.

Past 1876 Contracts - HMO/CMP risk plans that remain in compliance with current contracting standards and comply with new requirements established under this statutory authority automatically transitioned into the Part C Medicare+Choice program. Section 1876 risk-based contractors were paid under a new Medicare+Choice payment methodology rather than the prior method in section 1876(a), and were subject to certain other Medicare+Choice provisions. The Secretary no longer accepts new 1876 risk applications. As of January 1, 1999, existing 1876 risk-based contracts were terminated, and plans in good standing transitioned to the Medicare+Choice program.

Repeal of Cost Option - As of August 5, 1997, the Secretary is prohibited from entering into any new 1876 cost-based contracts, unless the plan is a Health Care Prepayment Plan with an agreement under section 1833 of the Social Security Act. The 1876 cost-based payment authority is repealed and all cost contracts are terminated as of December 31, 2002.

Limited HCPP Option - Beginning January 1, 1999, the Secretary may only contract with those HCPPs that are sponsored by Union or Employer groups, or HCPPs that do not "provide, or

arrange for the provision of, any inpatient hospital services" This amendment will result in the termination of Section 1833 agreements with any organization that does not meet the new definition. CMS will establish transition rules for Section 1876 risk-based contractors that currently receive reimbursement on a cost basis for enrollees remaining under a previous HCPP agreement.

(For more information, see the Managed Care Manual.)

Medicare General Information, Eligibility, and Entitlement

Chapter 3 - Deductibles, Coinsurance Amounts, and Payment Limitations

Table of Contents

10 - Hospital Insurance (F	Part A)
----------------------------	---------

- 10.1 Inpatient Hospital Deductible
- 10.2 Coinsurance
 - 10.2.1 Inpatient Services
 - 10.2.2 Durable Medical Equipment (DME) Furnished as a Home Health Benefit
- 10.3 Basis for Determining the Part A Coinsurance Amounts
- 10.4 Benefit Period (Spell of Illness)
 - 10.4.1 Starting a Benefit Period
 - 10.4.2 Ending a Benefit Period
 - 10.4.3 Definition of Hospital or SNF for Ending a Benefit Period
 - 10.4.3.1 Hospital Stay and End of Benefit Period
 - 10.4.3.2 SNF Stay and End of Benefit Period
 - 10.4.4 Definition of Inpatient for Ending a Benefit Period

20 - Supplementary Medical Insurance (Part B)

- 20.1 Supplementary Medical Insurance Incurred Expenses
- 20.2 Part B Annual Deductible
- 20.3 Part B Coinsurance
- 20.4 Exceptions to Annual Deductible and Coinsurance

- 20.4.1 Applications of Deductible and Coinsurance in Liability and Indemnification Situations
- 20.5 Blood Deductibles (Part A and Part B)
 - 20.5.1 Part A Blood Deductible
 - 20.5.2 Part B Blood Deductible
 - 20.5.3 Items Subject to Blood Deductibles
 - 20.5.4 Obligations of the Beneficiary to Pay for or Replace Deductible Blood
 - 20.5.4.1 Replacement of Blood
- 30 Outpatient Mental Health Treatment Limitation
 - 30.1 Application of Mental Health Limitation Status of Patient
 - 30.2 Disorders Subject to Mental Health Limitation
 - 30.3 Diagnostic Services
- 40 Limitation on Services of Physical or Occupational Therapist in Independent Practice

10 - Hospital Insurance (Part A)

10.1 - Inpatient Hospital Deductible

(Rev. 1, 09-11-02)

The patient is responsible for a deductible amount for inpatient hospital services in each benefit period. For each year after 1991, the Secretary of the Department of Health and Human Services (DHHS) is required to set the deductible and coinsurance amounts between September 1 and September 15 of the preceding year. The deductible will be set at an amount equal to the deductible for the preceding year, changed by the same percentage as applies to PPS payment rates, and adjusted to reflect changes in real case mix. The deductible and coinsurance amounts are shown in the chart in §10.3 of this chapter.

The coinsurance amount is based on the deductible applicable for the calendar year in which the coinsurance days occur. The deductible is satisfied only by charges for covered Part A services. Expenses for covered services count toward the deductible on an incurred, rather than paid, basis. Expenses incurred in one benefit period cannot be applied toward the deductible in a later benefit period. Expenses incurred in meeting the blood deductible do not count toward the inpatient hospital deductible.

A reduction in benefit days resulting from confinement in a psychiatric hospital, on and immediately preceding the date of entitlement, does not affect the amount of the deductible for which the patient is responsible.

If the actual charge is less than the deductible and the customary charge, the customary charge is applied to the deductible.

A beneficiary is not responsible for payment of the deductible for an inpatient stay if the provider has been determined to be liable because the care was not medically necessary or because the care provided was custodial.

10.2 - Coinsurance

(Rev. 1, 09-11-02)

The following subsections describe coinsurance.

10.2.1 - Inpatient Services

(Rev. 1, 09-11-02)

In each benefit period, the patient is responsible for coinsurance amounts equal to:

- One-fourth of the inpatient hospital deductible for each day of inpatient hospital services from the 61st through the 90th days;
- One-half of the inpatient hospital deductible for each lifetime reserve day (the 91st through the 150th days of inpatient hospital services); and
- One-eighth of the inpatient hospital deductible for each day of extended care services from the 21st through the 100th days. A beneficiary is not responsible for payment of the coinsurance for a stay if the provider has been determined to be liable because the care was not medically necessary or because the care provided was custodial.

Use the chart in §10.3 of this chapter to determine the applicable coinsurance amounts.

Where the actual charge to the patient for the 61st through the 90th days of inpatient hospital services is less than the applicable coinsurance amount, the coinsurance is the actual charge per day. Where the actual charge to the patient for lifetime reserve days is less than the coinsurance amount for those days, the beneficiary may be deemed to have elected not to use the days because he/she would not benefit from their utilization.

10.2.2 - Durable Medical Equipment (DME) Furnished as a Home Health Benefit (Rev. 1, 09-11-02)

The patient is responsible for 20 percent of the payment amount for DME furnished as a home health benefit.

10.3 - Basis for Determining the Part A Coinsurance Amounts

(Rev. 1, 09-11-02)

The applicable inpatient deductible is the one in effect during the calendar year in which the patient's benefit period begins (i.e., in most cases, the year in which the first inpatient hospital services are furnished in the benefit period). Except for 1989, the coinsurance amount is based on the deductible applicable for the calendar year in which the coinsurance days occur.

Applicability of Part A Deductible and Coinsurance

Inpatient Hospital- First 60 Days	Deductible
Inpatient Hospital- 61st thru 90th Day	Coinsurance per day always equal to 1/4 of inpatient hospital deductible
Inpatient Hospital- 60 Lifetime Reserve Days (nonrenewable) - 91st thru 150th day	Coinsurance always equal to ½ of inpatient hospital deductible
Skilled Nursing Facility - 21st thru 100th Day	Coinsurance Always equal to 1/8 of inpatient hospital deductible
Home Health Agency	No Deductible
	No Coinsurance (except for 20% coinsurance for DME and prosthetics/ orthotics)
Blood	1st 3 pints (or equivalent units of packed red blood cells) combined Part A and Part B annual deductible
Hospice a. Drugs and Biologicals	a. 5 percent of the cost determined by the drug copayment schedule (may not exceed \$5 per prescription)
b. Respite Care	b. 5 percent of the payment for a respite care day

Hospices may charge coinsurance for two services only, drugs and biologicals, and respite care. The amount of coinsurance for each prescription may not exceed \$5.00. The amount for respite care may not exceed the inpatient deductible for the year in which the hospital coinsurance period began.

Deductible and Coinsurance Amounts

Year	Inpatient Hospital, 1st 60 Days	Inpatient Hospital, 61st-90th Days	60 Lifetime Reserve Days	Skilled Nursing Facility
1986	\$492	123	246	61.50
1987	520	130	260	65.00
1988	540	135	270	67.50
1989	560	0 (1)	0 (1)	0(2)
1990	592	148	296	74.00
1991	628	157	314	78.50
1992	652	163	326	81.50
1993	676	169	338	84.50
1994	696	174	348	87.00
1995	716	179	358	89.50
1996	736	184	368	92.00
1997	760	190	380	92.00
1998	764	191	382	95.50
1999	768	192	384	96.00
2000	776	194	388	97.00
2001	792	198	396	99.00
2002	812	203	406	101.50

- 1. Coinsurance was not charged for inpatient hospital care in CY 1989 due to Catastrophic Coverage. The deductible was applied.
- 2. Under Catastrophic Coverage, a coinsurance payment of \$25.50 was due for days 1-8 of SNF care. No SNF coinsurance was due after day 8 in 1989.

10.4 - Benefit Period (Spell of Illness)

(Rev. 1, 09-11-02)

A benefit period is a period of time for measuring the use of hospital insurance benefits. It is a period of consecutive days during which covered services furnished to a patient, up to certain specified maximum amounts, may be paid for by the hospital insurance plan. For example, a patient is eligible for 90 days of hospital care in a benefit period and 100 days of extended care services during the same benefit period. A patient may be eligible for as many as 150 days of hospital care in a benefit period if he/she draws on his/her lifetime reserve. As long as a person continues to be entitled to hospital insurance, there is no limit on the number of benefit periods he/she may have. The term "benefit period" is synonymous with spell of illness. Since the term "spell of illness" could connote a single illness or a particular "spell" of sickness, the term benefit period is used in communications with the public.

10.4.1 - Starting a Benefit Period

(Rev. 1, 09-11-02)

A benefit period begins with the first day (not included in a previous benefit period) on which a patient is furnished inpatient hospital or extended care services by a qualified provider in a month for which the patient is entitled to hospital insurance benefits.

A provider qualified to start a benefit period is a hospital (including a psychiatric hospital) or SNF that meets all the requirements of the definition of such an institution. A hospital which meets all requirements in chapter 5, §20 of this manual is also a qualified hospital for purposes of beginning a benefit period when it furnishes the patient covered inpatient emergency services. Thus, generally, the benefit period begins when covered inpatient services are initially furnished to an entitled individual. However, the noncovered services furnished by a nonparticipating provider can begin a spell of illness only if the provider is a qualified provider. A qualified provider is a hospital (including a psychiatric hospital) or a SNF which meets all requirements in the definition of such an institution even though it may not be participating. A qualified hospital in Canada or Mexico is also a qualified provider for purposes of beginning a benefit period when it furnishes covered inpatient hospital services. If a person is in a nonqualified institution and is subsequently transferred to a qualified hospital (general or psychiatric), his/her benefit period begins on admission to the qualified hospital.

Admission to a qualified SNF or to the SNF level of care in a swing bed hospital begins a benefit period even though payment for the services cannot be made because the prior hospitalization or transfer requirement has not been met. Inpatient care in a Religious Non-Medical Health Care Institution (whether as hospital or extended care services) can begin or prolong a benefit period.

10.4.2 - Ending a Benefit Period

(Rev. 1, 09-11-02)

The benefit period ends with the close of a period of 60 consecutive days during which the patient was neither an inpatient of a hospital nor of a SNF. To determine the 60 consecutive day

period, begin counting with the day the individual was discharged. (See §10.4.3.2 of this chapter for determining the end of a benefit period when an individual remains in a SNF.)

10.4.3 - Definition of Hospital or SNF for Ending a Benefit Period

(Rev. 1, 09-11-02)

It is important to note that a benefit period cannot end while a beneficiary is an inpatient of a hospital, even if the hospital does not meet all of the requirements that are necessary for starting a benefit period. Similarly, a benefit period cannot end while a beneficiary is an inpatient of a SNF, as defined below.

10.4.3.1 - Hospital Stay and End of Benefit Period

(Rev. 1, 09-11-02)

In order to end a benefit period, for at least 60 consecutive days, a beneficiary cannot have been in a hospital which meets the initial requirement in the definitions in chapter 5, §20.1 through §20.7 of this manual. That is, the beneficiary cannot have been in a facility that is primarily engaged in providing, by or under the supervision of a physician(s), to inpatients:

- Diagnostic and therapeutic services for medical diagnosis, treatment, and care of injured, disabled, or sick persons or rehabilitation services for injured, disabled, or sick persons; or
- Psychiatric services for the diagnosis and treatment of mentally ill persons.

A stay in a hospital outside the United States prolongs a benefit period. It may be assumed, in the absence of evidence to the contrary, that:

- A foreign hospital in which a beneficiary spent one or more days meets the requirement of the definition in chapter 5, §20 of this manual; and
- The beneficiary's statement about length and place of stay is correct.

10.4.3.2 - SNF Stay and End of Benefit Period

(Rev. 1, 09-11-02)

Similarly, to end a benefit period, a beneficiary cannot have been an inpatient (see subsection 10.4.4) of a SNF for at least 60 consecutive days; where SNF is defined as a facility which is primarily engaged in providing skilled nursing care and related services to residents who require medical or nursing care, or rehabilitation services for the rehabilitation of injured, disabled, or sick persons.

Examples: An individual may be discharged from and readmitted to a hospital or SNF several times during a benefit period and still be in the same benefit period if 60 consecutive days have not elapsed between discharge and readmission. The stays need not be for related physical or mental conditions.

Example I: X was born 8/9/36. On 7/28/2001, X entered a participating general hospital. After he or she had been in the hospital for 2 weeks, X was discharged on 8/11/2001. On his/her doctor's orders, X entered a participating SNF on 8/15/2001, and remained an inpatient there (see subsection 10.4.4) until his/her discharge on 10/27/2001. He or she had no further inpatient stays in 2001.

X's benefit period began on 8/1/2001, the first day of the month he or she attained age 65 and was entitled to hospital insurance. The benefit period ended 12/25/2001, the end of the 60-day period beginning with the date of his/her last discharge.

Example 2: Y, over age 65, entered a participating general hospital on 8/28/2000 for treatment of a heart condition. He or she was discharged on 9/11/2000. On 10/3/2000, Y entered a Medicaid-only nursing facility, and remained an inpatient of this facility (see subsection 10.4.4) until his/her discharge on 11/17/2000. On 12/26/2000, Y was again admitted to a participating hospital because of injuries suffered in an accident. He or she was discharged on 1/13/2001 and had no further inpatient stays in 2001.

Y's benefit period began on 8/28/2000. His/her stay in the nursing facility began less than 60 days after his/her hospital stay and the benefit period was continued because he or she remained an inpatient there (see subsection 10.4.4) even though Medicare did not cover the stay. The subsequent hospital stay began less than 60 days after the nursing facility stay and continued the benefit period although the condition treated was unrelated to his/her prior stays. The benefit period ended on 3/14/2001, the end of the 60-day period beginning with the day of last discharge.

Example 3: Z, over age 65 and entitled to hospital insurance benefits, was admitted to General Hospital on 8/l/2000 and discharged on 8/l0/2000, having received nonemergency hospital services. General Hospital met all the requirements in the definition of a hospital except those concerning utilization review and health and safety. While General Hospital met the minimum requirements of an emergency hospital, Z's benefit period did not begin with his/her admission to this hospital because:

- 1. The hospital did not meet all of the requirements in the definition of a hospital; and
- 2. Although the hospital satisfied the minimum requirements for coverage of emergency services, Z did not receive emergency inpatient care there.

(As noted previously, a stay in an emergency hospital does not begin a benefit period unless it actually involves the receipt of covered inpatient emergency services; by contrast, even a nonemergency stay in such a hospital can serve as a qualifying hospital stay for purposes of coverage under the posthospital extended care benefit.) Z was admitted to Haven Convalescent Home on 8/20/2000 and remained an inpatient of the home (see subsection 10.4.4) until his/her discharge on 3/1/2001. He or she had no further inpatient stays in 2001. Haven Convalescent Home became a participating SNF on 1/1/2001.

Z's benefit period began 1/1/2001, the day Haven Convalescent Home was determined to be a qualified SNF. The services Z received from that date through discharge were extended care services even though they were not covered and, therefore, not charged against Z's Medicare SNF utilization. (The services were not covered posthospital extended care services because Z

was not admitted to a participating SNF within 30 days after discharge from the hospital.) Z's benefit period ended 4/29/2001, the end of the 60-day period beginning with the date of his/her discharge from the convalescent home.

10.4.4 - Definition of Inpatient for Ending a Benefit Period

(Rev. 1, 09-11-02)

Generally, a beneficiary is an inpatient of a hospital if the beneficiary is receiving inpatient services in the hospital (i.e., not on an outpatient basis). The type of care actually received is not relevant.

However, a different definition of inpatient applies in determining the end of a benefit period for a beneficiary in a SNF. A beneficiary is an inpatient in a SNF only if the beneficiary's care in the SNF meets certain skilled level of care standards. The beneficiary must need and receive a skilled level of care while in the SNF. This means that in order to have been an inpatient while in a SNF, the beneficiary must have required and received skilled services on a daily basis which could, as a practical matter, only have been provided in a SNF on an inpatient basis. If these provisions were not met during the prior SNF stay, the beneficiary was not an inpatient of the SNF for purposes of prolonging the benefit period.

Use the following presumptions for determining whether the skilled level of care standards were met during a prior SNF stay.

Presumption 1: A beneficiary's care in a SNF met the skilled level of care standards if a Medicare SNF claim was paid for the care, unless such payment was made under limitation on liability rules.

Presumption 2: A beneficiary's care in a SNF met the skilled level of care standards if a SNF claim was paid for the services provided in the SNF under the special Medicare limitation on liability rules pursuant to placement in a non-certified bed.

Presumption 3: A beneficiary's care in a SNF did not meet the skilled level of care standards if a claim was paid for the services provided in the SNF pursuant to the general Medicare limitation on liability rules. (This presumption does not apply to placement in a non-certified bed. For claims paid under these special provisions, see Presumption 2.)

Presumption 4: A beneficiary's care in a Medicaid nursing facility (NF) did not meet the skilled level of care standards if a Medicaid claim for the services provided in the NF was denied on the grounds that the services received were not at the NF level of care (even if paid under applicable Medicaid administratively necessary days provisions which result in payment for care not meeting the NF level of care requirements).

Presumption 5: A beneficiary's care in a SNF met the skilled level of care standards if a Medicare SNF claim for the services provided in the SNF was denied on grounds other than that the services were not at the skilled level of care.

Presumption 6: A beneficiary's care in a SNF did not meet the skilled level of care standards if a Medicare claim for the services provided in the SNF was denied on the grounds that the services were not at the skilled level of care and no limitation of liability payment was made.

Presumption 7: A beneficiary's care in a SNF did not meet the skilled level of care standards if no Medicare or Medicaid claim was submitted by the SNF.

Presumptions 1 through 4 cannot be rebutted. Thus, prior Medicare and Medicaid claim determinations that necessarily required a level of care determination for the time period under consideration are binding for purposes of a later benefit period calculation. Presumptions 5 through 7 can be rebutted by the beneficiary showing that the level of care needed or received is other than that which the presumption dictates.

Presumption 6 can be rebutted because the Medicare skilled level of care definition for coverage purposes is broader than the skilled level of care definition used here for benefit period determinations. Specifically, the requirement referred to in Chapter 4, §40.2 regarding prior hospital care related to the SNF care is included in the Medicare SNF coverage requirements but is not included in the standard for benefit period determinations. Therefore, Medicare payment could have been denied for a SNF stay on level of care grounds (i.e., not even waiver payment was made) because of noncompliance with that requirement, even though skilled level of care requirements for benefit period determinations were in fact met by the SNF stay. Consequently, when Medicare SNF payment is denied on level of care grounds, the beneficiary must be given the opportunity to demonstrate that he/she still needed and received a skilled level of care for purposes of benefit period determinations.

NOTE: Effective October 1, 1990, the levels of care that were previously covered separately under the Medicaid SNF and intermediate care facility (ICF) benefits are combined in a single Medicaid nursing facility (NF) benefit. Thus, the Medicaid NF benefit includes essentially the same type of skilled care covered by Medicare's SNF benefit, but it includes less intensive care as well. This means that when a person is found not to require at least a Medicaid NF level of care (as under Presumption 4), it can be presumed that he or she also does not meet the Medicare skilled level of care standards. However, since the NF benefit can include care that is less intensive than Medicare SNF care, merely establishing that a person does require NF level care does not necessarily mean that he or she also meets the Medicare skilled level of care standards. Determining whether an individual who requires NF level care also meets the Medicare skilled level of care standards requires an actual examination of the medical evidence and cannot be accomplished through the simple use of a presumption. Therefore, the previous references to Medicaid claims have been deleted from those presumptions which establish that an individual does meet the Medicare standards.

Medicare no-payment bills submitted by a SNF result in Medicare program payment determinations (i.e., denials). Therefore, such no-payment bills trigger the appropriate presumptions. This also applies in any State where the Medicaid program utilizes no-payment bills which lead to Medicaid program payment determinations. If a SNF erroneously fails to submit a Medicare claim (albeit a no-pay claim) when Medicare rules require such submission, intermediaries request a SNF to submit one. Once the no-pay bill is submitted and denied, the applicable presumption (other than presumption 7) is triggered. If a patient is moving from a SNF level of care to a non-SNF level of care in a facility certified to provide SNF care,

occurrence code 22 (date active care ended) is used to signify the beginning of the no-pay period on the bill and trigger the appropriate presumptions.

Where the presumptions are rebuttable (i.e., 5 through 7), rebuttal showings are permitted at both intermediary determination levels under 42 CFR 405, Subpart G (i.e., a rebuttal showing regarding the status of a prior SNF stay is made at the time that an inpatient claim is submitted and/or at the reconsideration level). Intermediaries evaluate rebuttal documentation even if the presumption being rebutted was triggered by a Medicaid denial.

This special rule for determining whether a beneficiary in a SNF is an inpatient for benefit period purposes is applicable in all cases where a prior SNF stay affects benefit period status, not only when a beneficiary is in exhausted or copay status and is seeking to renew a benefit period. The rule has equal application where it results in the beneficiary starting a new benefit period and paying a new deductible without receiving an increase in the amount of Medicare benefits paid.

20 - Supplementary Medical Insurance (Part B)

(Rev. 1, 09-11-02)

Supplementary Medical Insurance is described in the following subsections.

20.1 - Supplementary Medical Insurance Incurred Expenses

(Rev. 1, 09-11-02)

The SMI plan includes coverage for expenses incurred for the services described in Chapter 1, §10.3:

Payment may not be made under Part B for services furnished an individual entitled to have payment made for those services under Part A, e.g., if the expenses incurred were to satisfy a Part A deductible or coinsurance amount, or if payment would be made under Part A except for the lack of request for payment or physician certification.

20.2 - Part B Annual Deductible

(Rev. 1, 09-11-02)

In each calendar year, a cash deductible must be satisfied before payment can be made under SMI. (See 20.4 of this chapter for exceptions.)

- As of January 1, 1991, the deductible is \$100.
- From 1982 through 1990, the deductible was \$75.
- From 1973 through 1981, the deductible was \$60.
- From 1966 through 1972, the deductible was \$50.

Expenses count toward the deductible on the basis of incurred, rather than paid expenses, and are based on Medicare allowed amounts. Noncovered expenses do not count toward the deductible.

Even though an individual is not entitled to Part B benefits for the entire calendar year (i.e., insurance coverage begins after the first month of a year or the individual dies before the last month of the year), he or she is still subject to the full deductible for that year. Medical expenses incurred in the portion of the year preceding entitlement to medical insurance are not credited toward the deductible.

The date of service generally determines when expenses were incurred, but expenses are allocated to the deductible in the order in which the bills are received. Services which are not subject to the deductible cannot be used to satisfy the deductible.

20.3 - Part B Coinsurance

(Rev. 1, 09-11-02)

After the Part B deductible has been satisfied, coinsurance of 20 percent is usually applicable.

For providers and suppliers that bill intermediaries, the 20 percent may be based on the allowed amount, billed charges, or a preset rate per service (APC), depending upon the type of service. See Claims Processing instructions for a description of coinsurance calculation for each benefit type.

Physicians and other suppliers will be paid 80 percent of the allowed amount under the fee schedule amounts or in some instances reasonable charges incurred during the balance of the calendar year. The patient is responsible for a coinsurance amount equal to 20 percent of the fee schedule amounts or reasonable charges for the items and services. (See §20.4 of this chapter for exceptions.)

20.4 - Exceptions to Annual Deductible and Coinsurance

(Rev. 1, 09-11-02)

There is no deductible for screening mammography effective for services January 1, 1998 and later

Neither the annual deductible nor the 20 percent coinsurance apply with respect to:

- Inpatient hospital radiology or pathology physician services only if the physician agreed to accept assignment for all radiology and pathology services furnished to inpatients;
- Parts A and B home health services, except that there is a coinsurance of 20 percent of the payment amount for supplies, drugs, DME and prosthetics /orthotics furnished as a home health benefit;
- Clinical diagnostic laboratory tests (including specimen collection fees) performed or supervised by a physician, laboratory, or other entity paid on an assigned basis;
- Certain surgical procedures performed by a physician in an ambulatory surgical center or outpatient department of a hospital;
- Ambulatory surgical services in an ambulatory surgical center;

- Pneumococcal vaccine and its administration;
- Influenza vaccine and its administration; and
- Services or items denied as medically unnecessary.

NOTE: Services which are not subject to the deductible cannot be used to satisfy the deductible.

NOTE: There is not therapy cap through 2002.

20.4.1 - Applications of Deductible and Coinsurance in Liability and Indemnification Situations

(Rev. 1, 09-11-02)

Under 1879 of the Act, a beneficiary is not responsible for payment of the Part B deductible or coinsurance for items or services that are neither reasonable and necessary to diagnose or treat the illness or injury, nor to improve the functioning of a malformed body member. If the provider knew, or should have known, that Medicare considered such services medically unnecessary, but failed to inform the beneficiary before furnishing them, the provider is held liable for their cost. If the beneficiary made payment for such items or services, he/she can be indemnified for them.

20.5 - Blood Deductibles (Part A and Part B)

(Rev. 1, 09-11-02)

Program payment may not be made for the first 3 pints of whole blood or equivalent units of packed red cells received under Part A and Part B combined in a calendar year. However, blood processing (e.g., administration, storage) is not subject to the deductible.

The blood deductibles are in addition to any other applicable deductible and coinsurance amounts for which the patient is responsible.

The deductible applies only to the first 3 pints of blood furnished in a calendar year, even if more than one provider furnished blood.

20.5.1 - Part A Blood Deductible

(Rev. 1, 09-11-02)

Blood must be furnished on a Medicare covered day in a hospital or SNF to be counted under Part A. Blood furnished to an inpatient after benefits exhausted or before entitlement is not counted toward the combined deductible. Blood furnished during a lifetime extension election period is counted toward the combined A/B 3 pint total.

20.5.2 - Part B Blood Deductible

(Rev. 1, 09-11-02)

Blood is furnished on an outpatient basis is subject to the Part B blood deductible and is counted toward the combined limit. It should be noted that payment for blood may be made to the hospital under Part B only for blood furnished in an outpatient setting. Blood is not covered for inpatient Part B services.

20.5.3 - Items Subject to Blood Deductibles

(Rev. 1, 09-11-02)

The blood deductibles apply only to whole blood and packed red cells. The term whole blood means human blood from which none of the liquid or cellular components have been removed. Where packed red cells are furnished, a unit of packed red cells is considered equivalent to a pint of whole blood. Other components of blood such as platelets, fibrinogen, plasma, gamma globulin, and serum albumin are not subject to the blood deductible. However, these components of blood are covered as biologicals.

20.5.4 - Obligations of the Beneficiary to Pay for or Replace Deductible Blood

(Rev. 1, 09-11-02)

A provider may charge the beneficiary or a third party its customary charge for whole blood or units of packed red cells which are subject to the blood deductible, unless the individual, another person, or a blood bank replaces the blood or arranges to have it replaced.

20.5.4.1 - Replacement of Blood

(Rev. 1, 09-11-02)

For replacement purposes, a pint of whole blood is considered equivalent to a unit of packed red cells. A deductible pint of whole blood or unit of packed red cells is considered replaced when a medically acceptable pint or unit is given or offered to the provider or, at the provider's request, to its blood supplier. Accordingly, where an individual or a blood bank offers blood as a replacement for a deductible pint or unit furnished a Medicare beneficiary, the provider may not charge the beneficiary for the blood, whether or not the provider or its blood supplier accepts the replacement offer. Thus a provider may not charge a beneficiary merely because it is the policy of the provider or its blood supplier not to accept blood from a particular source which has offered to replace blood on behalf of the beneficiary. However, a provider would not be barred from charging a beneficiary for deductible blood, if there is a reasonable basis for believing that replacement blood offered by or on behalf of the beneficiary would endanger the health of a recipient or that the prospective donor's health would be endangered by making a blood donation. Once a provider accepts a pint of replacement blood from a beneficiary or another individual acting on his/her behalf, the blood is deemed to have been replaced, and, the beneficiary may not be charged for the blood, even though the replacement blood is later found to be unfit and has to be discarded.

When a provider accepts blood donated in advance, in anticipation of need by a specific beneficiary, whether the beneficiary's own blood, that is, an autologous donation, or blood

furnished by another individual or blood assurance group, such donations are considered replacement for pints or units subsequently furnished the beneficiary.

30 - Outpatient Mental Health Treatment Limitation

(Rev. 1, 09-11-02)

Regardless of the actual expenses a beneficiary incurs for treatment of mental, psychoneurotic, and personality disorders while the beneficiary is not an inpatient of a hospital at the time such expenses are incurred, the amount of those expenses that may be recognized for Part B deductible and payment purposes is limited to 62.5 percent of the Medicare allowed amount for these services. The limitation is called the outpatient mental health treatment limitation. Since Part B deductible also applies the program pays for about half of the allowed amount recognized for mental health therapy services.

Expenses for diagnostic services (e.g., psychiatric testing and evaluation to diagnose the patient's illness) are not subject to this limitation. This limitation applies only to therapeutic services and to services performed to evaluate the progress of a course of treatment for a diagnosed condition.

30.1 - Application of Mental Health Limitation - Status of Patient

(Rev. 1, 09-11-02)

The limitation is applicable to expenses incurred in connection with the treatment of an individual who is not an inpatient of a hospital. Thus, the limitation applies to mental health services furnished to a person in a physician's office, in the patient's home, in a skilled nursing facility, as an outpatient, and so forth. The term "hospital" in this context means an institution which is primarily engaged in providing to inpatients, by or under the supervision of a physician(s):

- Diagnostic and therapeutic services for medical diagnosis, and treatment, and care of injured, disabled, or sick persons;
- Rehabilitation services for injured, disabled, or sick persons; or
- Psychiatric services for the diagnosis and treatment of mentally ill patients.

30.2 - Disorders Subject to Mental Health Limitation

(Rev. 1, 09-11-02)

The term "mental, psychoneurotic, and personality disorders" is defined as the specific psychiatric conditions described in the American Psychiatric Association's Diagnostic and Statistical Manual of Mental Disorders, Third Edition - Revised (DSM-III-R).

If the treatment services rendered are for both a psychiatric condition and one or more nonpsychiatric conditions, the charges are separated to apply the limitation only to the mental

health charge. Normally HCPCS code and diagnoses are used. Where HCPCS code is not available on the claim, revenue code is used.

If the service is primarily on the basis of a diagnosis of Alzheimer's Disease (coded 331.0 in the International Classification of Diseases, 9th Revision) or Alzheimer's or other disorders (coded 290.XX in DSM-III-R), treatment typically represents medical management of the patient's condition (rather than psychiatric treatment) and is not subject to the limitation.

30.3 - Diagnostic Services

(Rev. 1, 09-11-02)

The mental health limitation does not apply to tests and evaluations performed to establish or confirm the patient's diagnosis. Diagnostic services include psychiatric or psychological tests and interpretations, diagnostic consultations, and initial evaluations. However, testing services performed to evaluate a patient's progress during treatment are considered part of treatment and are subject to the limitation.

40 - Limitation on Services of Physical or Occupational Therapist in Independent Practice (Rev. 1, 09-11-02)

Coverage of outpatient physical or occupational therapy under Part B includes services of a qualified therapist in private practice when furnished in the therapist's office or the beneficiary's home. Payment for these services is based on the Medicare physician fee schedule less coinsurance and any deductible amounts due.

There was a limit for the amount of therapy expenses that would be recognized as payable for years before 2000. (For 1999 the limit is \$1500; for years 1994 through 1998, the limit is \$900; for years 1990 through 1993, the limit is \$750; and for years prior to 1990, the limit is \$500.)

Medicare General Information, Eligibility, and Entitlement

Chapter 4 - Physician Certification and Recertification of Services

Table of Contents

- 10 Certification and Recertification by Physicians for Hospital Services General
 - 10.1 Failure to Certify or Recertify for Hospital Services
 - 10.2 Who May Sign Certification or Recertification
 - 10.3 Certification for Hospital Admissions for Dental Services
 - 10.4 Inpatient Hospital Services Certification and Recertification
 - 10.5 Selection by Hospital of Format and Method for Obtaining Statement
 - 10.6 Criteria for Continued Inpatient Hospital Stay
 - 10.7 Utilization Review (UR) in Lieu of Separate Recertification Statement
 - 10.8 Timing of Certifications and Recertifications
 - 10.8.1 Admissions On or After January 1, 1970 for Non-PPS Hospitals
 - 10.8.2 Patients Discharged During Hospital Fiscal Years Beginning On or After October 1, 1983 Under PPS
 - 10.9 Inpatient Psychiatric Hospital Services Certification and Recertification
- 20 Certification for Hospital Services Covered by the Supplementary Medical Insurance Program
 - 20.1 Delayed Certifications and Recertifications
 - 20.2 Timing for Certification and Recertification for a Beneficiary Admitted Before Entitlement
- 30 Certification and Recertification by Physicians for Home Health Services
 - 30.1 Content of the Physician's Certification

- 30.2 Method and Disposition of Certifications for Home Health Services
- 30.3 Recertifications for Home Health Services
- 40 Certification and Recertification by Physicians for Extended Care Services
 - 40.1 Who May Sign the Certification or Recertification for Extended Care Services
 - 40.2 Certification for Extended Care Services
 - 40.3 Recertifications for Extended Care Services
 - 40.4 Timing of Recertifications for Extended Care Services
 - 40.5 Delayed Certifications and Recertifications for Extended Care Services
 - 40.6 Disposition of Certification and Recertifications for Extended Care Services
- 50 Physician's Certification and Recertification for Outpatient Physical Therapy
 - 50.1 Content of Physician's Certification
 - 50.2 Recertification
- 60 Certification and Recertification by Physicians for Hospice Care
- 70 DME Certification
- 80 Summary Table for Certifications/Recertifications

10 - Certification and Recertification by Physicians for Hospital Services - General (Rev. 1, 09-11-02)

Payments may be made for covered hospital services only if a physician certifies and recertifies to the medical necessity for the services at designated intervals of the hospital inpatient stay. Appropriate supporting material may be required. The physician certification or recertification statement must be based on a current evaluation of the patient's condition.

For patients admitted to a general hospital, regardless of whether the patients are under PPS, a physician certification is not required at the time of admission for patient services. For services continued over a period of time or for a day outlier case (i.e., an appropriately admitted case results in an extraordinarily long stay) or for a PPS cost outlier case (i.e., an appropriately admitted case results in the expenditure of extraordinary resources), a physician must certify or recertify the continued need for the services at specified intervals. (See §80 for timing of physician certification and recertification.) Psychiatric and tuberculosis hospitals (which are excluded from PPS) are required to obtain a physician certification on admission.

Hospitals do not transmit physician certification and recertification statements to the intermediary or to CMS. The hospital must itself certify on the appropriate billing form that the

required physician certification and recertification statements have been obtained and are on file. The physician certification and recertification statements are retained in the hospital's file where they are available for verification if needed.

A hospital must also have available in its files a written description of the procedure it adopts on the timing of certifications and recertifications, i.e., the intervals at which the necessary certification statements are required and whether review of long stay cases by the utilization review committee may serve as an alternative to recertification by a physician in the case of the second or subsequent recertification.

10.1 - Failure to Certify or Recertify for Hospital Services

(Rev. 1, 09-11-02)

If a hospital fails to obtain the required certification or recertification statements in an individual case, program payments may not be made in that case.

If the hospital's failure to obtain a certification or recertification is not due to a question as to the necessity for the services, but rather to the physician's refusal to certify based on other grounds (e.g., he/she objects in principle to the concept of certification and recertification), the hospital may not bill the program or the beneficiary for covered items or services. The provider agreement precludes the hospital from charging the patient for covered items and services.

10.2 - Who May Sign Certification or Recertification

(Rev. 1, 09-11-02)

A certification or recertification statement must be signed by the attending physician responsible for the case or by another physician who has knowledge of the case and is authorized to do so by the attending physician, or by a member of the hospital's medical staff with knowledge of the case.

Ordinarily for purposes of certification and recertification, a "physician" must meet the definition in chapter 5, §70 and §70.3.

10.3 - Certification for Hospital Admissions for Dental Services

(Rev. 1, 09-11-02)

The attending doctor of dental surgery or of dental medicine is authorized to certify that the patient's underlying medical condition and clinical status or the severity of the dental procedure requires the patient to be admitted to the hospital for the performance of the dental procedure; and to recertify the patient's continuing need for hospitalization when required. This applies even if the dental procedure is not covered.

10.4 - Inpatient Hospital Services Certification and Recertification

(Rev. 1, 09-11-02)

A certification or recertification statement must contain the following information:

- An adequate written record of the reason for either;
 - Continued hospitalization of the patient for medical treatment or for medically required inpatient diagnostic study, or
 - Special or unusual services for cost outlier cases for hospitals under the prospective payment system (PPS);
- The estimated period of time the patient will need to remain in the hospital and, for cost outlier cases, the period of time for which the special or unusual services will be required; and
- Any plans for posthospital care.

10.5 - Selection by Hospital of Format and Method for Obtaining Statement (Rev. 1, 09-11-02)

The individual hospital determines the method by which certifications and recertifications are to be obtained and the format of the statement. Thus, the medical and administrative staffs of each hospital may adopt the form and procedure they find most convenient and appropriate.

There is no requirement that the certification or recertification be entered on any specific form or handled in any specific way, as long as the approach adopted by the hospital permits the intermediary to determine that the certification and recertification requirements are, in fact, met. The certification or recertification could, therefore, be entered or preprinted on a form the physician already has to sign; or a separate form could be used. If all the required information is included in progress notes, the physician's statement could indicate that the individual's medical record contains the information required and that continued hospitalization is medically necessary.

10.6 - Criteria for Continued Inpatient Hospital Stay (Rev. 1, 09-11-02)

A physician who certifies or recertifies to the need for continued inpatient stay should use the same criteria that apply to the hospital's utilization review committee. These criteria include not only medical necessity, but also the availability of out-of-hospital facilities and services which will assume continuity of care. A physician should certify or recertify need for continued hospitalization if the physician finds that the patient could receive treatment in a SNF but no bed is available in the participating SNF. Where the basis for the certification or recertification is the need for continued inpatient care because of the lack of SNF accommodations, the certification or recertification should so state. The physician is expected to continue efforts to place the patient in a participating SNF as soon as the bed becomes available.

10.7 - Utilization Review (UR) in Lieu of Separate Recertification Statement (Rev. 1, 09-11-02)

For cases not subject to PPS and for PPS day outlier cases, a separate recertification statement is not necessary where the requirements for the second or subsequent recertification are satisfied by review of a stay of extended duration, pursuant to the hospital's UR plan. However, it is necessary to satisfy the certification and recertification content standards. It would be sufficient if records of the UR committee show that consideration was given to the three items required for certifications and recertifications: the reasons for continued hospitalization (e.g., consideration was given to the need for special or unusual care in cost outlier status under PPS), estimated time the patient will need to remain in the hospital (e.g., the time period during which such special or unusual care would be needed), and plans for posthospital care.

10.8 - Timing of Certifications and Recertifications

(Rev. 1, 09-11-02)

The timing of certifications and recertifications is described in the following subsections.

10.8.1 - Admissions On or After January 1, 1970 for Non-PPS Hospitals (Rev. 1, 09-11-02)

For services furnished to beneficiaries admitted on or after January 1, 1970, the initial certifications are required no later than as of the 12th day of hospitalization. A hospital may at its option, provide for the certification to be made earlier, or it may vary the timing of the certification within the 12-day period by diagnostic or clinical categories. The first recertification is required no later than as of the 18th day of hospitalization. Subsequent recertifications must be made at intervals established by the UR committee (on a case-by-case basis), but in no event may the interval between recertifications exceed 30 days.

The UR committee will be reviewing long-stay cases and may be in the best position to decide when subsequent recertifications are needed.

A hospital can, if it wishes, coordinate its physician recertifications with the process of review by the UR committee of longstay cases not subject to PPS, and for PPS day-outlier cases. At the option of the hospital, review of a stay of extended duration under the hospital's utilization review plan may take the place of the second and any subsequent physician recertifications. (Such review may be the initial review, or a second or subsequent review of an extended case by the UR committee.)

Where review of an extended stay case by the UR committee is deemed to take the place of a physician recertification, it would be possible for the recertification to be made later than the specified day, because the review of an extended duration case may be made at any time within the 7-day period following the last day of the period of extended duration defined in the utilization review plan. Such a recertification will be treated as a delayed recertification; however, no explanation for the normal delay is required.

10.8.2 - Patients Discharged During Hospital Fiscal Years Beginning On or After October 1, 1983 Under PPS

(Rev. 1, 09-11-02)

For cases subject to the prospective payment system (PPS), certification is not required at the time of admission for inpatient services. The admission is reviewed by a hospital review organization upon discharge of the patient. For outlier cases certification is required as follows:

- For day-outlier cases (now discontinued), certification was required no later than 1 day after the hospital reasonably assumes that the case meets the established outlier criteria, or no later than 20 days into the hospital stay, whichever is earlier. The first and subsequent recertifications are required at intervals established by the utilization review committee, on a case-by-case basis if it so chooses, but not less than every 30 days.
- For cost-outlier cases, if possible, certification must be made before the hospital incurs cost for which it will seek cost outlier payment. However, certification is required no later than the date on which the hospital requests cost outlier payment or 20 days into the hospital stay, whichever is earlier. For cost-outlier cases, the first and subsequent recertifications are required at intervals established by the UR committee, on a case-by-case basis if it chooses.

As previously stated the UR committee will be reviewing long-stay cases and may be in the best position to decide when subsequent recertifications are needed. Review by the UR committee used in place of recertification for PPS day outlier cases is considered timely if performed within 7 days after the physician recertification would have been required.

10.9 - Inpatient Psychiatric Hospital Services Certification and Recertification (Rev. 1, 09-11-02)

The requirements for physician certification and recertification for inpatient psychiatric hospital services are similar to the requirements for certification and recertification for inpatient hospital services. However, there is an additional certification requirement: At the time of admission or as soon thereafter as is reasonable and practicable, a physician (the admitting physician or a medical staff member with a knowledge of the case) must certify the medical necessity for inpatient psychiatric hospital services. As a result, the first and second recertification for psychiatric hospital services correspond to the initial certification and first recertification requirements for inpatient hospital services.

For services furnished to beneficiaries admitted on or after January 1, 1970, the first recertification is required no later than the 12th day of hospitalization. The second recertification is required no later than the 18th day of hospitalization. Subsequent recertifications must be made at intervals established by the utilization review committee (on a case-by-case basis) but in no event may the interval between recertifications exceed 30 days.

For services furnished to beneficiaries admitted before January 1, 1970, the first recertification was required no later than the 14th day of hospitalization. The second recertification was required no later than the 21st day of hospitalization and subsequent recertifications at intervals of no more than 30 days thereafter.

There is also a difference in the content of the certification and recertification statements. The required physician's statement should certify that the inpatient psychiatric hospital admission was medically necessary for either (1) treatment which could reasonably be expected to improve the patient's condition, or (2) diagnostic study.

The physician's recertification should state (1) that inpatient psychiatric hospital services furnished since the previous certification or recertification were, and continue to be, medically necessary for either (a) treatment which could reasonably be expected to improve the patient's condition , or (b) diagnostic study; and that the hospital records indicate that the services furnished were either intensive treatment services, admission and related services necessary for diagnostic study, or equivalent services .

For convenience, the period covered by the physician's certification and recertification is referred to a period during which the patient was receiving active treatment. If the patient remains in the hospital but the period of "active treatment" ends (e.g., because the treatment cannot reasonably be expected to improve the patient's condition, or because intensive treatment services are not being furnished), program payment can no longer be made even though the patient has not yet exhausted his/her benefits. Where the period of "active treatment" ends, the physician is to indicate the ending date in making his/her recertification. If "active treatment" thereafter resumes, the physician should indicate, in making his/her recertification, the date on which it resumed.

20 - Certification for Hospital Services Covered by the Supplementary Medical Insurance Program

(Rev. 1, 09-11-02)

A physician must certify that medical and other health services covered by medical insurance which were provided by (or under arrangement made by) the hospital were medically required.

Physician certification is not required for the following outpatient services furnished on or after January 3, 1968:

- Hospital services and supplies incident to physicians' services rendered to outpatients and
- Diagnostic services furnished by a hospital or which the hospital arranges to have furnished in other facilities operated by or under the supervision of the hospital or its medical staff.

Hospitals must obtain a physician's certification with respect to other services furnished to outpatients.

Primarily, this means that a certification statement is needed for diagnostic services furnished under arrangements by a facility that is not operated by or under the supervision of the hospital or its organized medical staff, e.g., services obtained from an independent laboratory.

This certification requires a brief description of the services and the signature of the physician. It needs to be made only once for a course of treatment. Where services are provided on a continuing basis, such as a course of radium treatments, the physician's certification may be

made at the beginning or end of the course of treatment, or at any other time during the period of treatment

There is no requirement that the certification be entered on any specific form or handled in any specific way, as long as the approach adopted by the hospital permits the intermediary to determine that the certification requirement is in fact met. Therefore, the certification could be entered or pre-printed on a form the physician already has to sign; or a separate certification form could be used.

Certification by a physician in connection with ambulance services furnished by a participating hospital is required. In cases in which the hospital provides ambulance service to transport the patient from the scene of an accident and no physician is involved until the patient reaches the hospital, any physician in the hospital who examines the patient or has knowledge of the case may certify as to the medical need for the ambulance service.

In addition, physician's certifications are required for the rental and purchase of durable medical equipment (see §70), outpatient physical therapy (see §50), and outpatient speech pathology services.

20.1 - Delayed Certifications and Recertifications

(Rev. 1, 09-11-02)

Hospitals are expected to obtain timely certification and recertification statements. However, delayed certifications and recertifications will be honored where, for example, there has been an oversight or lapse.

In addition to complying with the appropriate content requirements, delayed certifications and recertifications must include an explanation for the delay and any medical or other evidence which the hospital considers relevant for purposes of explaining the delay. The hospital will determine the format of delayed certification and recertification statements, and the method by which they are obtained. A delayed certification and recertification may appear in one statement; separate signed statements for each certification and recertification would not be required as they would if timely certification and recertification had been made.

20.2 - Timing for Certification and Recertification for a Beneficiary Admitted Before Entitlement

(Rev. 1, 09-11-02)

If an individual is admitted to a hospital (including a psychiatric hospital) before he/she is entitled to hospital insurance benefits (for example, before attainment of age 65), no certification is required as of the date of admission or entitlement. Certifications and recertifications are required as of the time they would be required if the patient had been admitted to the hospital on the day he/she became entitled. (The time limits for certification and recertification are computed from the date of entitlement instead of the date of admission.)

30 - Certification and Recertification by Physicians for Home Health Services

(Rev. 1, 09-11-02)

30.1 - Content of the Physician's Certification

(Rev. 1, 09-11-02)

Under both the hospital insurance and the supplementary medical insurance programs, no payment can be made for covered home health services that a home health agency provides unless a physician certifies that:

- The home health services are because the individual is confined to his/her home and needs intermittent skilled nursing care, physical therapy and/or speech therapy, or continues to need occupational therapy;
- A plan for furnishing such services to the individual has been established and is periodically reviewed by a physician; and
- The services are or were furnished while the individual was under the care of a physician.

Since the certification is closely associated with the plan of care (POC), the same physician who establishes the plan must also certify to the necessity for home health services. Certifications must be obtained at the time the plan of care is established or as soon thereafter as possible.

The attending physician signs and dates the POC/certification prior to the claim being submitted for payment; rubber signature stamps are not acceptable. The form may be signed by another physician who is authorized by the attending physician to care for his/her patients in his/her absence. While the regulations specify that documents must be signed, they do not prohibit the transmission of the POC or oral order via facsimile machine. The Home Health Agency (HHA) is not required to have the original signature on file. However, the HHA is responsible for obtaining original signatures if an issue surfaces that would require verification of an original signature.

HHAs which maintain patient records by computer rather than hard copy may use electronic signatures. However, all such entries must be appropriately authenticated and dated. Authentication must include signatures, written initials, or computer secure entry by a unique identifier of a primary author who has reviewed and approved the entry. The HHA must have safeguards to prevent unauthorized access to the records and a process for reconstruction of the records upon request from the intermediary, state surveyor, or other authorized personnel, in the event of a system breakdown.

(See §10.1 for the effects of failure to certify or recertify.)

30.2 - Method and Disposition of Certifications for Home Health Services

(Rev. 1, 09-11-02)

There is no requirement that the certification or recertification be entered on any specific form or handled in any specific way as long as the intermediary can determine, where necessary, that the certification and recertification requirements are met. The CMS Form CMS-485 (the Home Health Certification and Plan of Care) meets regulatory and national survey requirements for the physician's POC, certification and recertification. (See the Program Integrity Manual for Form CMS-485 and instructions for completion.) The certification by the physician must be retained by the home health agency.

The following instructions pertain to required documentation of the certification and recertification period both before and after the implementation of the home health prospective payment system.

For Dates of Service before the effective date of the Home Health Prospective Payment System (HH PPS) (October 1, 2000):

The HHA enters the month, day, year, e.g., MMDDYYYY that identifies the period covered by the physician's POC. The "From" date for the initial certification must match the SOC date. The "To" date can be up to, but never exceed 2 calendar months and, mathematically, never exceed 62 days. The "To" date is repeated on a subsequent re-certification as the next sequential "From" date. Services delivered on the "To" date are covered in the next certification period.

EXAMPLE: Initial certification "From" date 03012000; Initial certification "To" date 05012000; Re-certification "From" date 05012000; Re-certification "To" date 07012000.

For Dates of Service on or after the effective date of HH PPS (October 1, 2000):

The HHA enters the month, day, year, e.g., MMDDYYYY that identifies the period covered by the physician's POC. The "From" date for the initial certification must match the SOC date. The "To" date is up to and including the last day of the episode which is not the first day of the subsequent episode. The "To" date can be up to, but never exceed a total of 60 days that includes the SOC date plus 59 days.

EXAMPLE: Initial certification "From" date 10012000; Initial certification "To" date 11292000; Re-certification "From" date 11302000; Re-certification "To" date 01282001.

NOTE: Services delivered on 11292000 are covered in the initial certification episode.

30.3 - Recertifications for Home Health Services

(Rev. 1, 09-11-02)

Under both the hospital insurance and supplementary medical insurance programs, when services are continued for a period of time, the physician must recertify at intervals of at least once every 60 days that there is a continuing need for services and should estimate how long services will be needed. The recertification should be obtained at the time the plan of care is reviewed since the same interval (at least once every 60 days) is required for the review of the plan.

The physician must recertify that an individual needs or needed skilled nursing care on an intermittent basis or physical or speech therapy or, in the case of an individual who has been furnished home health services based on such a need and who no longer has such a need for such care or therapy, needs or continues to need occupational therapy. Recertifications must be signed by the physician who reviews the plan of treatment. The form of the recertification and the manner of obtaining timely recertifications are up to the individual agency.

40 - Certification and Recertification by Physicians for Extended Care Services (Rev. 1, 09-11-02)

Payment for covered posthospital extended care services may be made only if a physician makes the required certification and, where services are furnished over a period of time, the required recertification regarding the services furnished.

The skilled nursing facility is responsible for obtaining the required physician certification and recertification statements and for retaining them in file for verifications, if needed, by the intermediary. The skilled nursing facility determines the method by which the physician certification and recertification statements are to be obtained. There is no requirement that a specific procedure or specific forms be used, as long as the approach adopted by the facility permits a verification to be made that the certification and recertification requirements are in fact met. Certification and recertification statements may be entered on or included in forms, notes, or other records a physician normally signs in caring for a patient, or a separate form may be used. Except as otherwise specified, each certification and recertification statement is to be separately signed by a physician.

If the facility's failure to obtain a certification or recertification is not due to a question as to the necessity for the services, but rather to the physician's refusal to certify based on other grounds (e.g., he or she objects in principle to the concept of certification and recertification), the facility may not bill the program or the beneficiary for covered items or services. The provider agreement which the facility files with the Secretary precludes it from charging the patient for covered items and services.

If a physician refuses to certify because, in his/her opinion, the patient does not require skilled care on a continuing basis for a condition for which he/she was receiving inpatient hospital services, the services are not covered and the facility can bill the patient directly. The reason for the physician's refusal to make the certification must be documented in the facility records. For such documentation to be adequate, there must be some statement in the facility's records, signed by a physician or a responsible facility official, indicating that the patient's physician feels that the patient does not require skilled care on a continuing basis for any of the conditions for which he/she was hospitalized.

40.1 - Who May Sign the Certification or Recertification for Extended Care Services (Rev. 1, 09-11-02)

A certification or recertification statement must be signed by the attending physician or a physician on the staff of the skilled nursing facility who has knowledge of the case or by a nurse

practitioner or clinical nurse specialist who does not have a direct or indirect employment relationship with the facility, but who is working in collaboration with the physician.

Ordinarily, for purposes of certification and recertification, a "physician" must meet the definition contained in chapter 5, §70 of this manual.

40.2 - Certification for Extended Care Services

(Rev. 1, 09-11-02)

The certification must clearly indicate that posthospital extended care services were required to be given on an inpatient basis because of the individual's need for skilled care on a continuing basis for any of the conditions for which he/she was receiving inpatient hospital services, including services of an emergency hospital (see chapter 5, §20.2 of this manual) prior to transfer to the SNF. Certifications must be obtained at the time of admission, or as soon thereafter as is reasonable and practicable. The routine admission procedure followed by a physician would not be sufficient certification of the necessity for posthospital extended care services for purposes of the program.

If ambulance service is furnished by a skilled nursing facility, an additional certification is required. It may be furnished by any physician who has sufficient knowledge of the patient's case, including the physician who requested the ambulance or the physician who examined the patient upon his/her arrival at the facility. The physician must certify that the ambulance service was medically required.

40.3 - Recertifications for Extended Care Services

(Rev. 1, 09-11-02)

The recertification statement must contain an adequate written record of the reasons for the continued need for extended care services, the estimated period of time required for the patient to remain in the facility, and any plans, where appropriate, for home care. The recertification statement made by the physician does not have to include this entire statement if, for example, all of the required information is in fact included in progress notes. In such a case, the physician's statement could indicate that the individual's medical record contains the required information and that continued posthospital extended care services are medically necessary. A statement reciting only that continued extended care services are medically necessary is not, in and of itself, sufficient.

If the circumstances require it, the first recertification and any subsequent recertifications must state that the continued need for extended care services is for a condition requiring such services which arose after the transfer from the hospital and while the patient was still in the facility for treatment of the condition(s) for which he or she had received inpatient hospital services.

40.4 - Timing of Recertifications for Extended Care Services

(Rev. 1, 09-11-02)

The first recertification must be made no later than the l4th day of inpatient extended care services. A skilled nursing facility can, at its option, provide for the first recertification to be made earlier, or it can vary the timing of the first recertification within the l4-day period by diagnostic or clinical categories. Subsequent recertifications must be made at intervals not exceeding 30 days. Such recertifications may be made at shorter intervals as established by the utilization review committee and the skilled nursing facility.

At the option of the skilled nursing facility, review of a stay of extended duration, pursuant to the facility's utilization review plan (if a UR review plan is in place), may take the place of the second and any subsequent physician recertifications. The skilled nursing facility should have available in its files a written description of the procedure it adopts with respect to the timing of recertifications. The procedure should specify the intervals at which recertifications are required, and whether review of long-stay cases by the utilization review committee serves as an alternative to recertification by a physician in the case of the second or subsequent recertifications.

40.5 - Delayed Certifications and Recertifications for Extended Care Services (Rev. 1, 09-11-02)

Skilled nursing facilities are expected to obtain timely certification and recertification statements. However, delayed certifications and recertifications will be honored where, for example, there has been an isolated oversight or lapse.

In addition to complying with the content requirements, delayed certifications and recertifications must include an explanation for the delay and any medical or other evidence which the skilled nursing facility considers relevant for purposes of explaining the delay. The facility will determine the format of delayed certification and recertification statements, and the method by which they are obtained. A delayed certification and recertification may appear in one statement; separate signed statements for each certification and recertification would not be required as they would if timely certification and recertification had been made.

40.6 - Disposition of Certification and Recertifications for Extended Care Services (Rev. 1, 09-11-02)

Skilled nursing facilities do not have to transmit certification and recertification statements to the intermediary; instead, the facility must itself certify, in the admission and billing form, that the required physician certification and recertification statements have been obtained and are on file.

50 - Physician's Certification and Recertification for Outpatient Physical Therapy (Rev. 1, 09-11-02)

50.1 - Content of Physician's Certification (Rev. 1, 09-11-02)

No payment is made for outpatient physical therapy, occupational therapy, or speech pathology services unless a physician certifies that:

- The outpatient physical therapy, occupational therapy, or speech pathology services are or were furnished while the patient was under the care of a physician;
- A plan for furnishing such services is or was established by the physician, physical therapist, occupational therapist, or speech pathologist and periodically reviewed by the physician; and
- Services are or were required by the patient.

Since the certification is closely associated with the plan of treatment, the same physician who establishes or reviews the plan must certify to the necessity for the services. Providers obtain the certification at the time the plan of treatment is established or as soon thereafter as possible. Physician means a doctor of medicine, osteopathy (including an osteopathic practitioner), or podiatric medicine legally authorized to practice by the State in which he/she performs these services. In addition, physician certifications or recertifications by doctors of podiatric medicine must be consistent with the scope of the professional services provided by a doctor of podiatric medicine as authorized by applicable State law.

50.2 - Recertification

(Rev. 1, 09-11-02)

When outpatient physical therapy, occupational therapy, or speech pathology services are continued under the same plan of treatment, the physician must recertify at intervals of at least once every 30 days that there is a continuing need for such services and estimate how long services are needed. Providers obtain the recertification at the time the plan of treatment is reviewed since the same interval (at least once every 30 days) is required for the review of the plan. Recertifications are signed by the physician who reviews the plan of treatment. The form and manner of obtaining timely recertification is up to the provider.

60 - Certification and Recertification by Physicians for Hospice Care

(Rev. 1, 09-11-02)

The hospice must obtain written certification of terminal illness for each period of hospice care received by an individual. For the initial 90-day period, the hospice must obtain written certification statements from the medical director of the hospice or the physician member of the hospice interdisciplinary group, and the individual's attending physician (if the individual has one). The certification must specify that the individual's prognosis is for a life expectancy of 6 months or less if the terminal illness runs its normal course. Recertification for subsequent periods only requires the written certification by the hospice medical director or the physician member of the hospice interdisciplinary group. Certification statements must be dated and signed by the physician.

If written certification is not obtained within 2 calendar days of the initiation of hospice care, a verbal certification must be obtained within the 2 days. A written certification from the medical director of the hospice or the physician member of the interdisciplinary group must be on file in

the beneficiary's record prior to the submission of a claim to the intermediary. If these requirements are not met, no payment may be made for the days prior to certification. Instead payment will begin with the day certification is obtained, i.e., the date verbal certification is obtained.

70 - DME Certification

(Rev. 1, 09-11-02)

The DME supplier must retain a copy of the physician's order for DME in its files; and in some cases must furnish a Certificate of Medical Necessity to the Carrier.

80 - Summary Table for Certifications/Recertifications

(Rev. 1, 09-11-02)

The following is a table summarizing the certification/recertification signature requirements and timeframes for various provider types. Please review sections above for more detailed information on Certifications/Recertifications and their required content:

	Who Signs Certification	o Signs Certification Certification Timeframe	
Hospital Inpatient	Attending physician or by another physician with knowledge of the case with authorization from attending physician or by a member of hospital's medical staff with knowledge of the case.	No later than the 12 th day of hospitalization	Interval between recertifications not to exceed 30 days
SNF	Attending physician or physician on staff at SNF with knowledge of case	Obtain at time of admission or shortly thereafter	First recertification no later than the 14 th day of inpatient extended care services. Subsequent at intervals not exceeding 30 days.
ННА	Attending physician	Obtain at time POC is established or shortly thereafter	Physician must recertify at least once every 60 days
Hospice	For initial 90-day period, must obtain written certification statements from medical director of hospice or physician member of the	If written certification is not obtained within 2 calendar days of the initiation of hospice care, a verbal	Must be obtained for each period of hospice care; written certification by hospice medical director or physician member of

hospice interdisciplinary group certification must be interdisciplinary group. and the attending physician. obtained.

Medicare General Information, Eligibility, and Entitlement

Chapter 5 - Definitions

Table of Contents

10 - Part A Provider and Related Definition	10	- Part	Α	Provi	der	and	Rel	ated	De	fini	tio	ns
---	----	--------	---	-------	-----	-----	-----	------	----	------	-----	----

- 10.1 Provider Agreements
 - 10.1.1 Basic Commitment in Provider Agreement
 - 10.1.2 Part A Deductible and Coinsurance
 - 10.1.3 Part B Deductible and Coinsurance
 - 10.1.4 Blood
 - 10.1.5 Services Requested by Beneficiary
 - 10.1.6 Provider Charges to Beneficiary Where Provider Customarily Furnishes More Expensive Services Not Requested by Beneficiary
 - 10.1.7 Hospitals Participating in State Reimbursement Control Systems or Demonstration Projects
 - 10.1.8 Medicare Secondary Payer Involvement of Failure to File Proper Claims
 - 10.1.9 Advance Directive Requirements
 - 10.1.10 Posting of Signs in Hospital Emergency Departments
- 10.2 Admission of Medicare Patients for Care and Treatment
- 10.3 Under Arrangements
- 10.4 Term of Agreements
 - 10.4.1 Agreement with a SNF
 - 10.4.2 Agreement with Rural Health Clinic (RHC)/Federally Qualified Health Clinic (FQHC)
- 10.5 Responsibilities of Participating Provider

10.6 - Termination of Provider Participation

- 10.6.1 Voluntary Termination
- 10.6.2 Involuntary Termination, Including SNF Agreement Cancellations
- 10.6.3 Expiration and Renewal Nonrenewal of SNF Term Agreements
- 10.6.4 Determining Payment for Services Furnished After Termination, Expiration, or Cancellation
- 10.6.5 Change of Provider Ownership

20 - Hospital Defined

- 20.1 Definition of Emergency Inpatient and Outpatient Services
- 20.2 Definition of an Emergency Services Hospital
- 20.3 Psychiatric Hospital
- 20.4 Certification of Parts of Institutions as Hospital
- 20.5 Part of a Psychiatric Institutions as a Psychiatric Hospital
- 20.6 General Hospital Facility of Psychiatric Hospital
- 20.7 Part of a General Hospital as a Psychiatric Hospital

30 - Skilled Nursing Facility Defined

- 30.1 Distinct Part of an Institution as a SNF
- 30.2 Transfer Agreements
- 30.3 Hospital Providers of Extended Care Services

40 - Religious Non-medical Health Care Institutions

- 40.1 Beneficiary Eligibility
- 40.2 Election
 - 40.2.1 Revocation of Election
 - 40.2.2 Election after Revocation
- 40.3 Beneficiary Liability
- 40.4 Qualifying Provisions for a Provider

50 - Home Health Agency Defined

50.1 - Subdivisions of Agencies as Home Health Agencies

- 50.2 Arrangements by Home Health Agencies
- 50.3 Arrangements with Parent Agency and Other Entities
- 50.4 Notice of Noncoverage of Services
- 50 5 Rehabilitation Centers

60 - Hospice Defined

- 60.1 Subdivision of Organizations as Hospices
- 60.2 Arrangements by Hospices

70 - Physician Defined

- 70.1 Doctors of Medicine and Osteopathy
- 70.2 Dentists
- 70.3 Doctors of Podiatric Medicine
- 70.4 Physicians in Federal Hospitals
- 70.5 Optometrists
- 70.6 Chiropractors
- 70.7 Interns and Residents
- 80 Health Maintenance Organizations (HMOs) Defined
- 90 Other Definitions
 - 90.1 Supplier Defined
 - 90.2 Laboratory Defined
 - 90.3 Practitioners Defined
 - 90.4 Group Practice Defined

10 - Part A Provider and Related Definitions

(Rev. 1, 09-11-02)

Section 1866(e) of the Social Security Act defines the term "provider of services" (or provider) as:

(1) A clinic, rehabilitation agency, or public health agency if, in the case of a clinic or rehabilitation agency, such clinic or agency meets the requirements of section 1861(p)(4)(A) (or

meets the requirements of such section through the operation of section 1861(g)), or if, in the case of a public health agency, such agency meets the requirements of section 1861(p)(4)(B) (or meets the requirements of such section through the operation of section 1861(g)), but only with respect to the furnishing of outpatient physical therapy services (as therein defined) or (through the operation of section 1861(g)) with respect to the furnishing of outpatient occupational therapy services; and

(2) A community mental health center (as defined in section 1861(ff)(3)(B)), but only with respect to the furnishing of partial hospitalization services (as described in section 1861(ff)(1)).

Definitions of providers, physicians, practitioners, and suppliers, and a description of the requirements that each must meet in order for their services to be considered covered are described in the following sections.

10.1 - Provider Agreements

(Rev. 1, 09-11-02)

The following provider types must have provider agreements under Medicare:

- Hospitals,
- Skilled nursing facilities (SNFs),
- Home health agencies (HHAs),
- Clinics, rehabilitation agencies, and public health agencies,
- Comprehensive outpatient rehabilitation facilities (CORFs),
- Hospices,
- Critical access hospitals (CAHs), and
- Community mental health centers (CMHCs).

Clinics, rehabilitation agencies, and public health agencies may enter into provider agreements only for furnishing outpatient therapy services as identified in §10. CMHCs may enter into provider agreements only to furnish partial hospitalization services.

The term "provider agreement" is defined in 42 CFR 489.3 as an agreement between CMS and one of the providers specified in this section to provide services and to comply with the requirements of section 1866 of the Act.

A provider which has executed an agreement becomes qualified to participate after the agreement is accepted. When the agreement is made retroactive, the provider must comply with the terms of the agreement and the provisions of title XVIII and regulations issued thereunder as of the retroactive date. For payment to be made to the provider for covered items and services it furnishes on or after the effective date of the agreement, the provider must have a record keeping capability sufficient to determine the costs of services furnished to Medicare beneficiaries.

Provider agreements require the providers to comply with regulations. Therefore, new provider agreements are not made when regulations change.

Providers as defined in this section may also function as suppliers and bill the program for other services provided as suppliers if they meet the applicable requirements for supplying the specific service.

10.1.1 - Basic Commitment in Provider Agreement

(Rev. 1, 09-11-02)

Section 1866 of the Act and 42 CFR 489 require the provider to agree to the following:

- 1. To limit its charges to beneficiaries and to other individuals on their behalf to:
- The deductible and coinsurance amounts (see §10.1.2 for details),
- The blood deductible (see §10.1.4 for details), and
- Services requested by the beneficiary. (See §10.1.5)
- 2. To return any amounts incorrectly collected from a beneficiary or any other person on the beneficiary's behalf;
- 3. To notify the intermediary promptly if it hires an individual who at any time during the preceding year was employed in a managerial, accounting, auditing, or similar capacity by an intermediary or carrier;
- 4. In the case of a hospital or an critical access hospital (CAH), either to furnish directly or to make arrangements (as defined in §10.3 of this chapter) for all Medicare-covered services to inpatients of a hospital or CAH except the following:
- Physicians' services that meet the criteria of 42 CFR 405.550(b) for payment on a reasonable charge basis,
- Physician assistant services, as defined in section 1861(s)(2)(K)(I) of the Act, that are furnished after December 31, 1990,
- Certified nurse-midwife services, as defined in section 1861(ff) of the Act, that are furnished after December 31, 1990,
- Qualified psychologist services, as defined in section 1861(ii) of the Act, that are furnished after December 31, 1990, and
- Services of an anesthetist, as defined in 42 CFR 410.69.
- 5. In the case of a hospital or CAH that furnishes inpatient hospital services or inpatient CAH services for which payment may be made under Medicare, to maintain an agreement with a PRO for that organization to review the admissions, quality, appropriateness, and diagnostic information related to those inpatient services. The requirement of this paragraph applies only if,

for the area in which the hospital or CAH is located, there is a PRO that has a contract with CMS under Part B of title XI of the Act;

- 6. To maintain a system that, during the admission process, identifies any primary payers other than Medicare so that incorrect billing and Medicare overpayments can be prevented;
- 7. To bill other primary payers before billing;
- 8. If the provider receives payment for the same services from Medicare and another payer that is primary to Medicare, to reimburse Medicare any overpaid amount within 60 days;
- 9. If the provider receives, from a payer that is primary to Medicare, a payment that is reduced because the provider failed to file a proper claim; to reimburse Medicare any overpaid amount up to the amount that would have been paid had the provider filed a proper claim timely.
- 10. In the State of Oregon, because of a court decision, and in the absence of a reversal on appeal or a statutory clarification overturning the decision, hospitals may bill liability insurers first. However, if the liability insurer does not pay "promptly," the hospital must withdraw its claim or lien and bill Medicare for covered services;
- 11. In the case of home health agencies, to offer to furnish catheters, catheter supplies, ostomy bags, and supplies related to ostomy care to any individual who requires them as part of their furnishing of home health services;
- 12. In the case of hospital emergency department services that provide for medical screening to determine if an emergency medical condition exists, CMS guidelines provided in CFR 42 489.24(d) for transfer of patients to other facilities should be followed;
- 13. In the case of hospital emergency department services report to CMS or the State Survey Agency any time it has reason to believe it may have received an individual who has been transferred in an unstable emergency medical condition from another hospital in violation of the requirements of CFR 42 489.24(d):
- 14. In the case of inpatient hospital services for admissions on and after January 1, 1987, to participate in the Tricare program;
- 15. In the case of inpatient hospital services for admissions on and after July 1, 1987, to admit veterans whose admission is authorized by the VA and to meet related VA admission and payment requirements;
- 16. In the case of a hospital, to give each beneficiary a notice about his or her discharge rights at or about the time of the individual's admission;
- 17. In the case of a hospital with an emergency department:
- To post conspicuously in any emergency department or in a place or places likely to be noticed by all individuals entering the emergency department, as well as those individuals waiting for examination and treatment in areas other than traditional emergency departments (that is, entrance, admitting area, waiting room, treatment area), a sign (in a form specified by

- the Secretary) specifying rights of individuals under Section 1867 of the Act with respect to examination and treatment for emergency medical conditions and women in labor; and
- To post conspicuously information indicating whether or not the hospital or critical access hospital participates in the Medicaid program under a State plan approved under title XIX;
- 18. In the case of a hospital with an emergency department (including both the transferring and receiving hospitals), to maintain:
- Medical and other records related to individuals transferred to or from the hospital for a period of 5 years from the date of the transfer;
- A list of physicians who are on call for duty after the initial examination to provide treatment necessary to stabilize an individual with an emergency medical condition; and
- A central log on each individual who comes to the emergency department seeking assistance and whether he or she refused treatment, was refused treatment, or whether he or she was transferred, admitted and treated, stabilized and transferred, or discharged.
- 19. Effective December 1, 1991, in the case of a hospital to comply with the advance directive provisions of 4206 of OBRA 1990. Hospitals must, in accordance with written policies and procedures, for all adult individuals:
- Inform them, in writing, of State laws regarding advance directives;
- Inform them, in writing, of its policies regarding the implementation of advance directives (including a clear and concise explanation of a conscientious objection, to the extent that State law permits for a hospital or any agent of a hospital that, as a matter of conscience, cannot implement an advance directive);
- Document in the individual's medical record whether the individual has executed an advance directive;
- Not condition the provision of care or otherwise discriminate against an individual based on whether that individual has executed an advance directive (since the law does not require the individual to do so); and
- Educate staff and the community on issues concerning advance directives.

10.1.2 - Part A Deductible and Coinsurance

(Rev. 1, 09-11-02)

This section is a further explanation of §10.1.1 above.

The provider may charge the beneficiary or his or her representative:

• The amount of the inpatient hospital deductible or, if less, the actual charges for the services;

- The amount of inpatient hospital coinsurance applicable for each day the individual is furnished inpatient hospital services after the 60th day, during a benefit period;
- The posthospital extended care coinsurance amount; and
- In the case of durable medical equipment (DME) furnished as a home health service, 20 percent of the fee schedule amount.

10.1.3 - Part B Deductible and Coinsurance

(Rev. 1, 09-11-02)

This section is a further explanation of §10.1.1above.

The provider may charge the beneficiary or other person on his or her behalf: the \$100 deductible and 20 percent of the customary (insofar as reasonable) charges in excess of that deductible.

For hospital outpatient services, the allowable deductible charges depend on whether the hospital can determine the beneficiary's deductible status. If the hospital is unable to determine the deductible status, it may charge the beneficiary its full customary charges up to \$100. If the beneficiary provides official information as to deductible status, the hospital may charge only the unmet portion of the deductible.

The hospital is required to indicate on the UB-92 the amounts collected.

In the case of DME furnished as a home health service under Medicare Part B, the coinsurance is 20 percent of the fee schedule amount for the services, with the following exception: If the DME is used, purchased by, or on behalf of the beneficiary at a price at least 25 percent less than the reasonable charge for comparable new equipment, no coinsurance is required.

10.1.4 - Blood

(Rev. 1, 09-11-02)

This section is a further explanation of §10.1.1.

A provider may charge the beneficiary (or other person on his or her behalf) only for the first three pints of blood or units of packed red cells furnished during the calendar year. The charges may not exceed the provider's customary charges.

The provider may not charge for any whole blood or packed red cells in any of the following circumstances:

- The provider obtained the blood or red cells at no charge other than a processing or service charge;
- The blood or packed red cells have been replaced; or

• The provider (or its blood supplier) receives from an individual, or a blood bank, a replacement offer. This offer is applicable if the replacement blood would not endanger the health of a recipient and if the prospective donor's health would not be endangered by making a blood donation. In this case the provider is precluded from charging even if it or its blood supplier rejects the replacement offer.

10.1.5 - Services Requested by Beneficiary

(Rev. 1, 09-11-02)

This section is a further explanation of §10.1.1.

If services furnished at the request of a beneficiary (or his or her representative) are more expensive than, or in excess of, services covered under Medicare, a provider may charge the beneficiary an amount that does not exceed the difference between (1) the provider's customary charges for the services furnished; and (2) the provider's customary charges for the kinds and amounts of services that are covered under Medicare.

A provider may not charge for these services unless they have been requested by the beneficiary (or his or her representative) and may not require a beneficiary to request services as a condition of admission.

To avoid misunderstanding and disputes, a provider must inform any beneficiary who requests a service for which a charge will be made that there will be a specified charge for that service.

10.1.6 - Provider Charges to Beneficiary Where Provider Customarily Furnishes More Expensive Services Not Requested by Beneficiary

(Rev. 1, 09-11-02)

A provider that customarily furnishes an individual items or services that are more expensive than the items or services determined to be necessary in the efficient delivery of needed health services, may charge an individual entitled to benefits under Medicare for such more expensive items or services even though not requested by the individual. The charge, however, may not exceed the amount by which the cost of (or, if less, the customary charges for) more expensive items or services furnished by such provider in the second cost reporting period immediately preceding the cost reporting period in which such charges are imposed, exceeds the applicable limit imposed under the provisions of 42 CFR 413.30.

This charge may be made only if:

- The intermediary determines that the charges have been calculated properly in accordance with the provisions of this regulation section;
- The services are not emergency services as defined in §20.1 or §20.2 of this chapter;
- The admitting physician has no direct or indirect financial interest in such provider;

- CMS has provided notice to the public through notice in a newspaper of general circulation servicing the provider's locality and such other notice as the Secretary may require, of any charges the provider is authorized to impose on individuals entitled to benefits under Medicare on account of costs in excess of the costs determined to be necessary in the efficient delivery of needed health services under Medicare; and
- The provider has identified such charges to such individual or person acting on his/her behalf
 as charges to meet the costs in excess of the costs determined to be necessary in the efficient
 delivery of needed health services under Medicare.

10.1.7 - Hospitals Participating in State Reimbursement Control Systems or Demonstration Projects

(Rev. 1, 09-11-02)

This section is a further explanation of §10.1.1.

A hospital receiving payment for a covered hospital stay under:

- Either a State reimbursement control system approved under 1886 (c) of the Act; or
- A demonstration project authorized under section 402(a) of Pub. L. 90 248 (42 U.S.C. 1395b 1) or section 222(a) of Pub. L. 92 603 (42 U.S.C.1395b 1 (note)); and
- Would otherwise be subject to the prospective payment system set forth in part 412 of this chapter, may also charge a beneficiary for custodial care and medically unnecessary services described in CFR CUR 412.42(c), after the conditions of 412.42(c)(1) through(c)(4) are met.

10.1.8 - Medicare Secondary Payer Involvement of Failure to File Proper Claims (Rev. 1, 09-11-02)

This section is a further explanation of §10.1.1.

A provider that has not filed a proper claim and has received a reduced payment because of this from a payer primary to Medicare agrees:

- To bill Medicare for an amount no greater than would have been payable as secondary payment if the primary insurer's payment had been based on a proper claim; and
- To charge the beneficiary only (a) the amount it would have been entitled to charge if it had filed a proper claim and received payment based on such a claim; and (b) an amount equal to any third-party payment reduction attributable to failure to file a proper claim, but only if the provider can show that:
 - It failed to file a proper claim solely because the beneficiary, for any reason other than mental or physical incapacity, failed to give the provider the necessary information; or
 - The beneficiary, who was responsible for filing a proper claim, failed to do so for any reason other than mental or physical incapacity.

10.1.9 - Advance Directive Requirements

(Rev. 1, 09-11-02)

Effective December 1, 1991, participating hospitals must comply with the advance directive provisions of §4206 of OBRA 1990. Therefore, an agreement per §1866 of the Act with a hospital includes that the hospital must, in accordance with written policies and procedures, for all adult individuals: inform them, in writing, of State laws regarding advance directives; inform them, in writing, of its policies regarding the implementation of advance directives (including a clear and concise explanation of a conscientious objection, to the extent that State law permits for a hospital or any agent of a hospital that, as a matter of conscience, cannot implement an advance directive); document in the individual's medical record whether the individual has executed an advance directive; not condition the provision of care or otherwise discriminate against an individual based on whether that individual has executed an advance directive (since the law does not require the individual to do so); and educate staff and the community on issues concerning advance directives

10.1.10 - Posting of Signs in Hospital Emergency Departments

(Rev. 1, 09-11-02)

Section 6018(a)(2) of the Omnibus Budget Reconciliation Act of 1989 (OBRA §89), effective July 1, 1990, requires hospitals with emergency departments to post signs which specify the rights (under section 1867 of the Social Security Act) of women in labor and individuals with emergency medical conditions to examination and treatment.

To comply with these requirements, hospitals must post signs that meet the following criteria:

- At a minimum, the signs must specify the rights of unstable individuals with emergency conditions and women in labor who come to the emergency department for health care services:
- It must indicate whether the facility participates in the Medicaid program;
- The wording of the sign must be clear and in simple terms understandable by the population serviced;
- Print the signs in English and other major languages that are common to the population of the area serviced;
- The letters within the signs must be clearly readable at a distance of at least 20 feet or the expected vantage point of the emergency department patrons; and
- Post signs in a place or places likely to be noticed by all individuals entering the emergency department, as well as those individuals waiting for examination and treatment (e.g., entrance, admitting area, waiting room, treatment area).

The sample below may be adapted, contains sufficient information to satisfy these requirements, and may be adapted to satisfy the visibility requirement.

IT'S THE LAW!

IF YOU HAVE A MEDICAL EMERGENCY OR ARE IN LABOR

YOU HAVE THE RIGHT TO RECEIVE, WITHIN THE CAPABILITIES OF THIS HOSPITAL'S STAFF AND FACILITIES:

An appropriate medical SCREENING EXAMINATION

Necessary STABILIZING TREATMENT (including treatment for an unborn child)

And if necessary

An appropriate TRANSFER to another facility

Even if

YOU CANNOT PAY OR DO NOT HAVE MEDICAL INSURANCE

OR

YOU ARE NOT ENTITLED TO MEDICARE OR MEDICAID

This hospital (does/does not) participate in the Medicaid program

10.2 - Admission of Medicare Patients for Care and Treatment

(Rev. 1, 09-11-02)

The participation of a provider of services which voluntarily files an agreement to participate in the health insurance program contemplates that such provider will admit Medicare beneficiaries for care and treatment, and upon admission, will provide them with such services as are ordinarily furnished by the provider to its patients generally.

A provider may have restrictions on the types of services it makes available and/or the types of health conditions it accepts, or may establish other criteria relating to the admission of persons for care and treatment. However, the law does not contemplate that such restrictions or criteria will apply only to Medicare beneficiaries as a class. It does contemplate, however, that if such restrictions or criteria apply to Medicare beneficiaries, they will be applied in the same manner in which they are applied to all other persons seeking care and treatment by the provider. Thus, a provider admission or patient policy or practice which is not consistent with the objective contemplated in the law may be used by CMS as a basis for termination of the agreement for cause.

10.3 - Under Arrangements

(Rev. 1, 09-11-02)

A provider may have others furnish certain covered items and services to their patients through arrangements under which receipt of payment by the provider for the services discharges the liability of the beneficiary or any other person to pay for the service. In permitting providers to furnish services under arrangements, it was not intended that the provider merely serve as a billing mechanism for the other party. Accordingly, for services provided under arrangements to be covered, the provider must exercise professional responsibility over the arranged-for services.

The provider's professional supervision over arranged-for services requires application of many of the same quality controls as are applied to services furnished by salaried employees. The provider must accept the patient for treatment in accordance with its admission policies, and maintain a complete and timely clinical record on the patient, which includes diagnoses, medical history, physician's orders, and progress notes relating to all services received, and must maintain liaison with the attending physician regarding the progress of the patient and the need for revised orders. In the case of home health services and outpatient physical therapy or speech pathology services, the provider must ensure that the required plan of treatment is periodically reviewed by the physician and secure from the physician the required certifications and recertifications. Additionally, the provider (other than a SNF) must ensure that the medical necessity of such services is reviewed on a sample basis by the utilization review (UR) committee if one is in place, the facility's health professional staff, or an outside UR group. (Effective October 1, 1990, a SNF is no longer required to have a plan for UR.) The provider, including a SNF that conducts optional UR services, is responsible for medical necessity decisions made under arrangement by an outside group.

10.4 - Term of Agreements

(Rev. 1, 09-11-02)

An agreement with a hospital, HHA, hospice, and (for the purposes of furnishing outpatient physical therapy, occupational therapy, or speech pathology services) a clinic, a rehabilitation agency, or public health agency is not time limited and has no fixed expiration date. The agreement remains in effect until such time as there is a voluntary termination, or involuntary termination, or a change of ownership.

10.4.1 - Agreement with a SNF

(Rev. 1, 09-11-02)

All agreements with SNFs are required to be for a specified term of up to 12 full calendar months with fixed expiration dates. In appropriate situations, the agreement may also contain a cancellation clause. Therefore, when it is determined that a SNF is eligible and that its agreement for participation in the Medicare program will be accepted for filing, the term of the agreement will be determined in the following manner:

- A. No Deficiencies If a SNF is certified as being in full compliance with all standards contained in the requirements for participation for SNFs, the term of the agreement is, as appropriate to the period of certification for it has been approved, for a term of up to 12 full calendar months.
- B. Deficiencies If a SNF is certified as not being in full compliance, the term of the agreement is, as appropriate to the period of certification or conditional period of certification for which it has been approved:
 - 1 For a term which expires no later than the close of the 60th day following the last day of the time period specified in the written plan providing for the correction of deficiencies in meeting the requirements for participation, provided that such term does not exceed 12 full calendar months or
 - 2 Provide a conditional term of up to 12 full calendar months, subject to a cancellation clause that the agreement may be canceled on a predetermined date (no later than the close of the 60th day following the last day of the time period specified in the written plan providing for the correction of deficiencies in meeting the requirements for participation), provided that date will occur within the 12-month term.

If the health and safety of Medicare patients are not jeopardized, the term of an agreement may be extended for 2 full calendar months when necessary to prevent irreparable harm to the facility, to prevent hardship to Medicare beneficiaries furnished care, or if it is impracticable within the term of the agreement to determine whether the SNF is complying with the provisions of the Act and regulations issued thereunder.

10.4.2 - Agreement with Rural Health Clinic (RHC)/Federally Qualified Health Clinic (FQHC)

(Rev. 1, 09-11-02)

Agreements between RHCs/FQHCs and CMS are generally for a term of 1 year. They may be annually renewed by mutual agreement of the RHC/FQHC and CMS, i.e., a new agreement need not be signed each year. Special circumstances may result in a term of less than 1 year for an initial agreement, e.g., a clinic or center may wish the agreement year to run concurrently with the RHC/FQHC's fiscal year or have some other technical considerations.

10.5 - Responsibilities of Participating Provider

(Rev. 1, 09-11-02)

Upon acceptance for participation, a provider becomes responsible for remaining in compliance with the terms of the agreement and the provisions of title XVIII and regulations issued thereunder. Where the agreement is made retroactive, the provider must comply with its responsibilities as of the retroactive date. Payment to the provider for covered items and services it furnishes on or after the effective date of the agreement will require that the provider have a recordkeeping capability sufficient for determining the cost of services furnished Medicare beneficiaries.

The termination of participation (see §10.6 of this chapter) does not immediately abrogate all of the provider's responsibilities and in specific matters a responsibility may extend beyond the effective date of termination. For example, the provider continues to be responsible (as applicable) for those provisions of the law and regulations which provide for program coverage to remain in effect for specified periods of time beyond the effective termination date, for those beneficiaries who were accepted for care and treatment by the provider before such date. The provider also continues to be responsible for filing a final cost report and/or the repayment of any overpayment, as these actions relate to final program cost settlement after termination.

10.6 - Termination of Provider Participation

(Rev. 1, 09-11-02)

A provider may voluntarily terminate its participation in the program or have it terminated by the Secretary for cause.

10.6.1 - Voluntary Termination

(Rev. 1, 09-11-02)

A provider may terminate its agreement (and in the case of a SNF, it may terminate its agreement prior to the close of the specified term of its agreement) by filing with the Secretary a written notice of its intention to terminate the agreement. The Secretary may accept the termination date stated in the notice (the date must be the first day of a month and, in the case of a SNF, must occur within the specified term of such facility's agreement) or set a different date. However, the termination date set by the Secretary may not be more than 6 months from the date the provider's notice is filed.

10.6.2 - Involuntary Termination, Including SNF Agreement Cancellations (Rev. 1, 09-11-02)

The Secretary may terminate an agreement (and in the case of a SNF, he/she may terminate its agreement prior to the close of the specified term of the agreement) with a provider if it is determined that the provider:

- Is not complying substantially with the provisions of the agreement or with the applicable provisions of title XVIII of the Act and regulations;
- No longer meets the appropriate requirements for participation;
- Has failed to supply information which is necessary to determine whether payments are due or were due and the amounts of such payments; or
- Refuses to permit examinations of fiscal and other records, including medical records.

The cancellation of a SNF agreement at the close of the predetermined date stated in the cancellation clause contained in such agreement (see §10.6.2 of this chapter) is viewed as an involuntary termination of the agreement by the Secretary for cause. Such actions involve a finding that the SNF has not satisfactorily completed its written plan providing for the correction of deficiencies with respect to one or more of the standards in the applicable requirements for participation, or that the facility has not made substantial effort and progress in correcting such deficiencies.

A provider which is dissatisfied with the Secretary's determination terminating its agreement is entitled to request a hearing thereon in accordance with the appeals procedures contained in 42 CFR Part 498. There is no reconsideration step before the opportunity for a hearing.

NOTE: The involuntary termination of a hospital's approval authorizing it to provide extended care services, i.e., to be a swing bed facility, does not automatically result in the involuntary termination of the hospital's agreement relating to the provision of hospital services.

10.6.3 - Expiration and Renewal - Nonrenewal of SNF Term Agreements (Rev. 1, 09-11-02)

All agreements with skilled nursing facilities are required to be for a specified term of up to 12 full calendar months with fixed expiration dates. The agreement expires at the close of the last day of its specified term and is not automatically renewable from term to term. When the term of an agreement is extended (see §10.6.3 of this chapter), the close of the last day of its specified term is the close of the day of the extension of the agreement. Thus, when the term of an agreement is extended, the provider's participation in the program continues, and the agreement does not expire until the close of the last day to which it has been extended.

Since an agreement with a SNF is not automatically renewable from term to term, each term agreement with a SNF requires that the SNF qualify for participation and that its agreement be accepted for filing. A participating SNF may, however, continue its participation under the agreement form previously accepted for filing, provided the SNF continues to qualify for

participation and the agreement form is again accepted for filing and renewed for a term which begins on the date immediately following the close of the last day of the prior term of the agreement. When the requirements for participation continue to be met, there is no limit to the number of times that the SNF's agreement form may again be accepted and renewed for a specified term.

When the time-limited agreement (including an agreement which has had its term extended) is renewed on the day immediately following the close of the last day of its term, the expiration of the agreement is not considered a termination of participation in the program.

However, once an agreement with a SNF is (1) not renewed, or (2) voluntarily terminated by the SNF, or (3) involuntarily terminated (including cancellations) by the Secretary, the previously accepted agreement cannot again be accepted and renewed. In such cases, the SNF is required to execute and file a new agreement if it is again found eligible to participate in the Medicare program. The effective date of the new agreement must be determined in accordance with regulatory provisions (42 CFR 489.13).

The Secretary's determination not to accept and renew a SNF agreement is a determination relating to the qualifications of the SNF in the period immediately following the close of the SNF's existing agreement and the SNF is entitled to request a reconsideration of the determination in accordance with the appeals procedure contained in 42 CFR part 405, subpart 0. Such determinations involve a finding that:

- Based on a State agency resurvey and recertification, the SNF will not be approved for a
 period of certification because it is out of compliance with one or more requirements for
 participation;
- Based on a State agency resurvey and recertification, the SNF continues to be out of compliance with the same standard(s) in the requirements for participation as were found out of compliance during the term of the agreement and the facility will not be approved for a new period of certification; or
- The SNF has violated the terms of its agreement or the provisions of title XVIII or regulations promulgated thereunder.

In cases of nonrenewal by the Secretary, the intermediary's role is the same as for involuntary terminations.

10.6.4 - Determining Payment for Services Furnished After Termination, Expiration, or Cancellation

(Rev. 1, 09-11-02)

Effective with the date a provider agreement (or swing bed approval) terminates, expires, or is cancelled, no payment is made to the provider under such agreement for the following:

A. Hospital

- 1. Termination-Hospital Agreement Inpatient hospital services (including inpatient psychiatric hospital services) and swing bed extended care services furnished on or after the effective date of the hospital's termination, except that payment can continue to be made for up to 30 days of inpatient hospital services and/or swing bed extended care services (total of no more than 30 days) furnished on or after the termination date to beneficiaries who were admitted (at either the acute or extended care level) prior to the termination date.
- 2. Termination-Swing Bed Approval Swing bed extended care services furnished on or after the effective date of the termination of the hospital's swing bed approval, except that payment can continue for up to 30 days of extended care services furnished on or after the termination date to beneficiaries who were admitted (at either the acute or extended care level) prior to the termination date

B. Skilled Nursing Facility

- l. Termination-SNF Posthospital extended care services furnished on or after the effective date of termination of the agreement, where such agreement has been voluntarily terminated by the provider (see §10.6.1 of this chapter) or involuntarily terminated (see 3 below for cancellations) by the Secretary for cause (see §10.6.2 of this chapter), except that payment can continue to be made for up to 30 days of posthospital extended care services furnished on and after the termination date to beneficiaries who were admitted prior to the termination date.
- 2. Expiration-SNF Posthospital extended care services furnished on or after the date which follows the last day of the specified term of the agreement, where such agreement has expired at the close of the last day of its specified term (see §10.6.3 of this chapter), except that where the agreement has not been renewed, payment can be made for up to 30 days of posthospital extended care services furnished on and after the date which follows the last day of the specified term of such agreement to beneficiaries who were admitted on or before such last day.
- 3. Cancellation-SNF Posthospital extended care services furnished on or after the date which follows the predetermined date specified in a cancellation clause, where such agreement is cancelled by the Secretary pursuant to such cancellation clause (see §10.6.2 of this chapter), except that payment can be made for up to 30 days of posthospital extended care services furnished on and after the date which follows the predetermined date specified in such clause, to beneficiaries who were admitted on or before such predetermined date.

C. HHA and Hospice

Payment may be made for services under a plan of treatment for up to 30 days following the effective termination date of a home health agency or hospice if the plan was established before the termination date.

D. Providers - Termination, Expiration, and Cancellation

Other items and services, including outpatient physical therapy or speech pathology and diagnostic services, furnished on or after the effective date of termination or, in the case of an expiration or cancellation of a SNF agreement, on or after the day following the close of such agreement.

(See Medicare Claims Processing Manual Chapter 1 for billing instructions concerning providers who are no longer participating.)

10.6.5 - Change of Provider Ownership

(Rev. 1, 09-11-02)

When an organization having a provider agreement undergoes a change of ownership, the agreement is automatically assigned to the new owner. A participating provider which plans to change ownership should give advance notice of its intention so that necessary action can be taken in the event the newly-owned institution does not wish to participate in the Medicare program.

A participating provider which plans to enter into a lease arrangement (in whole, or in part) should also give advance notice of its intention. A change of ownership would occur for example:

- Where a sole proprietor transfers title to an enterprise to another party;
- Where, in the case of a partnership, the addition, removal, or substitution of a partner effects a termination of such partnership and creates a successor partnership or other entity;
- Where an incorporated provider merges with an incorporated institution which is not
 participating in the program and the nonparticipating institution is the surviving corporation
 or where two or more corporate providers consolidate and such consolidation results in the
 creation of a new corporate entity; or
- Where an unincorporated provider (a sole proprietorship or partnership) becomes incorporated.

Whenever intermediaries learn of an impending change of ownership, or possible change of ownership, they inform the regional office immediately. They refer any questions they may have concerning change of ownership situations to the regional office.

20 - Hospital Defined

(Rev. 1, 09-11-02)

A hospital (other than psychiatric) means an institution which is primarily engaged in providing, by or under the supervision of physicians, to inpatients, diagnostic and therapeutic services for medical diagnosis, treatment, and care of injured, disabled, or sick persons; or rehabilitation services for the rehabilitation of injured, disabled, or sick persons.

To be eligible to participate in Medicare, a hospital must also be an institution which:

- Maintains clinical records on all patients;
- Has bylaws in effect with respect to its staff of physicians;
- Has a requirement that every patient must be under the care of a physician;

- Provides 24-hour nursing service rendered or supervised by a registered professional nurse, and has a licensed practical nurse or registered professional nurse on duty at all times;
- Has in effect a hospital utilization review plan;
- Is licensed or is approved by the State or local licensing agency as meeting the standards established for such licensing;
- Meets other health and safety requirements found necessary by the Secretary of Health, and Human Services. (These additional requirements may not be higher than comparable ones prescribed for accreditation by the Joint Commission on Accreditation of Hospitals with exceptions specified in the law); and
- Is not primarily for the care and treatment of mental diseases.

Such an institution, if approved to participate as a hospital, may also be approved as a swing bed facility pursuant to demonstration authority or if the hospital is a rural hospital with less than 50 beds. (See §30.3 below.)

20.1 - Definition of Emergency Inpatient and Outpatient Services

(Rev. 1, 09-11-02)

Payment may be made for certain Part A inpatient hospital services and Part B outpatient hospital services provided in a nonparticipating U.S. hospital where they are necessary to prevent the death or serious impairment of the health of the individual. Because of the threat to the life or health of the individual, the use of the most accessible hospital equipped to furnish such services is necessary. The determination of emergency services depends upon three separate findings:

- The hospital meets the definition of an emergency hospital (see §20.2 of this chapter);
- The services meet the definition of emergency services (see Claims Processing Manual, Chapter 3, §110), and
- The hospital is substantially more accessible from the site of the emergency than is the nearest participating hospital.

20.2 - Definition of an Emergency Services Hospital

(Rev. 1, 09-11-02)

An emergency services hospital is a nonparticipating hospital which meets the requirements of the law's definition of a "hospital" relating to full-time nursing services and licensure under State or applicable local law. (See §20.2 of this chapter.) (A Federal hospital need not be licensed under State or local licensing laws to meet the definition of emergency hospital.) In addition, the hospital must be primarily engaged in providing, under the supervision of doctors of medicine or osteopathy, services of the type that §20.1 describes in defining the term hospital, and must not be primarily engaged in providing skilled nursing care and related services for patients who

require medical or nursing care. (See the definition of a SNF in §30 of this chapter.) Psychiatric hospitals that meet these requirements can qualify as emergency hospitals.

Inpatient hospital services and related physician and ambulance services furnished outside the U.S. are covered under the limited conditions in Medicare Claims Processing Manual Chapter 1.

20.3 - Psychiatric Hospital

(Rev. 1, 09-11-02)

A psychiatric hospital is an institution which is primarily engaged in providing by or under the supervision of a physician, psychiatric services for the diagnosis and treatment of mentally ill persons. To be eligible for participation in the program as a psychiatric hospital, it must meet the Medicare conditions of participation for hospitals or be deemed to meet those conditions based on accreditation by the Joint Commission on Accreditation of Hospitals (JCAH), have a utilization review plan, and comply with additional staffing and medical record requirements necessary to carry out an active program of treatment and intensive care.

20.4 - Certification of Parts of Institutions as Hospital

(Rev. 1, 09-11-02)

Under certain conditions a distinct part of a psychiatric institution may be certified as a psychiatric or general hospital.

20.5 - Part of a Psychiatric Institutions as a Psychiatric Hospital

(Rev. 1, 09-11-02)

A distinct part of a psychiatric institution can be certified as a psychiatric hospital if it meets the conditions of participation even though the institution of which it is a part does not. If the distinct part meets requirements equivalent to the accreditation requirements of the JCAH, it can qualify under the program even though the institution itself is not accredited.

20.6 - General Hospital Facility of Psychiatric Hospital

(Rev. 1, 09-11-02)

A general hospital facility within a psychiatric hospital may be certified as a general hospital independent of the institution as a whole provided the general facility is a self-contained operational entity distinct from the rest of the institution. The general hospital facility would be regarded as a separate institution for this purpose since the law does not provide for certifying a "distinct part' of an institution as a general hospital. Services furnished in a separately certified general hospital facility are not subject to any of the benefit limitations applicable to the other parts of the institution, i.e., the reduction in benefit days in the first spell of illness and the 190-day lifetime maximum on inpatient services in psychiatric hospitals.

20.7 - Part of a General Hospital as a Psychiatric Hospital

(Rev. 1, 09-11-02)

There is no provision for a psychiatric wing of a general hospital to be certified as a psychiatric hospital. The distinct part provisions apply only to psychiatric institutions and not to general hospitals. However, this does not prevent the certification of a psychiatric hospital which is a part of a medical center or other large complex, provided the hospital operates as a separate functioning entity, i.e., it is located in a separate building, wing, or part of a building, has its own administration and maintains separate fiscal records.

30 - Skilled Nursing Facility Defined

(Rev. 1, 09-11-02)

A SNF is an institution or a distinct part of an institution (see §30.1 of this chapter), such as a skilled nursing home or rehabilitation center, which has a transfer agreement in effect with one or more participating hospitals (see §30.2 of this chapter for transfer agreements and §10.1 of this chapter, for definition of a participating hospital) and which:

- Is primarily engaged in providing skilled nursing care and related services for residents who
 require medical or nursing care; or rehabilitation services for the rehabilitation of injured,
 disabled, or sick persons, and
- Meets the requirements for participation in sections 1819(a) through 1819 (d) as amended by 4201 of OBRA 1987 of the Social Security Act and in regulations at 42 CFR 483, B.

A qualified SNF is one that meets all the requirements in the preceding definition.

For Medicare purposes, the term SNF does not include any institution which is primarily for the care and treatment of mental diseases. (This restriction does not apply to title XIX (Medicaid)). Also, the term "skilled nursing facility" does not include swing bed hospitals authorized to provide and be paid for extended care services. Swing bed hospitals must meet many of the same requirements that apply to SNFs (for more details regarding swing bed hospitals, see §30.3).

30.1 - Distinct Part of an Institution as a SNF

(Rev. 1, 09-11-02)

The term "distinct part" refers to a portion of an institution or institutional complex (e.g., a nursing home or a hospital) that is certified to provide SNF and/or nursing facility (NF) services. To qualify for participation in the program as a distinct part SNF of an institution, it must be physically separate from the rest of the institution, i.e., it must represent an entire physically identifiable unit consisting of all the beds within that unit, such as a separate building, floor, wing, or ward. A distinct part must be fiscally separate for cost reporting purposes. Although it is required that the distinct part be identifiable as a separate unit within the institution, it need not necessarily be confined to a single location within the institution's physical plant. The distinct part may, for example, consist of several floors or wards which are scattered throughout several

different buildings within the institutional complex. In each case, however, the patients of the distinct part must be located in units which are physically separate from those units housing all other patients of the institution. Various beds scattered throughout the institution do not comprise a distinct part for purposes of being approved by Medicare as a SNF.

An institution or institutional complex can only be certified with one distinct part SNF and/or one distinct part NF. A hospital-based SNF is by definition a distinct part. Multiple certifications within the same institution or institutional complex are strictly prohibited. The distinct part must consist of all beds within the designated area. Where an institution or institutional complex owns and operates a SNF and/or a NF distinct part, that SNF and/or NF distinct part is a single distinct part even if it is operated at various locations throughout the institution or institutional complex. The aggregate of the SNF and/or NF locations represents a single distinct part subprovider, not multiple subproviders, and must be assigned a single provider number.

30.2 - Transfer Agreements

(Rev. 1, 09-11-02)

To participate in the program, a SNF must have a written transfer agreement with one or more participating hospitals (see §10.1 of this chapter) providing for the transfer of patients between the hospital and the SNF, and for the interchange of medical and other information. If an otherwise qualified SNF has attempted in good faith, but without success, to enter into a transfer agreement, this requirement may be waived by the State agency. (See 42 CFR 483.75 (n) for the detailed requirements for transfer agreements.)

30.3 - Hospital Providers of Extended Care Services

(Rev. 1, 09-11-02)

In order to address the shortage of rural SNF beds for Medicare patients, effective July 20, 1982, rural hospitals with fewer than 50 beds could be reimbursed under Medicare for furnishing post hospital extended care services to Medicare beneficiaries. Such a hospital, known as a swing bed hospital, can "swing" its beds between hospital and SNF levels of care, on an as needed basis, if it has obtained a swing bed approval from the Department of Health and Human Services. Under 4005(b)(2) of the Omnibus Budget Reconciliation Act of 1987, effective for agreements entered into after March 31, 1988, rural hospitals with fewer than 100 beds must make application and request approval to be a swing bed hospital from the Regional Office. In order to obtain a swing bed approval, the hospital must:

- As noted above, be located in a rural area (i.e., located outside of an "urbanized area," as defined by the Census Bureau and based on the most recent census) and have fewer than 100 beds (excluding beds for newborns and intensive care-type units);
- Have a Medicare provider agreement, as a hospital;

- Be granted any necessary certificate of need for the provision of extended care services, as required by the State (this requirement was eliminated by §408 of the Balanced Budget Refinement Act of 1999 [BBRA, P.L. 106-113, Appendix F]);
- Be substantially in compliance with the SNF participation requirements identified in 42 CFR 482.66; (most other SNF participation requirements would be largely met by virtue of the facility's compliance with comparable hospital conditions);
- Not have in effect a 24-hour nursing waiver granted under 42 CFR 488.54(c); and
- Not have had a swing bed approval terminated within the 2 years previous to application for swing bed participation.

However, the Department may grant a swing bed approval, on a demonstration basis, with hospitals meeting all of the statutory requirements except bed size and geographic location.

When a hospital has a swing bed approval from the Department, it may provide and be reimbursed by Medicare Part A for providing extended care or SNF-type services. When a swing bed hospital provides extended care services, Medicare reimbursement for those services will be based on the average State Medicaid rate paid for SNF services in the prior calendar year. This rate is set under explicit statutory conditions and is described at 42 CFR 413.114.

When a hospital is providing extended care services, it will be treated as a SNF for purposes of applying coverage rules. This means that those services are subject to the same Part A coverage, physician certification/recertification, deductible and coinsurance provisions that are applicable to SNF extended care services.

Under §4005(b)(2) of the Omnibus Budget Reconciliation Act of 1987 (P.L. 100-203), effective for swing bed agreements entered into after March 31, 1988, rural hospitals with more than 49 beds (but less than 100 beds) are subject to the following conditions. However, these conditions were eliminated by §408 of the BBRA, effective with the facility's fourth cost reporting period that begins on or after July 1, 1998. (For those facilities that received no Medicare payment prior to October 1, 1995, this change is effective as of the date of the BBRA's enactment, November 29,1999.)

- If there is an available SNF bed in the geographic region, the extended care patient must be transferred within 5 days of the availability date (excluding weekends and holidays) unless the patient's physician certifies, within that 5-day period, that transfer of that patient to that facility is not medically appropriate on the availability date. In order to do this, hospitals need to identify all SNFs in their geographic region and enter into agreements with them for the transfer of extended care patients under which SNFs are to notify the hospitals of the availability of beds and the dates these beds will be available for extended care patients;
- For each cost reporting period, payment may not be made for patient days of extended care services that exceed 15 percent of the total number of available patient days (except that such payment shall continue to be made for those patients who are receiving extended care services at the time the hospital reaches the 15 percent limit). The limit is calculated by multiplying the average number of licensed beds by the total number of days in the cost reporting period (See §2230.10 of the Provider Reimbursement Manual, Part 1); and

• The 5-day transfer requirement and the 15 percent payment limitation do not apply for Medicaid reimbursement purposes.

Hospitals which have fewer than 50 beds and rural hospitals which entered into agreements before March 31, 1988 (i.e., those which were licensed for more than 49 beds but who were operating as a 50 or less bed facility) are not subject to the 5-day transfer requirement or the payment limitation for extended care days.

"Geographic region" is an area which includes the SNFs with which a hospital has traditionally arranged transfers and all other SNFs within the same proximity to the hospital. In the case of a hospital without existing transfer practices upon which to base a determination, the geographic region is an area which includes all the SNFs within 50 miles of the hospital unless the hospital can demonstrate that the SNFs are inaccessible to its patients. In the event of a dispute as to whether a SNF is within this region or the SNF is inaccessible to hospital patients, the CMS regional office shall make a determination.

40 - Religious Non-medical Health Care Institutions

(Rev. 1, 09-11-02)

Section 4454 provides for coverage of services furnished in a Medicare qualified religious nonmedical health care institution (RNHCI), when the beneficiary meets specific coverage conditions set forth in Section 1821(a) and (b) of the Act. The beneficiary must have a valid election for RNHCI services and would otherwise qualify for care in a conventional hospital or post hospital extended care facility, that was not a religious nonmedical health care institution. The programs will only pay for nonmedical health care services furnished in RNHCIs, as defined in Section 1861(ss)(1) of the Act and 42 CFR 403 Subpart G. The new provisions do not provide for supporting religious services or payment for the religious practitioner. The cost of using a religious practitioner is a personal financial responsibility and not covered by Medicare.

Pertinent Definitions:

Excepted medical care means medical care that is received involuntarily or required under Federal, State, or local laws. It is a new term specific to the provisions implementing section 4454 of BBA '97 and is intended to identify the kinds of medical services that can be provided to a beneficiary with an election for RNHCI services without revoking the election.

Medical care or treatment means health care furnished by or under the direction of a licensed physician that can involve diagnosing, treating, or preventing disease and other damage to the mind and body. It may involve the use of pharmaceuticals, diet, exercise, surgical intervention, and technical procedures.

Nonexcepted medical care means medical care, other than excepted medical care, that is sought by or for a beneficiary who has elected religious nonmedical health care institution services. It is a new term specific to the provisions implementing section 4454 of BBA '97 and is intended to define the kinds of medical services that, if received by a beneficiary who has previously elected RNHCI services, would revoke the individual's election of services. Under section 1821(b)(3) of the Act, a beneficiary may revoke an election in writing or by receiving nonexcepted medical care.

Religious non-medical care or religious method of healing means health care furnished under established religious tenets that prohibit conventional or unconventional medical care for the treatment of a beneficiary, and the sole reliance on these religious tenets to fulfill a beneficiary's total health care needs.

40.1 - Beneficiary Eligibility

(Rev. 1, 09-11-02)

This rule permits a Medicare beneficiary to elect to receive care in an RNHCI based on his or her own religious convictions or to revoke that election at any time if for any reason he or she decides to pursue medical care. Section 1821(a) of the Act requires that as a condition for Part A Medicare coverage, the beneficiary must have a condition that would qualify under Medicare Part A for inpatient hospital services or extended care services furnished in a hospital or skilled nursing facility that is not an RNHCI.

When a Medicare beneficiary has an effective election on file with CMS but does not have a condition that would qualify for Medicare Part A inpatient hospital or posthospital extended care services if the beneficiary were an inpatient of a hospital or a resident of a SNF that is not an RNHCI, then services furnished in an RNHCI are not covered by Medicare. A Medicare claim for services that were furnished to that beneficiary would be treated as a claim for noncovered services. If the beneficiary only needs assistance with activities of daily living, then the beneficiary's condition could not be considered as meeting the Medicare Part A requirements.

The beneficiary must have a valid election in effect to receive RNHCI services. A beneficiary who meets all other applicable requirements and who has in effect a valid election to receive services in an RNHCI is eligible for coverage of those services in an RNHCI.

If no valid election is filed or the election has been revoked and no new election is in effect, the beneficiary does not have Medicare coverage for services furnished in an RNHCI. Consequently, a Medicare claim for services furnished to such a beneficiary would also be treated as a claim for noncovered services.

40.2 - Election

(Rev. 1, 09-11-02)

The term election is a written statement signed by a beneficiary or the beneficiary's legal representative indicating the beneficiary's choice to receive nonmedical care or treatment for religious reasons. The "election" term is specific to the section 4454 provisions of BBA '97, and it is the mechanism by which a beneficiary chooses RNHCI services rather than other Medicare covered medical services.

40.2.1 - Revocation of Election

(Rev. 1, 09-11-02)

Revocation is the cancellation of the RNHCI election and can be achieved in two ways: either by submitting a written statement to the fiscal intermediary indicating the desire to cancel the election or by seeking nonexcepted medical care for which Medicare payment is sought.

40.2.2 - Election after Revocation

(Rev. 1, 09-11-02)

After an initial revocation, the individual may again file a written election to receive the religious nonmedical health care benefit. This second election takes effect immediately upon its execution. If an individual revokes a second election, the next (third) election cannot become effective until 1 year after the date of the most recent revocation. Subsequent elections are not effective until 5 years after the most recent revocation. Once an election is revoked, Medicare payment cannot be made to an RNHCI unless a valid election is filed. It should be noted that a revocation does not in any way limit a beneficiary's ability to access covered original Medicare benefits.

40.3 - Beneficiary Liability

(Rev. 1, 09-11-02)

RNHCIs are subject to Medicare rules for deductibles and coinsurance. Under normal Medicare rules, a provider of services may only bill a beneficiary deductible and coinsurance amounts. However, RNHCIs may also bill individuals an amount equal to any reduction in payments applied under sections 1821(c)(2) (A) or (B) of the Act.

40.4 - Qualifying Provisions for a Provider

(Rev. 1, 09-11-02)

In order for a Medicare or Medicaid provider to meet the definition of an RNHCI, it must satisfy the ten qualifying provisions as contained in new section 1861(ss)(1) of the Act. Section 1861(ss)(1) of the Act states that an RNHCI means an institution that:

- 1. Is described in subsection (c)(3) of section 501 of the Internal Revenue Code of 1986 and is exempt from taxes under subsection (a) of that section. The inability to either gain or retain this status will disqualify an institution from participation as an RNHCI.
- 2. Is lawfully operated under all applicable Federal, State, and local laws and regulations. Federal law supersedes State and local laws unless the State and local requirements are more stringent than the Federal requirements.
- 3. Furnishes only nonmedical nursing items and services to patients who choose to rely solely upon a religious method of healing and for whom the acceptance of medical services would be inconsistent with their religious beliefs. Medicare does not cover the religious component of the healing.
- 4. Furnishes nonmedical items and services exclusively through nonmedical nursing personnel who are experienced in caring for the physical needs of these patients. This care frequently

involves: assistance in moving, turning, positioning, and ambulation; meeting nutritional needs; and comfort and support measures.

- 5. Furnishes nonmedical items and services to inpatients on a 24-hour basis.
- 6. Does not furnish, on the basis of its religious beliefs, through its personnel or otherwise, medical items and services (including any medical screening, examination, diagnosis, prognosis, treatment, or the administration of drugs) for its patients.
- 7. Is not owned by, under common ownership with, or has an ownership interest of 5 percent or more in, a provider of medical treatment or services, and is not affiliated with a provider of medical treatment or services, or with an individual who has an ownership interest of 5 percent or more in, a provider of medical treatment or services. For purposes of this requirement, an affiliation does not exist in the circumstances described in section 1861(ss)(4) of the Act or 42 CFR 403.738(c).
- 8. Has in effect a utilization review plan that:
- Provides for review of admissions to the institution, of the duration of stays, of cases of continuous extended duration, and of the items and services furnished by the institution;
- Requires that the reviews be made by an appropriate committee of the institution that includes the individuals responsible for overall administration and for supervision of nursing personnel at the institution;
- Provides that records be maintained of the meetings, decisions, and actions of the committee;
 and
- Meets other requirements as the Secretary finds necessary to establish an effective utilization review plan.
- 9. Provides information the Secretary may require to implement section 1821 of the Act, including information relating to quality of care and coverage determinations.
- 10. Meets other requirements the Secretary finds necessary in the interest of the health and safety of individuals who are furnished services in the institution. These requirements include the conditions of participation in 42 CFR 403, Subpart G. An RNHCI must meet or exceed the conditions of participation in order to qualify as a Medicare provider. The RNHCI must also have a valid provider agreement with CMS.

50 - Home Health Agency Defined

(Rev. 1, 09-11-02)

A home health agency is a public agency or private organization, or a subdivision of such an agency or organization, which meets the following requirements:

• It is primarily engaged in providing skilled nursing services and other therapeutic services, such as physical, speech, or occupational therapy, medical social services, and home health aide services. A public or voluntary nonprofit health agency may qualify by:

- Furnishing both skilled nursing and at least one other therapeutic service directly to patients, or
- Furnishing directly either skilled nursing services or at least one other therapeutic service and having arrangements with another public or voluntary nonprofit agency to furnish the services which it does not provide directly.

NOTE: A proprietary agency can qualify only by providing directly both skilled nursing services and at least one other therapeutic service

- It has policies established by a professional group associated with the agency or organization (including at least one physician and at least one registered professional nurse) to govern the services, and provides for supervision of such services by a physician or a registered professional nurse;
- It maintains clinical records on all patients;
- It is licensed in accordance with State or local law or is approved by the State or local licensing agency as meeting the licensing standards (where State or local law provides for the licensing of such agencies or organizations); and
- It meets other conditions found by the Secretary of the Department of Health and Human Services to be necessary for health and safety.

A private organization which is not exempt from Federal income taxation under section 501 of the Internal Revenue Code of 1954 (sometimes referred to as a "proprietary" organization) must be licensed pursuant to State law. If the State has no licensing law for such organizations, a proprietary agency cannot participate in the health insurance program.

For services under hospital insurance, the term "home health agency" does not include any agency or organization which is primarily for the care and treatment of mental disease. There is no such restriction under supplementary medical insurance.

50.1 - Subdivisions of Agencies as Home Health Agencies

(Rev. 1, 09-11-02)

When the subdivision of an agency, such as the home care department of a hospital or the nursing division of a health department, wishes to participate as a home health agency, the subdivision must meet the conditions of participation and must maintain records in such a way that subdivision activities and expenditures attributable to services provided under the health insurance program are identifiable.

50.2 - Arrangements by Home Health Agencies

(Rev. 1, 09-11-02)

A home health agency (HHA) may have others furnish covered items or services through arrangements under which receipt of payment by the HHA for the services discharges the

liability of the patient or any other person to pay for the services. Whether the items and services are provided by the HHA itself or by another agency under arrangements, both must agree not to charge the patient for covered items and services and must also agree to return money incorrectly collected.

In permitting HHAs to furnish services under arrangements, it was not intended that the agency merely serve as a billing mechanism for the other party. Accordingly, for services provided under arrangements to be covered, the agency must exercise professional responsibility over the arranged-for services.

The agency's professional supervision over arranged-for services requires application of many of the same quality controls as are applied to services furnished by salaried employees. The agency must accept the patient for treatment in accordance with its admission policies, maintain a complete and timely clinical record of the patient that includes diagnosis, medical history, physician's orders, and progress notes relating to all services received; maintain liaison with the attending physician with regard to the progress of the patient and to assure that the required plan of treatment is periodically reviewed by the physician; secure from the physician the required certifications and recertifications; and ensure that the medical necessity of such services is reviewed on a sample basis by the agency's staff or an outside review group.

There are three situations in which an HHA may have arrangements with another health organization or person to provide home health services to patients:

- Where an agency or organization, in order to be approved to participate in the program, makes arrangements with another organization or individual to provide the nursing or other therapeutic services that it cannot provide directly;
- Where an agency that is already approved for participation, makes arrangements with others to provide services or items it does not provide directly; and
- Where an agency that is already approved for participation makes arrangements with a hospital, skilled nursing facility, or rehabilitation center for services on an outpatient basis because the services involve the use of equipment that cannot be made available to the patient in his/her place of residence.

50.3 - Arrangements with Parent Agency and Other Entities

(Rev. 1, 09-11-02)

If an agency's subdivision (acting in its capacity as an HHA) makes an arrangement with its parent agency for the provision of certain items or services, there need not be a contract or formal agreement. If, however, the arrangement is made between the HHA and another provider participating in the health insurance program (hospital, skilled nursing facility, or HHA, and, in the case of physical therapy, occupational therapy, or speech-language pathology services, clinics, rehabilitation agencies, and public health agencies), there must be a written statement regarding the services to be provided and the financial arrangements.

If the arrangements are with an agency or organization that is not a qualified provider of services, there must be a written contract that includes all of the following:

- A description of the services to be provided;
- The duration of the agreement and how frequently it is to be reviewed;
- A description of how personnel will be supervised;
- A statement that the contracting organization will provide services in accordance with the plan of care established by the patient's physician in conjunction with the HHA's staff;
- A description of the contracting organization's standards for personnel, including qualifications, functions, supervision, and inservice training;
- A description of the method of determining reasonable costs and reimbursement by the HHA for the specific services to be provided by the contracting organization; and
- An assurance that the contracting organization will comply with title VI of the Civil Rights Act.

50.4 - Notice of Noncoverage of Services

(Rev. 1, 09-11-02)

If an HHA notifies a beneficiary of noncoverage of services that another party has been furnishing under arrangements entered into by the agency, the initial notice, in and of itself, does not negate the contract between the agency and the other party. Unless the evidence shows that the contract has been formally terminated, the beneficiary is still considered to be the agency's patient and the other party to be the representative of the agency. Consequently, if upon initial notice that a service is no longer covered the other party continues to provide services to the patient, the other party is considered to be furnishing the services under arrangement.

50.5 - Rehabilitation Centers

(Rev. 1, 09-11-02)

When the services are of such a nature that they cannot be administered at the patient's residence and are administered at a rehabilitation center which is not participating in the program as a hospital, skilled nursing facility, or home health agency, the rehabilitation center must meet certain standards. The physical plant and equipment of such a rehabilitation center must meet all applicable State and local legal requirements for construction, safety, health, and design, including safety, sanitation and fire regulations, building codes, and ordinances.

60 - Hospice Defined

(Rev. 1, 09-11-02)

A hospice is a public agency or private organization or a subdivision of either that is primarily engaged in providing care to terminally ill individuals and meets the conditions of participation for hospices, and has a valid provider agreement.

60.1 - Subdivision of Organizations as Hospices

(Rev. 1, 09-11-02)

When a subdivision of an organization, such as the home care department of a hospital, wishes to participate as a hospice, the subdivision must meet the hospice conditions of participation and must maintain records in such a way that activities and expenditures attributable to services provided under the hospice program are identifiable.

60.2 - Arrangements by Hospices

(Rev. 1, 09-11-02)

Hospices are required to provide core services directly, that is, nursing services, medical social services, and counseling. Other covered services may be provided under arrangement.

70 - Physician Defined

(Rev. 1, 09-11-02)

Physician means doctor of medicine, doctor of osteopathy (including osteopathic practitioner), doctor of dental surgery or dental medicine (within the limitations in subsection §70.2), doctor of podiatric medicine (within the limitations in subsection §70.3), or doctor of optometry (within the limitations of subsection §70.5), and, with respect to certain specified treatment, a doctor of chiropractic legally authorized to practice by a State in which he/she performs this function. The services performed by a physician within these definitions are subject to any limitations imposed by the State on the scope of practice.

The issuance by a State of a license to practice medicine constitutes legal authorization. Temporary State licenses also constitute legal authorization to practice medicine. If State law authorizes local political subdivisions to establish higher standards for medical practitioners than those set by the State licensing board, the local standards determine whether a particular physician has legal authorization. If State licensing law limits the scope of practice of a particular type of medical practitioner, only the services within the limitations are covered.

The issuance by a State of a license to practice medicine constitutes legal authorization. Temporary State licenses also constitute legal authorization to practice medicine. If State law authorizes local political subdivisions to establish higher standards for medical practitioners than those set by the State licensing board, the local standards determine whether a particular physician has legal authorization. If State licensing law limits the scope of practice of a particular type of medical practitioner, only the services within the limitations are covered.

NOTE: The term physician does not include such practitioners as a Christian Science practitioner or naturopath.

70.1 - Doctors of Medicine and Osteopathy

(Rev. 1, 09-11-02)

The requirement that a doctor of medicine be legally authorized to practice medicine and surgery by the State in which he/she performs his/her services means a physician is licensed to practice medicine and surgery.

A doctor of osteopathy who is legally authorized to practice medicine and surgery by the State in which he/she performs his/her services qualifies as a physician. In addition, a licensed osteopath or osteopathic practitioner qualifies as a physician to the extent that he/she performs services within the scope of his/her practice as defined by State law.

70.2 - Dentists

(Rev. 1, 09-11-02)

A dentist qualifies as a physician if he/she is a doctor of dental surgery or of dental medicine who is legally authorized to practice dentistry by the State in which he/she performs such function and who is acting within the scope of his/her license when he/she performs such functions. Such services include any otherwise covered service that may legally and alternatively be performed by doctors of medicine, osteopathy and dentistry; e.g., dental examinations to detect infections prior to certain surgical procedures, treatment of oral infections and interpretations of diagnostic X-ray examinations in connection with covered services. Because the general exclusion of payment for dental services has not been withdrawn, payment for the services of dentists is also limited to those procedures which are not primarily provided for the care, treatment, removal, or replacement of teeth or structures directly supporting the teeth. The coverage of any given dental service is not affected by the professional designation of the physician rendering the service; i.e., an excluded dental service remains excluded and a covered dental service is still covered whether furnished by a dentist or a doctor of medicine or osteopathy.

70.3 - Doctors of Podiatric Medicine

(Rev. 1, 09-11-02)

A doctor of podiatric medicine is a physician, but only with respect to those functions which he/she is legally authorized to perform in the State in which he/she performs them. The professional services furnished by a doctor of podiatric medicine within the scope of his/her applicable State license (except services which are specifically excluded) are physician's services payable on a reasonable charge basis under Part B. Where permissible by State law, these services include ordering laboratory tests that are reasonably related to the legal scope of podiatric practice, that are reasonable and necessary for the diagnosis or treatment of a patient's condition and are not in connection with excluded services, such as treatment of flat foot and routine foot care.

A doctor of podiatric medicine may hold any of the following professional degrees: Pod. D. or D. P. (Doctor of Podiatry), D.S.C. (Doctor of Surgical Chiropody), D.P.M. (Doctor of Podiatric Medicine), D.S.P. (Doctor of Surgical Podiatry), Graduate in Podiatry, Master Chiropodist,

Graduate Chiropodist, or in some instances another podiatry degree. Within a particular State, all individuals holding any of these degrees are licensed to perform the same functions; however, there are variations from State to State as to the authorized scope of podiatric practice.

For purposes of the Medicare program, a doctor of podiatric medicine is considered a physician for any of the following purposes:

- Making the required physician certification and recertification of the medical necessity for services;
- Having a patient in a home health agency under his/her care, and establishing and periodically reviewing a home health plan of treatment; or
- Serving as a member of a Utilization Review (UR) committee, but only if at least two of the physicians on the UR committee are doctors of medicine or osteopathy. The performance of these functions must be consistent with the scope of the professional services provided by a doctor of podiatric medicine as authorized by applicable State law.

A doctor of podiatric medicine is not a physician for the purpose of performing any of the physician activities required to qualify an institution or organization as a SNF.

70.4 - Physicians in Federal Hospitals

(Rev. 1, 09-11-02)

There are many physicians performing services in hospitals operated by the Federal Government, e.g., military, Veterans Administration, and Public Health Service hospitals. Normally, the services provided by a physician in a Federal hospital are not payable except when the hospital provides services to the public as a community institution. A physician working in the scope of his/her Federal employment is considered a physician even though he/she may not have a license to practice in the State in which he/she is employed.

70.5 - Optometrists

(Rev. 1, 09-11-02)

A. Services Furnished Through March 31, 1987

Prior to April 1, 1987, a doctor of optometry who was legally authorized to practice optometry by the State in which he or she performed such a function was considered a physician under Medicare, but only for the purpose of services related to the condition of aphakia. Aphakia is defined as the absence of the natural crystalline lens of the eye, whether or not an intraocular lens has been implanted. The services performed by optometrists within this definition were subject to limitations set by the State relating to the scope of practice of optometry.

The following are examples of examination services which were covered when furnished by optometrists if related to the condition of aphakia: case history, external examination, ophthalmoscopy, biomicroscopy, tonometry, visual fields, ocular motility, binocular function,

and evaluation for contact lenses, if the optometrist furnishing these services is legally authorized to perform them.

B. Services Furnished After March 31, 1987

Effective April 1, 1987, a doctor of optometry is considered a physician with respect to all services the optometrist is authorized to perform under State law or regulation. To be covered under Medicare, the services must be medically reasonable and necessary for the diagnosis or treatment of illness or injury, and must meet all applicable coverage requirements. (See Benefit Policy Manual for information concerning exclusions from coverage that apply to vision care services.)

70.6 - Chiropractors

(Rev. 1, 09-11-02)

A. General

A licensed chiropractor who meets uniform minimum standards (see subsection C) is a physician for specified services. Coverage extends only to treatment by means of manual manipulation of the spine to correct a subluxation demonstrated by X-ray, provided such treatment is legal in the State where performed. All other services furnished or ordered by chiropractors are not covered. An X-ray obtained by a chiropractor for his or her own diagnostic purposes before commencing treatment may suffice for claims documentation purposes. This means that if a chiropractor orders, takes, or interprets an X-ray to demonstrate a subluxation of the spine, the X-ray can be used for claims processing purposes. However, there is no coverage or payment for these services or for any other diagnostic or therapeutic service ordered or furnished by the chiropractor.

In addition, in performing manual manipulation of the spine, some chiropractors use manual devices that are hand-held with the thrust of the force of the device being controlled manually. While such manual manipulation may be covered, there is no separate payment permitted for use of this device

B. Licensure and Authorization to Practice

A chiropractor must be licensed or legally authorized to furnish chiropractic services by the State or jurisdiction in which the services are furnished.

C. Uniform Minimum Standards

- l. Prior to July 1, 1974, Chiropractors licensed or authorized to practice prior to July 1, 1974, and those individuals who commenced their studies in a chiropractic college before that date must meet all of the following minimum standards to render payable services under the program:
- a. Preliminary education equal to the requirements for graduation from an accredited high school or other secondary school;
- b. Graduation from a college of chiropractic approved by the State's chiropractic examiners that included the completion of a course of study covering a period of not less than 3 school years of

6 months each year in actual continuous attendance covering adequate course of study in the subjects of anatomy, physiology, symptomatology and diagnosis, hygiene and sanitation, chemistry, histology, pathology, and principles and practice of chiropractic, including clinical instruction in vertebral palpation, nerve tracing and adjusting; and

- c. Passage of an examination prescribed by the State's chiropractic examiners covering the subjects listed in subsection b.
- 2. After June 30, 1974 Individuals commencing their studies in a chiropractic college after June 30, 1974, must meet all of the following additional requirements:
- a. Satisfactory completion of 2 years of pre-chiropractic study at the college level;
- b. Satisfactory completion of a 4-year course of 8 months each year (instead of a 3-year course of 6 months each year) at a college or school of chiropractic that includes not less than 4,000 hours in the scientific and chiropractic courses specified in subsection 1.b, plus courses in the use and effect of X-ray and chiropractic analysis; and
- c. The practitioner must be over 21 years of age.

70.7 - Interns and Residents

(Rev. 1, 09-11-02)

A. General

For Medicare purposes, the terms "interns" and "residents" include physicians participating in approved graduate training programs and physicians who are not in approved programs but who are authorized to practice only in a hospital setting; e.g., individuals with temporary or restricted licenses, or unlicensed graduates of foreign medical schools. Where a senior resident has a staff or faculty appointment or is designated, for example, a "fellow," it does not change the resident's status for the purposes of Medicare coverage and payment. As a general rule, services of interns and residents are paid as provider services by the intermediary.

B. Services Furnished by Interns and Residents Within the Scope of an Approved Training Program

Medical and surgical services furnished by interns and residents within the scope of their training program are covered as provider services. Effective with services furnished on or after July 1, 1987, this includes services furnished in a setting which is not part of the provider where a hospital has agreed to incur all or substantially all of the costs of training in the nonprovider facility. The Medicare intermediary is required to notify the carrier of such agreements. Where the provider does not incur all or substantially all of the training costs and the services are performed by a licensed physician, the services are payable on a fee schedule basis by the carrier. Prior to July 1, 1987, the covered services of interns and residents were paid by the carrier on a reasonable charge basis as physician services if furnished by a licensed physician off the provider premises regardless of who incurred the training costs.

C. Services Furnished by Interns and Residents Outside the Scope of an Approved Training Program-Moonlighting

Medical and surgical services furnished by interns and residents that are not related to their training program, and are performed outside the facility where they have their training program, are covered as physicians' services and paid on a fee schedule or reasonable charge basis where the requirements in the first 2 bullets below are met. Medical and surgical services furnished by interns and residents that are not related to their training program, and are performed in an outpatient department or emergency room of the hospital where they have their training program, are covered as physicians' services and paid on a fee schedule or reasonable charge basis where the following criteria are met:

- The services are identifiable physicians' services, the nature of which requires performance by a physician in person and which contributes to the diagnosis or treatment of the patient's condition;
- The intern or resident is fully licensed to practice medicine, osteopathy, dentistry, or podiatry by the State in which the services are performed; and
- The services performed can be separately identified from those services that are required as part of the training program.

When these criteria are met, the services are considered to have been furnished by the individuals in their capacity as physicians and not in their capacity as interns and residents.

80 - Health Maintenance Organizations (HMOs) Defined

(Rev. 1, 09-11-02)

An HMO for Medicare purposes is a public or private organization that provides, either directly or through arrangement with others, comprehensive health services to enrolled members. An HMO must service those who live within a specified service area. It must provide services based on a predetermined periodic rate or periodic per capita rate basis without regard to the frequency or extent of covered services it furnishes. An HMO must also meet other statutory requirements.

An HMO's service area is a geographic area in which a full range of its services are offered to its members. This geographic area differs from an HMO's enrollment area since it may include locations outside its service area where it offers less than its full range of services. (For example, an HMO may cover house calls in emergencies in its service area but not for members who live outside the service area.)

Section 1876 of the Act allows a Medicare beneficiary eligible for Part A and Part B, or Part B only, to choose to have covered items and services furnished through a Medicare qualified HMO. An HMO enters into a contract with the Secretary in order to participate under Medicare.

90 - Other Definitions

(Rev. 1, 09-11-02)

NOTE: We anticipate adding to this section as we find the need to define other terms.

90.1 - Supplier Defined

(Rev. 1, 09-11-02)

The term supplier means an entity that is qualified to furnish health services covered by Medicare, other than providers, physicians, and practitioners.

The following suppliers must meet the conditions in order to receive Medicare payment: ambulatory surgical centers (ASCs), independent physical therapists, mammorgraphy facilities, DMEPOS suppliers, independent occupational therapists, clinical laboratories, portable X-ray suppliers, dialysis facilities, rural health clinics, and Federally-qualified health centers.

An ASC is a distinct entity that operates exclusively for the purpose of furnishing outpatient surgical services to patients. It enters into an agreement with CMS to do so. An ASC is either independent (i.e., not a part of a provider of services or any other facility), or operated by a hospital (i.e., under the common ownership, licensure, or control of a hospital).

A DME supplier is an entity that furnishes DME and has a number assigned by the National Supplier Clearinghouse.

90.2 - Laboratory Defined

(Rev. 1, 09-11-02)

Laboratory means a facility for the biological, microbiological, serological, chemical, immuno-hematological, hematological, biophysical, cytological, pathological, or other examination of materials derived from the human body for the purpose of providing information for the diagnosis, prevention, or treatment of any disease or impairment of, or the assessment of the health of, human beings. These examinations also include procedures to determine, measure, or otherwise describe the presence or absence of various substances or organisms in the body. Facilities only collecting or preparing specimens (or both) or only serving as a mailing service and not performing testing are not considered laboratories.

90.3 - Practitioners Defined

(Rev. 1, 09-11-02)

Practitioners except for physicians are health professionals who may deliver covered Medicare services if the services are incident to a physician's service or if there is specific authorization in the law. The following practitioners may deliver services without direct physician supervision: nurse practitioners and physician assistants in rural health clinics, designated manpower shortage area or HMOs, qualified clinical psychologists, clinical social workers, certified nurse midwives, and certified registered nurse anesthetists.

90.4 - Group Practice Defined

(Rev. 1, 09-11-02)

A group practice is a group of two or more physicians and non-physician practitioners legally organized in a partnership, professional corporation, foundation, not-for-profit corporation, faculty practice plan, or similar association:

- In which each physician who is a member of the group provides substantially the full range of services which the physician routinely provides (including medical care, consultation, diagnosis, or treatment) through the joint use of shared office space, facilities, equipment, and personnel;
- For which substantially all of the services of the physicians who are members of the group are provided through the group and are billed in the name of the group and amounts so received are treated as receipts of the group;
- In which the overhead expenses of and the income from the practice are distributed in accordance with methods previously determined by members of the group; and
- Which meets such other standards as the Secretary may impose by regulation to implement \$1877(h)(4) of the Social Security Act. The group practice definition also applies to health care practitioners.

Medicare General Information, Eligibility, and Entitlement

Chapter 6 - Disclosure of Information

Table of Contents

- 10 The Privacy Act of 1974
 - 10.1 Disclosure of Information
 - 10.2 Procedures for Handling Requests
 - 10.3 Processing Freedom of Information Act (FOIA) Requests
 - 10.4 Reporting Freedom of Information Act Activity
- 20 Case Numbering
- 21 Fees for Information (Contractors)
 - 21.1 Fees
 - 21.2 Billing of Fees and Charges
 - 21.3 Advance Payment/Requests Fulfilled Without Remittance
- 30 Disclosure of Program Materials (Contractors)
 - 30.1 Manuals, PMs, and Other CMS Materials
 - 30.2 Internal Guidelines
- 40 Disclosure of Information about Identifiable Beneficiaries (Contractors)
 - 40.1 Prohibition Against Disclosure
 - 40.2 The Authority for Disclosure
- 50 Disclosure Necessary for Proper Administration of the Health Insurance Program
- 60 Disclosure of Beneficiary Specific Data by Contractors to Medicare Providers and Suppliers in Coordination with Social Security Administration (SSA)
 - 60.1 Telephone Requests from Institutional Providers, Physicians, Suppliers, and Other Providers

- 60.2 Written Inquiries from Institutional Providers, Physicians, Suppliers, and Other Providers
- 60.3 Online Access for Medicare Certified Providers
- 60.4 Beneficiary Outreach
- 70 Disclosure of Information after the Death of a Beneficiary (Contractors)
- 80 Disclosure to Third Parties for Other than Program Purposes (Contractors)
 - 80.1 Release for Title XVIII Claims Information for Complementary Insurance Purposes (Contractors)
- 90 Matching Intermediary/Carrier Health Insurance
- 100 Disclosure to State Agencies or to Contractors Acting for State Agencies Administering Programs receiving Grants in Aid (Contractors)
 - 100.1 Information That May Be Disclosed and Authorization Required
 - 100.2 Procedures for Supplying Title XVIII Information to State Agencies
 - 100.3 Disclosure to Title V and Title XIX Agencies of Information Indicating Unprofessional or Unethical Practices by Physicians and Other Practitioners
 - 100.4 Disclosure to Title XIX Agencies of Physician Charge Information
 - 100.5 Electronic Reporting of Crossover Claims
- 110 Disclosure to Peer Review Organizations (PROs)
- 120 Disclosure to the Internal Revenue Service (Contractors)
- 130 Disclosure to the Office for the Civilian Health and Medical Program of the Uniformed Services (TriCare) by Contractors
 - 130.1 General
 - 130.2 Type of Information
 - 130.3 Procedures for Release
- 140 Disclosure of Information About Providers and Suppliers by CMS
 - 140.1 Disclosure Necessary for Proper Administration of the Medicare Program (CMS)
 - 140.2 Disclosure of Medicare Reports
 - 140.3 Disclosure of Information About Named Physicians and Other Suppliers of Service (RO's)
 - 140.4 Disclosure of Names of Providers, Physicians, and Suppliers of Services
 - 140.4.1 Names of Providers Terminated from Program Participation (Carriers)

- 140.4.2 Names of Providers, Suppliers, and Physicians Found Guilty of Fraud (CMS)
- 140.4.3 Names of Physicians on Second Opinion Lists (Contractors)
- 140.4.4 Referral to State Licensing Boards, Medical Review Boards, and Professional Societies
- 140.4.5 Disclosure of Medicare Statistics
- 150 Disclosure of Reports About Named Contractors
 - 150.1 Disclosure by ROs of Annual Reports of Contractors Performance (RCP)
 - 150.2 Disclosure of Medicare Audit Reports to the Press and Public (Audit Agency)
- 160 Disclosure of Statistics (CMS or Contractors)
 - 160.1 Information That the Contractor May Disclose
 - 160.2 Information That May Be Released Upon Authorization by the Regional Office
- 170 Disclosure of Health Insurance Information by Providers
 - 170.1 Disclosure of Health Insurance Information to a Beneficiary or in Connection With a Claim
 - 170.2 Disclosure to Contractors
 - 170.3 Disclosure to Third parties for Proper Administration of the Health Insurance Program
 - 170.4 Disclosure to Third Parties for Other Than Program Purposes
 - 170.5 Disclosure of Claims Payment Information in Alcohol and Drug Abuse Cases
 - 170.6 Disclosure of Itemized Statement to an Individual for Any Item or Service Provided
- 180 Cost to a Provider That Requests Information Available to the Public

Exhibit A

Exhibit B

Exhibit C

10 - The Privacy Act of 1974

(Rev. 1, 09-11-02)

A. General

The purpose of the Privacy Act of 1974 is to provide safeguards for individuals against an invasion of privacy by Federal agencies. Among other things, Federal agencies are required to permit an individual to:

- Determine what records pertaining to the individual are collected, used, or disseminated by such agencies;
- Prevent records pertaining to the individual obtained by Federal agencies for a specific purpose from being used for another purpose without the individual's consent; and
- Gain access to information pertaining to the individual in Federal agency records, and to correct such records when appropriate.

Contractors considered to be Federal agencies for purposes of administering the Privacy Act must comply with its provisions. Contractors must:

- Inform each individual from whom information is requested of their rights under the Privacy Act;
- Describe the method of accessing an individual's records;
- Prepare a method of accounting for disclosures; and
- Devise a method of reviewing records at the request of individuals and making corrections if appropriate.

Additional information concerning Medicare privacy policies can be accessed on the CMS web site under the Privacy Policy link at the bottom of the web page.

B. Definition of "Individual"

"Individual" means a living person on whom CMS has any personal (as opposed to business) information. "Individual" does not include so-called persons such as sole proprietorships, partnerships, or corporations. Except for disclosure of and access to medical information about minors, a parent or legal guardian of a minor, or a legal guardian of someone the court has declared incompetent, has the same rights as the individual to the individual's records. No one may act on behalf of an individual who has not been declared incompetent by a court or gain access to his/her records under the Privacy Act without the individual's written consent.

C. Physicians' and Suppliers' Rights to Access Under the Privacy Act

Providers, physicians, and suppliers as business entities do not have access to business information about themselves under the Privacy Act, since the Privacy Act concerns individuals only. However, a physician or other supplier who is also an individual has the same access rights as any other individual to personal information maintained about themselves. Purely business information which is retrievable by the physician's unique identifier is not subject to the Privacy Act, but it may be disclosed to the physician to the extent that we would not deny the physician the information under the Freedom of Information Act (FOIA).

D. Informing Individuals of Rights

The contractor must inform the individual of his/her rights under the Privacy Act when it solicits any information directly from the beneficiary in connection with a Medicare claim. This would usually be

a result of the failure of the individual or provider to furnish all the information required on a claims form. When information is collected by telephone, the contractor will give the individual a brief oral explanation. When information is requested by mail a written notice is used.

E. Method of Accessing Individual Records - General

The law requires that an agency must inform an individual, upon request, whether a system of records contains a record pertaining to the individual, permit the individual to review such record and to be accompanied for the purpose of reviewing the record by a person of his or her choice. Further, the individual is permitted to obtain a copy of such record in a comprehensible form at a reasonable cost. There is a charge of 10 cents per page with all fees under \$25 waived. There is no charge for searching; nor is there a charge for a copy furnished as a means of permitting an individual access to their records

Each requester shall be asked for proof of identity as well as such general information as is necessary to determine where and how to look for records about the data subject. Within 10 working days of receiving a request, the contractor must decide whether to release the records. The requested information must be furnished within 30 working days unless good cause exists. An example of "good cause" would be an inactive record filed in a records center that cannot be obtained within 30 days. The requester should be told of the delay and given an approximate date to expect the information.

If an individual's request for information concerning themselves is not in the contractor's files, the contractor should advise the individual that it does not have the information and, if available elsewhere, that it is forwarding the request to the office which has it. The contractor should furnish any requested information available if the request cannot be satisfied in full. The contractor should identify requests under the Privacy Act that require referral to CMS for data in central office systems. As required by the Privacy Act, the contractor should send the request for processing, along with a copy of the interim response, to the systems manager listed in the Annual Publication: Systems of Records.

F. Methods of Accessing Individual Medical Records

The official responsible for the records, or their designated medical officer, may disclose medical information directly to the individual if the official determines, based on review of the medical evidence, that such disclosure is not likely to have an adverse effect. In such cases the responsible official will give the requested information to the individual and annotate the record to show that the disclosure was made.

If the responsible official determines that direct disclosure of the medical record would be likely to have an adverse effect on the individual or does not consider themselves qualified to make such a determination, the official will disclose the medical information to the representative designated in writing by the individual. The representative must be a medical professional (licensed medical practitioner or nurse) who would be willing to review the record and discuss it with the individual. The contractor will retain a record showing the reason for its determination and a copy of the correspondence transmitting the information to the individual or to his/her medical representative.

If there is sensitive medical information in the records and the individual refuses to name a medical professional or cannot afford the fee for the service, or if the designated medical professional refuses to serve, the contractor will refer the file to the CMS regional office.

G. Disclosure of Medical Information Relating to Minors

In order to protect the privacy of a minor, a parent or authorized guardian who requests access to the minor's medical record may not be given direct access to the record. Contractors shall ask a parent or guardian who requests access to such a record to designate a physician or other professional (other than a family member) to whom the record may be sent. The physician or health professional to whom the record is sent will be asked by the contractor to consider the effect that the disclosure of the record to the parent or guardian would have on the minor in determining whether the record should be made available to the parent or guardian.

The contractor will prepare a response to the parent or guardian in substantially the following form:

We have co	ompleted	processing	your	request for	or access	to t	he med	ical	record	is of
(name)				, a mir	nor					

The medical records have been sent to (name and address of designated health professional) in accordance with your instructions

In each case where a minor's medical records are sent to a physician or health professional, reasonable efforts will be made by the contractor to so inform the minor. In the event the parent or guardian refuses to name a medical professional or cannot afford the fee for the service or the designated medical professional refuses to serve, the contractor will refer the file to the CMS regional office for appropriate action.

H. Disclosure to Third Parties

The Privacy Act permits disclosure to any third party with the written consent of the individual to whom the record pertains. It also permits disclosure in certain instances without the individual's consent. However, the contractor must not disclose information unless the disclosure is specifically authorized in this chapter, or the individual has consented to the disclosure in writing, or the CMS regional office authorizes the disclosure.

I. Disclosures with Consent

Except for those disclosures discussed below in §10L of this chapter, Disclosure Without Consent, information may not be disclosed without the written consent of the subject individual, or their legal guardian, or, in the case of a minor, their parent (a parent or legal guardian of a minor may not consent to the disclosure of medical information about the minor). Other persons, regardless of relationship (except members of Congress and representative payees), may not receive information about the individual without their consent. In addition, a person who receives information about an individual with the individual's consent may not authorize disclosure of that information to someone else. Awkward situations may develop from a refusal, in compliance with the Privacy Act, to divulge information to the child or spouse of an aged or infirm individual. In order to avoid lengthy and unproductive correspondence, the requester should be advised that the Privacy Act precludes our disclosing any information to anyone other than the individual to whom the information pertains without the specific written consent of that individual, but that the requested information will be sent directly to the individual concerned with an explanation of the inquiry received on their behalf.

J. Disclosures to Members of Congress

Information requested by members of Congress and their staffs may be disclosed as follows:

- Where a member of Congress (or a staff member) inquires on behalf of a constituent, the contractor may respond to the member of Congress (MC) without the written consent of the individual, but must make a record of the disclosure.
- Where the MC's inquiry is on behalf of a relative of the subject individual (but the inquirer is not the legal guardian of the individual, or the parent of the individual minor), the contractor should advise the MC that it can respond directly only to the individual unless the individual furnishes written consent to release the information about themselves to the MC (if we respond directly to the individual or with their consent, no accounting record is required). Where the inquirer is the legal guardian, or the parent of a minor, the contractor may disclose the information to the MC, but must make a record of the disclosure.
- Where the MC's inquiry is in writing and does not indicate whether the request is from the subject individual, the contractor will contact the MC, usually by telephone, to clarify the situation.

A copy of the reply (or a report of telephone call if the response is by telephone) will provide an adequate accounting record. The record should show the date of disclosure, the information disclosed, and to whom the disclosure was made. The contractor should file this record by name or social security claim number so that retrieval may be made expeditiously.

K. Disclosure to Representative Payees

An SSA-appointed representative payee is entitled to receive information on, or act on, behalf of a beneficiary to the extent necessary to protect the beneficiary's rights under title II or XVIII.

L. Disclosure without Consent

Below are listed the situations in which data on identifiable individuals may be released without the individual's consent. The disclosure is permitted if the disclosure would be:

- To DHHS employees and officers who need the records to perform their duties;
- Required by the FOIA;
- To the Bureau of Census;
- For research purposes under certain circumstances;
- To the National Archives;
- For law enforcement activities if the activity is authorized by law and the request is from the head of the agency and specifies the particular record desired and the law enforcement activity for which the record is sought;
- For compelling circumstances affecting the health and safety of any person if notice of the disclosure is sent to the last known address of the individual;
- To either House of Congress or to any congressional committee or subcommittee (see section I above for requests from members of Congress on behalf of constituents);
- To the Comptroller General or an authorized representative to perform the duties of the General Accounting Office;

- Pursuant to the order of a court of competent jurisdiction;
- For a "routine use" A "routine use" is a disclosure of information which may be made without the individual's written consent because the disclosure is compatible with the purpose for which the information was collected in the first place. An explanation of the purposes and uses of each "routine use" disclosure must be published in the Federal Register at least 30 days prior to the disclosure and at least annually thereafter. Current routine uses are:
 - Part B Payment Records
 - Payment Record Reference File
 - Tape containing amount, type, and cost of health care services.
 - Summary Records
 - Group Health Plan Membership Data
 - Identification Data
 - Group Health Membership Data
 - Group Health Membership List
 - Quality Assurance Program Releases and Corrections

M. Accounting for Disclosure

The Privacy Act requires that agencies account for disclosures of personal data to organizations outside DHHS or made in response to requests under the FOIA. The purposes of the accounting are:

- To allow individuals to learn to whom records about themselves have been disclosed; and
- To provide a basis for subsequently advising recipients of any amended or disputed records.

The accounting must enable the contractor to tell the individual when and what information about the individual was released and to whom. It is not necessary to maintain a separate record of disclosures if such information can be retrieved from the contractor's operating records.

An individual has the same access rights to an accounting of disclosures as to other information, with no exception. It is not mandatory to tell an individual about disclosures made to an agency for law enforcement purposes. (Refer requests for accountings that include disclosures to law enforcement agencies to the CMS regional office.)

The contractor must maintain accounting records for 5 years, or the life of the basic record, whichever is longer. The lives of various basic records are furnished in Chapter 1 of the Claims Processing Manual

N. Reviewing and Correcting Records

Individuals are permitted by the Privacy Act to request correction of any record pertaining to themselves. The contractor should have a method of reviewing records at the request of the individual and making corrections where appropriate. In reviewing an individual's request to amend a record, the contractor should, whenever practicable, complete the review and advise the individual of the results within 10 days of the receipt of the request. Prompt action should be taken wherever possible to

reduce the administrative costs involved in issuing both a separate acknowledgment of the receipt of the request and a subsequent notice informing the individual of the action taken. If a contractor denies a request for correction, the individual can appeal to the system manager for the contractor's system of records.

O. Penalties

All employees must be aware of their responsibilities under the Privacy Act and guard against improper disclosure of personal information. Any officer or employee who willfully discloses individually identifiable information, the disclosure of which is prohibited by the Act, shall be guilty of a misdemeanor and fined not more than \$5,000.

P. Providers, Physicians, Suppliers, and the Privacy Act:

1. Confidentiality of Provider and Supplier Records

Medicare information may not be accepted from providers, physicians, and other suppliers of services on a confidential basis, expressed or implied, since any medical information obtained by a contractor is subject to disclosure to the individual to whom it pertains. Contractors are to make sure their providers and other suppliers of services are aware that no medical information marked "confidential" will be accepted on that basis and that any medical information received by the contractor may be disclosed to the patient or his/her representative upon request, either directly or through designated professional medical personnel. If a provider, physician, or other supplier of services documents medical findings on medical forms preprinted "confidential" or the provider or other supplier of services routinely stamps all records "confidential," such records, when transmitted to the contractor are to be accompanied by a signed statement to the effect that the provider, physician or supplier understands that the information is subject to disclosure at the request of the patient or his/her representative under the Privacy Act.

2. Release of Eligibility Data to Providers and Suppliers When the Individual is Unable to Sign an Authorization

In situations such as the admission to a hospital of an unconscious person, where the individual has not signed a statement authorizing the provider to pursue a Medicare claim, the disclosure of Medicare information to the provider should be treated as a routine use disclosure. If the individual has authorized the provider or supplier to pursue a Medicare claim, the release of information may be treated as a disclosure made with the beneficiary's consent.

10.1 - Disclosure of Information

(Rev. 1, 09-11-02)

The following sections contain instructions concerning the confidentiality and disclosure of information acquired and maintained by CMS, contractors (i.e., Medicare intermediaries and carriers), providers, and State agencies in the administration of the health insurance program. These instructions comply with the statutes and regulations governing disclosure of information, specifically section 1106 of the Social Security Act, the Freedom of Information Act (5 USC §552) and implementing HHS FOIA regulations at 45 CFR Part 5, and CMS Confidentiality and Disclosure regulations at 42 CFR 401.101, et seq.

10.2 - Procedures for Handling Requests

(Rev. 1, 09-11-02)

The Freedom of Information Act (FOIA) requires that within 20 working days of the receipt of a written request for records (or information known to be contained within an agency record), Federal agencies must decide whether the records/information will or will not be disclosed. By HHS regulation, only the CMS Freedom of Information Officer can make this decision, However, the CMS Freedom of Information Officer with the concurrence of the Office of the Assistant Secretary for Public Affairs, HHS, has delegated authority to Medicare contractors to directly release certain categories of frequently requested documents that the CMS Freedom of Information Officer has previously reviewed and decided to always release. These records are called "direct release" records. Some are identified within this manual; others are established by the CMS Freedom of Information Officer in administrative issuances.

Accordingly, within 20 working days of receipt of an FOI request, if the contractor is authorized to release all or at least some of the requested records, it will advise the requester in writing whether the request will be wholly or partially fulfilled.

CMS does not permit the contractor to issue a written denial of any request for information, except in response to the contractor's direct receipt of a state or local court subpoena duces tecum that seeks records on a beneficiary or individual practitioner that are contained in a Privacy Act System of Records, when such subpoena is not accompanied by a valid authorization to release the records signed by the subject of records. (See §40.1B.) CMS requires the contractor to refer to the RO all requests for which the contractor does not have clear authority to release the records/information.

Therefore, if the contractor receives a written request for records, the contractor must determine within 2 days of the receipt of the request whether it is clearly authorized to disclose the information. When able to furnish the requested materials, the contractor will furnish it whenever possible, within 20 working days. If it is not possible to furnish the materials within 20 working days, the contractor will immediately send the requester a substantive response that states that the records will be released, and explains the reason for the delay.

If the contractor is not clearly authorized to release the information, it notifies the RO immediately and refers the request to the RO for handling. It sends the request, along with one set of the requested records, immediately to the RO by first class mail, and clearly identifies the request as a Freedom of Information Act request.

The contractor acknowledges to the requester any request referred to the RO, using the following language:

Dear:

We have referred your (date) Freedom of Information Act request for (specify type of information requested) to: (name of the CMS Regional Office FOIA Coordinator, address of CMS RO, and telephone number of the CMS RO Coordinator).

Any questions you may have relative to your request should be directed to

that office.

Sincerely yours,

The contractor sends a copy of the acknowledgment letter with the request to the RO.

The contractor is not required to create records to comply with requests for information. Contractors may answer requests that require creation of records by stating that the requested records do not exist and that the FOIA does not require agencies to recreate records in order to respond to a request. Include the appeal rights statement set forth below in the response. (Note that deletion of non-releasable information from an existing record is not considered creation of a records, no matter how extensive or time consuming the deletion process might be. Respond to requests that require creation of records using the following language:

Dear:

This is in response to your request dated _____, seeking access to (specify the subject(s) of requested record). Because you seek access to one or more agency records, or information contained in such records, we have considered your request under the Freedom of Information Act (FOIA) (5USC §552).

We are unable to comply with your request because the agency does not maintain a records that is responsive to your request and because, under FOIA, we are not required to create records or to furnish information in a particular form or format when a record in such form or format is not readily reproducible employing reasonable efforts.

If you have reason to disagree with this decision, you may appeal. Your appeal should be mailed, within 30 days of the date of this letter, to the Deputy Administrator, Centers for Medicare & Medicaid Services, Room C5-16-03, 7500 Security Boulevard, Baltimore, Maryland 21244-1850. Please mark your envelope "Freedom of Information Act Appeal," and enclose a copy of this letter.

10.3 - Processing Freedom of Information Act (FOIA) Requests

(Rev. 1, 09-11-02)

Form CMS-632-FOI (Exhibit A) serves as a cover sheet for FOI requests. The form is designed to expedite the handling of requests and to provide the information necessary for reporting purposes. For each FOI request, the contractor will complete a Form CMS-632-FOI. Completion instructions for the form are included on the reverse side. The contractor may furnish estimated costs if actual postage costs are not readily available.

Order Form CMS-632-FOI routinely on an annual basis using the CMS 1961 Forms Order. Direct any interim requests for Form CMS-632-FOI to CMS Forms Distribution Officer, Forms Management and Distribution, Office of Internal Customer Support. Interim requests must be submitted in writing.

10.4 - Reporting Freedom of Information Act Activity

(Rev. 1, 09-11-02)

The contractor's authority to directly release certain kinds of records includes the responsibility for complete and accurate reporting on the use of this authority. The following are reporting requirements for Medicare contractors:

- Fill out completely and accurately a Form CMS-632-FOIA for each "direct release."
- Maintain in either hardcopy or electronic form a Freedom of Information Daily Log on a daily basis. This log must include the case control number, date of incoming letter, requester's name, subject of the request, the date received in the FOIA unit, the date the response was mailed and the invoiced fee.
- Prepare, on a monthly basis, the Summary Sheet for the CMS Monthly FOIA Report (Exhibit B). This Summary Sheet aggregates data from the individual Form CMS-632s.
- Submit the Daily Log and Summary Sheet for a given month to your RO FOIA Coordinator by the end of the fifth work day of the following month.
- Retain copies of all completed Forms CMS-632-FOI in accordance with record retention requirements for FOIA Administrative Files.

20 - Case Numbering

Since many requesters make multiple FOIA requests, name identification is not adequate to assure proper tracking, or fiscal accounting. Therefore, CMS employs a Case Numbering System which includes unique identifiers for each releasing activity.

- Use the Case Numbering System for all FOIA cases.
- Use the case number in item one of the Form CMS-632-FOI and in the "case number" block at the top left of the Form CMS-633 Invoice of Fees for FOIA Services (Exhibit C). Requesters will be asked to put the case number on checks rendered for payment of FOIA services and on any correspondence relating to that case.
- 1. All case numbers in this system consist of ten characters. These characters maybe numeric or alphabetical and are determined as follows:
- a. Case numbers for requests responded to directly by contractors begin with an arabic number for the Region.

Boston	1	Dallas	6
New York	2	Kansas City	7
Philadelphia	3	Denver	8
Atlanta	4	San Francisco	9
Chicago	5	Seattle	0

NOTE: Only one space is designated to indicate the responding regional office or contractor within the region. Therefore, Seattle is represented by "0" rather than by "10".

b. For example, a request responded to directly by any carrier within Region 1 would begin with the digit "1":

1-----

c. In all case numbers, the second character is the last digit of the calendar year in which the request is received by the responding activity. Thus, continuing the example above, for a 2001 request in Region 1:

11-----

d. Characters three through six indicate the contractor within the region. A request answered by a contractor would use the last four digits of the contractor number in positions three through six. Thus, a request responded to directly by Blue Cross and Blue Shield of Rhode Island (carrier number 00870) would be:

110870----

e. The last four characters indicate the consecutively assigned case number, beginning with "0001". Thus, the first request responded to by that carrier in calendar year 2001 would be:

1108700001

21 - Fees for Information (Contractors)

(Rev. 1, 09-11-02)

Under the provisions of the Freedom of Information Act, certain fees and charges have been established to recover some of the cost of disclosing information to the public. Providers, contractors, and State agencies are required to pay appropriate fees for copies of reports they request pertaining to other providers, contractors, or State agencies. Such fees are not reimbursable administrative costs under the Medicare program for contractors or State agencies. A provider may claim such fees as allowable costs only if it demonstrates that the information is necessary in developing and maintaining the operation of patient care facilities and activities. Members of Congress, when clearly requesting information on behalf of a constituent or other third party, are subject to the same fees and charges that would apply to the person represented. The contractor will not charge the public for inspection of disclosable documents, or for requests which result in charges under the minimum described in §21.3, below.

21.1 - Fees

(Rev. 1, 09-11-02)

- 1. Fees are to be charged differentially, depending on the category of requester/use of the requested material, as follows:
- <u>Commercial Use Request</u>- If the request is for a commercial use, charge the requester for search and review, plus duplication.

- Education and Scientific Institutions and News Media- If the requester is (1) an educational institution or a non-commercial scientific institution operated primarily for scholarly or scientific research, or (2) a representative of the news media, charge only for duplication. However, do not charge the first 100 pages of duplication.
- Other Requesters- If the request fits into neither of the above categories, charge for search and duplication. However, do not charge for the first 2 hours of search time and the first 100 pages of duplication. In addition, do not charge the requester if the charge to be billed is less than the established billing threshold.

NOTE: The CMS Freedom of Information Officer can rule on requests for waiver or reduction of fees, other than those listed above.

- 2. Regardless of the actual cost of responding to the requester, the law only permits charges to be levied for certain services and only in accord with a published FOIA fee schedule. The DHHS schedule at 45 CFR Part 5 is binding on CMS, and results in the following charges:
- a. Photocopying Charge 10 cents per page for copying records. The contractor may charge lower fees for particular documents where:
- The document has already been printed in large numbers,
- The contractor determines that using existing stock to answer this request, and any other anticipated FOI requests will not interfere with program requirements, and
- The contractor determines that the lower fee is adequate to recover the prorated share of the original printing costs.
- b. Search and Review For manual searching and for review (when used, as applicable, in connection with processing records for a commercial use or other request), base the cost on the hourly rate of employees involved. Equate employee hourly wage scales to these three categories:

	Hourly Wage Range	GS Grade Range	CMS Bills
Level 1	Up to \$20.74	Federal Grades 1 through 8	\$16.00
Level 2	\$20.75 - \$46.69	Federal Grades 9 through 14	\$33.00
Level 3	\$46.70 or more/hr	Federal Grades 15 and up	\$59.00

NOTE: The CMS Freedom of Information Officer updates the above scale based upon changes to the HHS fee schedule, and provides such updates to all CMS/contractor FOIA Coordinators.

The contractor will not charge the requester any fee at all if the costs of routine collection and processing of the fee are likely to equal or exceed the amount of the fee. As of November 2001, CMS's charge threshold is \$15.00.

If the contractor determines that the requester is breaking down a single request into a series of requests in order to avoid (or reduce) the fees charged, it may aggregate all these requests for purposes of calculating the fees charged.

Certifying that records are true copies - Charge \$10 per certification.

Performing any other special service requested, that the contractor agrees to provide - actual costs of operating any machinery, plus actual cost of any materials used, plus charges for the time of its employees, at the rates given in paragraph 2b of this section.

Special Forwarding Arrangements - Actual cost. This includes special delivery, airmail, registered mail, etc.

21.2 - Billing of Fees and Charges

(Rev. 1, 09-11-02)

The contractor is responsible for the billing of costs resulting from providing disclosable information. To invoice, use Form CMS-633 – Invoice of Fees for FOIA Services (Exhibit C). The requester sends payment derived from the release of information to the CMS Division of Accounting.

21.3 - Advance Payment/Requests Fulfilled Without Remittance

(Rev. 1, 09-11-02)

Where estimated costs exceed \$250.00, or if the requester has failed to pay previous bills in a timely fashion, collect fees and charges before the requested material is copied and furnished. Use the Division of Accounting, CMS's FOIA Bad Debtor Listing to determine delinquency. Notify the requester of the total estimated charges, using language similar to the following:

HHS regulations require that a charge be made to recover some of the costs incurred in the disclosure of information. The total (estimated) billable cost for the material you requested is \$______. The cost consists of the following charges: (list here breakdown of charges; e.g., "Photocopying: \$10; Searching: \$450). Please send a check or money order for the total amount made out to CMS and mail it to (contractor address).

The material will be mailed to you upon receipt of payment (subject to any adjustment to the estimate). If we do not hear from you within 30 days, we will assume your request has been withdrawn and no further action will be taken.

Be sure to point out to the requester that even though the invoice says to mail the check to the CMS Division of Accounting in Baltimore, in this case the check must be mailed to the contractor so that payment can be verified before the material is mailed. Subsequent to the contractor's receipt of the advance payment, send both the check and the finance copy of the invoice to the Division of Accounting in Baltimore. Because some requesters will neither provide advance payment nor notify the contractor that they no longer desire the requested material, do not send the "Finance" copy of the invoice to Accounting until after the advance payment check is received. For the same reason, do not begin searching for and photocopying the requested material until the advance payment is received. If you do not hear from the requester in 30 days, close the request.

30 - Disclosure of Program Materials (Contractors)

(Rev. 1, 09-11-02)

The Freedom of Information Act deals with the right of the public to information about Government rules and methods of operating. It requires that every Federal agency make available for inspection and copying:

"522(a)(2) (B) those statements of policy and interpretations which have been adopted by the agency and are not published in the Federal Register; and (C) administrative staff manual and instructions to staff that affect any member of the public, unless such materials are promptly published and copies offered for sale."

Information contained in Medicare program manuals and Program Memoranda can be obtained via the Internet. Where requests are made to contractors, contractors will photocopy Program Memoranda and individual pages from CMS manuals to respond to minor requests. When inquirers ask to see Social Security Manuals and letters that are not maintained, the contractor will refer them to the SSO. If the request is for Medicare related information, the contractor may disclose the following:

- Written policy used to evaluate and make payment decisions on claims subject to medical review. (Parameters are not components of medical payment policies and should not be disclosed.)
- Fee schedules and profiles of customary and prevailing charges of physicians/suppliers, including backup data, as long as the provider identifiers and the frequencies of procedures per provider have been deleted.
- Relative value units and conversion factors used to develop approved charges for procedures done too rarely to compile customary charges.
- Lists of contractors, physicians/suppliers and provider numbers.
- Documents listing and/or describing acceptable diagnostic or procedure codes, rate schedules or covered services.

The contractor will not engage in interpretive analysis or discussion of the material when responding to requests.

The contractor will not release information on the parameters and computer edits used to identify claims for medical review. These are tools for selection of claims for MR and are not determinants of whether a service is payable.

EXAMPLE: The contractor has a screen which identifies for review all claims for CAT Scans of the head in excess of 3 in 30 days. Written MR policy contains diagnoses and other descriptions of medical conditions that would justify CAT Scans in excess of that amount. The contractor is required to release the policy guidelines but must withhold the "3 in 30 days" parameter used to identify claims for MR.

The contractor must refer requests for information for which denial is recommended, including parameters and computer edits to the RO.

30.1 - Manuals, PMs, and Other CMS Materials

(Rev. 1, 09-11-02)

The contractor should provide a facility for the public to view the program manuals and other information available to the public on the Internet.

It will fulfill requests for printouts resulting in 50, or fewer, pages.

Requests to examine manuals or letters usually arise where a beneficiary calls at an office to discuss his/her claim. In such situations, the contractor will abstain from engaging in interpretative analysis or discussion of the material

30.2 - Internal Guidelines

(Rev. 1, 09-11-02)

All interpretive materials, guidelines, and clarification of policies that relate to payment of Medicare benefits must be released.

Upon request, the contractor will release written guidelines used to evaluate and make payment decisions on claims subject to MR. Precise information on the screens, parameters and computer edits used to identify claims for MR may be withheld.

EXAMPLE: A screen identifies for review any SNF claim with more than 30 covered days. Written MR policy contains the documentation standards and guidelines which assist the reviewer in the decision to pay or deny the claim.

Release of the policy guidelines is required, but the review screen can be withheld.

Guidelines may be withheld that use specific criteria and tolerances intended to identify claims that raise a strong possibility of fraud or abuse because of a pattern of payment requests for excessive or duplicate services, or services not rendered. If disclosure of such guidelines could be judged to adversely affect the program by allowing violators to go undetected, the contractor should refer the request to the RO with its recommendations. The contractor will acknowledge any requests that should have gone to another contractor and refer them to the appropriate contractor.

40 - Disclosure of Information about Identifiable Beneficiaries (Contractors)

(Rev. 1, 09-11-02)

Sections 3764 - 3769 set the guidelines for disclosure of information about a named beneficiary, to whom it may be disclosed, and the purposes for which it may be disclosed. In general, no information may be released except to the beneficiary (or the beneficiary's legal guardian) without the beneficiary's (or legal guardian's) explicit written authorization.

In cases requiring the beneficiary's consent, the authorization may be in any form, but it must:

- Be signed and dated;
- Specify the individual, organizational unit, class of individuals or organizational units to which the information may be disclosed;

- Specify the record(s), information, or type(s) of information which may be disclosed; and
- Indicate whether the consent is a one-time or on-going release of records.

A contractor will not honor a blanket consent to disclose all beneficiary records to unspecified individuals or organizational units.

The disclosure of information about beneficiaries is governed by the provisions of § 1106 if the Social Security Act as implemented by Regulation No. 1, the Privacy Act, the FOI Act, and the DHHS Public Information Regulation.

40.1 - Prohibition Against Disclosure

(Rev. 1, 09-11-02)

Section 1106 of the Social Security Act prohibits disclosure of any file, record, report, or other paper, or any information obtained at any time by the Secretary or an officer or employee of DHHS in the course of discharging their duties under the Act, except as prescribed by regulations. Where manual instructions permit disclosure, assume that related regulations have been published. The same prohibition applies to information received by any person outside DHHS, from the Secretary, or an officer or employee of DHHS.

The prohibition applies to any agency, organization (e.g., contractors), or institution, or any of its officers or employees, in the fulfillment of a contract or agreement with the Secretary.

A. Disclosure of Provider or Physician Records

The prohibition also relates to any information received from DHHS, a contractor, or any person or entity that furnishes services under arrangements with a provider or accepts an assignment under the program. However, patient records in the possession of a provider or physician are not subject to the prohibition against disclosure or to the Departmental rules and regulations concerning confidentiality merely because the patient is entitled to Medicare benefits. Disclosure of provider or physician records not in the possession of CMS or a contractor may, however, be subject to applicable State or local laws, or to hospital rules governing disclosure.

B. Authority for Refusal to Disclose Information

Denial of all or portions of requested records can only be made by the CMS Freedom of Information Officer. Therefore, when a request for information is received, disclosure of which is prohibited under these guidelines, the contractor will follow §10.2 above.

The CMS FOIA Officer has authorized Medicare contractors to issue a denial on his/her behalf in the following situation: If any officer, employee, agent or subcontractor is served a subpoena or other compulsory process requiring the production of records or information on a beneficiary or individual practitioner that are contained in a Privacy Act System of records and such a request is not accompanied by a valid authorization to release the records signed by the subject of the records, he/she will decline to produce the records or information. He/she will base the refusal on §1106 of the Social Security Act and on 5 USC §552a, 5 USC §552, 45 CFR Part 2 and 45 CFR Part 5. The contractor will notify the RO immediately.

If the contractor directly receives such a subpoena, it uses the following language to respond:

Dear Sir or Madam:

This is in response to the subpoena duces tecum, dated _____, initiated by your firm, for certain Medicare records in our possession.

The Department of Health and Human Services regulation at 45 CFR Part 2 states, among other things, that the Department will treat subpoenas duces tecum for records in its possession as requests under the Freedom of Information Act (5 U.S.C. §552).

Because the records the subpoena seeks are in a Privacy Act system of records, the Privacy Act (5 U.S.C. §552a) precludes release of those records except pursuant to a written authorization to release signed by the subject(s) of the records or unless the Freedom of Information Act requires release of the records or a court of competent jurisdiction orders release. Regarding the latter condition of disclosure, for purposes of the Privacy Act, a court of competent jurisdiction is a <u>Federal court only</u>.

Review of this matter indicates that your firm has not presented a written authorization to release records signed by the subject(s) of the records. Moreover, your firm's subpoena is not an order of a court of competent jurisdiction, and 45 CFR Part 2 requires us to treat the subpoena duces tecum as a Freedom of Information Act request. Further the Freedom of Information Officer for the Centers for Medicare & Medicaid Services has determined that the requested records are exempt from mandatory disclosure under the Freedom of Information Act by exemption (b)(6) of that Act. Exemption (b)(6) permits the withholding of information about individuals in personnel and medical files and similar files, when the disclosure of such information would constitute a clearly unwarranted invasion of personal privacy.

Based upon the foregoing, we respectfully decline to produce the Medicare records requested by your firm's subpoena duces tecum.

If you have reason to disagree with this decision, you may appeal. Your appeal should be mailed, within 30 days of the date of this letter, to the Deputy Administrator, Centers for Medicare & Medicaid Services, Room C5-16-03, 7500 Security Boulevard, Baltimore, MD 21244-1850. Please mark your envelope "Freedom of Information Act Appeal," and enclose a copy of this letter.

Sincerely yours,

Signature of Authorized Official

cc: FOI Officer, CMS

Process the State or local court subpoena duces tecum that seeks other kinds of CMS records in accord with §10.2 above.

C. Penalty for Failure to Comply With the Rules Relating to Disclosure of Information Obtained in the Administration of the Act

Section 1106(a) of the Act provides that any person who violates the disclosure provisions shall be deemed guilty of a misdemeanor and, upon conviction thereof, shall be punished by a fine not exceeding \$1,000, by imprisonment not exceeding 1 year, or both.

40.2 - The Authority for Disclosure

(Rev. 1, 09-11-02)

Regulation No. 1 of the Act defines the basic authorization for disclosure of information obtained in the administration of the program. The general rule is that information about an individual obtained in the administration of the program may not be disclosed without the authorization of that individual. Medical information relating to an individual will generally be disclosed under more restrictive conditions than other information and, where permitted, usually may be furnished only upon the written authorization of the individual. Specific exceptions to this rule are detailed in the following sections. As far as program operations are concerned, information about an individual may be disclosed without the individual's authorization when the disclosure is necessary in connection with any claim, or other proceeding, under the Social Security Act.

Information will be disclosed for other than program purposes only if the disclosure is authorized by Regulation No. 1 and is consistent with the proper and efficient administration of the program.

50 - Disclosure Necessary for Proper Administration of the Health Insurance Program (Rev. 1, 09-11-02)

A. Information About an Individual

Disclosure of any record, report, or information about an individual may be made without the individual's authorization if it is in connection with any claim, or other proceeding, under the Act when it is necessary for the proper performance of duties of:

- Any officer or employee of the Department; or
- Any officer or employee of a State agency, contractor, provider of services, or other agency or organization participating in the administration of the program by contract or agreement in carrying out such contract or agreement.

SSOs have a responsibility for public information activities. In its development of a human interest story concerning health insurance, an SSO may, on occasion, request the contractor to provide claims reimbursement information about a specific beneficiary. The contractor will honor such a request when it comes from an SSO employee having authority for liaison with the contractor.

Depending upon the agreements made with the parallel SSO, the employee authorized may be the manager, assistant manager, staff assistant, and/or HI coordinator. The contractor will not honor requests by other employees for beneficiary claims information.

B. Disclosure to Third Parties

In the administration of the program, the contractor may want to avail itself of the services of third parties such as auditors, court reporters, public stenographers, microfilm processors, or companies developing equipment for use in the program.

The arrangement between the contractor and such third parties, even where it is of short duration, is in the nature of an agreement supplementing its contract with the Secretary. Under such an arrangement, disclosure to these parties of necessary information that relates to, and is used in, the administration of title XVIII of the Social Security Act is permitted as provided by §1106 of the Social Security Act and Regulation No. 1.

However, when the contractor enters into an agreement with these parties, it will inform them that §401.1 of Regulation No. 1 precludes the disclosure of any information on identifiable individuals. It will call their attention to the penalty clause of §1106(a). It will call attention to these provisions by letter (see below), and obtain a written agreement to comply with the disclosure provisions before releasing any information.

Sample Letter to Third Party

Dear

We are prepared to make available to your organization certain social security records so that (state reason).

However, before making these records available, we must point out that by law, all information derived on identifiable individuals in the administration of the Social Security Act is confidential and may be disclosed to others only under very restrictive circumstances. Regulation No. 1 of the Social Security Administration which governs disclosure of official records and information precludes you from disclosing any information on identifiable individuals.

You should also note that §1106(a) of the Social Security Act imposes criminal penalties for unauthorized disclosure.

Any social security records which do not identify or make identifiable any individuals are not subject to these restrictions, but may be restricted under the provisions of the Freedom of Information Act.

In order to comply with the requirements of the Social Security Act and Freedom of Information Act you must agree to the following conditions before we can make any social security records available to you:

Any information which is turned over to you will be used only for the specific purpose intended.

All employees having access to this material will be instructed as to its confidential nature.

An official of your company will assume the responsibility for ensuring that the information is not revealed to another.

The material must be returned to us as soon as you have finished the job.

We must have a letter from a responsible official in your office agreeing to these conditions and assuming responsibility for carrying them out.

Upon receipt of such letter, the social security records will be made available to you.

Sincerely yours,

60 - Disclosure of Beneficiary Specific Data by Contractors to Medicare Providers and Suppliers in Coordination with Social Security Administration (SSA)

(Rev. 1, 09-11-02)

The contractor will use these guidelines to handle requests for beneficiary-specific information from providers, physicians, and suppliers. In general, it will emphasize that the Medicare card is to be used to obtain the necessary information for filing Medicare claims; that it contains the health insurance claim (HIC) number and entitlement dates; and is correct in the majority of cases.

60.1 - Telephone Requests from Institutional Providers, Physicians, Suppliers, and Other Providers

(Rev. 1, 09-11-02)

The standard method for sharing eligibility information with providers is through electronic data interchange (EDI). EDI is the most efficient and cost-effective way for intermediaries and carriers to make this eligibility information available to providers and ensures privacy safeguards through EDI agreements that bind providers. Instructions regarding provider EDI access to limited eligibility information and the specific data elements that can be disclosed can be found in the Claims Processing publication, Admission and EDI Support chapters.

Carriers and intermediaries are permitted to release eligibility information over the telephone to providers, subject to ensuring the protection of the beneficiary's privacy rights.

NOTE: Contractors should not automatically assume that a provider who submits claims electronically can verify eligibility electronically. For example, contractors provide free or low-cost software to some providers enabling electronic claims submission. That software, however, does not provide for electronic eligibility queries.

The eligibility information that may be released by telephone is limited to only that information that would be available via EDI.

Before releasing eligibility information to providers via the telephone the intermediary or carrier must validate the provider's name and number, <u>and</u> obtain the following information:

- Beneficiary last name and first initial;
- Beneficiary date of birth;
- Beneficiary Health Insurance Claim (HIC) number; and,
- Beneficiary gender.

These items must match exactly.

Information may be released via the Automated Response Unit (ARU) or Interactive Voice Response (IVR) systems.

60.2 - Written Inquiries from Institutional Providers, Physicians, Suppliers, and Other Providers

(Rev. 1, 09-11-02)

If the written inquiry is not accompanied by a written consent statement signed by the beneficiary or representative (and containing all of the elements indicated in §60.1 above), the contractor informs the requestor, in writing, that the information may not be released, and to first contact the beneficiary for the information. It tells the requestor that if that does not work, then the requestor should obtain a written and signed consent statement from the beneficiary containing all of the elements outlined in §60.1 above.

60.3 - Online Access for Medicare Certified Providers

(Rev. 1, 09-11-02)

The contractor should remind Medicare certified providers that it is providing online access to certain entitlement data for those certified providers who bill electronically. The contractor should be contacted for further information.

60.4 - Beneficiary Outreach

(Rev. 1, 09-11-02)

When making presentations before beneficiary groups, the contractor reminds them to carry their Medicare card with them whenever they are away from home and to always show their Medicare card when they receive services that are covered by Medicare. It refers beneficiaries to SSA for a replacement card if the beneficiary's Medicare card is lost or misplaced. If the beneficiary authorized the provider or supplier to pursue a Medicare claim, the contractor treats the release of information as a disclosure made with the beneficiary's consent.

70 - Disclosure of Information after the Death of a Beneficiary (Contractors)

(Rev. 1, 09-11-02)

The contractor may disclose information concerning the fact, date, or circumstances of death when efficient administration permits. The request must be in writing and must state why the information is sought. Additional information may be disclosed to a surviving relative or an authorized representative of that relative, the legal representative of the estate, or to a probate court for the purpose of appointing a legal representative of the decedent's estate. The contractor must exercise care, however, to see that it discloses no information which would appear to be detrimental to the individual or the individual's estate, i.e., unusual place or manner of death or information which could create a liability to the individual's estate.

The contractor may disclose medical information, in form and detail consistent with proper and efficient administration of the program, when such disclosure is reasonably necessary for a title XVIII purpose. (See §50 of this chapter for further information concerning the disclosure of non-medical and medical information when necessary for a program purpose.)

The contractor may disclose medical information obtained in the administration of title XVIII to a surviving relative or legal representative of the estate of the individual or to others for other than a title

XVIII purpose, when such information is necessary for a determination as to what supplementary benefits or services such deceased individual was eligible to receive under a private or public hospital or medical insurance program which is consistent with the purposes and objectives of title XVIII. The contractor may disclose such information only if the individual has consented to such disclosure or a surviving relative or the legal representative of the estate consents. (See §80.1 "Release of Title XVIII Claims Information for Complementary Insurance Purposes" in this chapter.)

80 - Disclosure to Third Parties for Other than Program Purposes (Contractors)

(Rev. 1, 09-11-02)

A. Non-medical Information

Non-medical information about an individual may be disclosed for other than a program purpose, to persons or organizations designated by the individual, if the individual authorizes disclosure, and if disclosure is consistent with the proper and efficient administration of the Act.

B. Disclosure of Medical Information to a Physician, Institution, or Other Supplier of Services

The contractor may disclose medical information to a supplier of medical services solely for the purpose of the beneficiary's care or treatment. Such supplier will be informed that the information is being furnished in connection with the treatment of the beneficiary and that its use should be restricted to that purpose.

C. Disclosure to the Source of Medical Evidence

Occasionally in the course of reviewing medical evidence submitted to substantiate a claim, contractor medical personnel may discover a medical condition of which the source of the information is unaware. (For purposes of this section, the "source" of the information is the part submitting the evidence.) In cases where a serious or potentially serious condition is found, the contractor may wish to inform the source of the information. If the source of the information indicates that the beneficiary is not institutionalized and the identity of the beneficiary's current physician is not known, the source should recommend that the beneficiary consult a physician or institution of the beneficiary's choice for further treatment.

D. Disclosure to a Source of Medical Services other Than the Source of the Information

Medical information may also be disclosed, upon request, to the beneficiary's physician or to a medial institution at which the beneficiary is or was a patient when such physician or institution is not the source of the information. However, consent for the release must be obtained from the beneficiary.

80.1 - Release for Title XVIII Claims Information for Complementary Insurance Purposes (Contractors)

(Rev. 1, 09-11-02)

The contractor may not release or use information obtained in the administration of the Medicare program for non-program activities. However, when the beneficiary has given written authorization the contractor is permitted to release certain information to its complementary insurance program under specific conditions in its capacity as insurance writer or administrator, or to other insurers for complementary health benefits purposes. Under no circumstances may the contractor use the

knowledge of an individual's entitlement or benefit utilization information for purposes of dropping an individual from a group health insurance plan.

A. Information That May be Released

Subject to necessary authorizations, copies, extracts, or summaries of only the following records may be released:

- Provider billing forms (e.g., Form CMS-1450);
- Explanation of benefits for Part B Provider Services, or denial letters; and
- Information on date of entitlement to Part A, date of enrollment under Part B, or the date Part B coverage began.

Requests for other information desired for complementary insurance purposes should be referred to the RO.

B. Form of Authorization

The contractor must make certain that information is not released without the required authorization. This authorization may be either indicated on the billing and admission form or on a dated statement from the beneficiary. Where the authorization is on a dated statement it must:

- Authorize release of information about his/her title XVIII Medicare claim;
- Designate to whom the release is authorized;
- Show that the release authorized is for complementary insurance purposes (this may be implied by the designation in the second bullet of this section);
- Indicate whether the authorization is for a one-time or ongoing release of data (i.e., for the duration of the claims and appeals process, but not to exceed 2 years); and
- Bear the signature of the beneficiary, the beneficiary's legal guardian, or the beneficiary's authorized representative.

Where the bill contains a beneficiary signature or indicates that the beneficiary's signature authorizing release of information is contained in the provider's records, the bill must contain the name of the complementary insurer unless a dated statement as outlined above is already on file.

Where the contractor has a dated statement on file, information may not be released for individual claims when the beneficiary indicated that they do not want disclosure on that claim. This may be indicated on certain claims forms by the beneficiary checking an appropriate block, or by attaching a separate statement.

If someone other than the beneficiary, legal guardian, or authorized representative has authorized disclosure to a third party by filling out the appropriate item on the billing form, the contractor may not release the information unless an authorization from the beneficiary is on file. Instead, where feasible, the contractor will inform the beneficiary (or legal guardian or authorized representative) that the signature on the claims form does not constitute a proper authorization for disclosure and that, if the beneficiary desires information to be disclosed to a third party, the beneficiary should send to either the

contractor or the third party, a statement authorizing release. (The contractor will indicate the necessary contents of a proper authorization or enclose an appropriate form or statement for use by the proper party in authorizing release.)

If invalid authorizations are a frequent problem, the contractor advises third party payers to obtain appropriate ongoing consent statements from enrollees.

C. Methods of Handling Requests

Where the complementary insurers desire title XVIII information for certain claims only, either the complementary insurer or the beneficiary may request its release.

The complementary insurer must furnish the required authorization(s) for release and must pay any charges. (The Medicare program will absorb charges for supplying duplicate MSN's or billing forms to beneficiaries, their authorized representatives, and to social security offices.) In the absence of a standing arrangement, the mere presence of an "authorization' to release and the identification of a complementary insurer on a title XVIII billing form does not constitute a request for the "release" of information. There must be a specific request for the information.

The contractor may enter into a standing arrangement with a complementary insurer to provide and charge for title XVIII information, such as MSNs or deductible non-met letters, in every case in which the contractor receives a title XVIII claim which contains the necessary authorizations and identifies the complementary insurer.

D. Release of Title XVIII Claims Information for Complementary Insurance Purposes by Providers

Contractors should be aware that, subject to specific written beneficiary authorization, providers are permitted to furnish certain limited information about Medicare eligibility status and related claims information to third part payers for complementary insurance purposes. (See §170.)

90 - Matching Intermediary/Carrier Health Insurance

(Rev. 1, 09-11-02)

In using an integrated claims processing system, it is permissible to match a currently updated subscriber file against a run of currently processed Medicare claims to extract information required to pay the complementary claim provided:

- When routinely securing beneficiary authorization for an ongoing release of information safeguards are maintained to ensure that, where a beneficiary revokes the authorization on the billing form or by a separate statement, there will be no release of information;
- When a statement is not routinely secured for the ongoing release of information, controls ensure that there is an authorization per §80.1 of this chapter from each Medicare beneficiary with complementary insurance before information is released; and
- Controls ensure that no information will be released in the matching process for any Medicare beneficiary who does not have complementary insurance.

Contractors must update the subscriber file to incorporate all terminations before each match with the Medicare claims file so that information will not be passed to the private business sector for beneficiaries who are not entitled to the complementary insurance plan.

The contractor may not permit the private business sector to have access to the Medicare claims history file. The private business sector may use only information concerning the current claim from the current run of processed claims.

100 - Disclosure to State Agencies or to Contractors Acting for State Agencies Administering Programs Receiving Grants in Aid (Contractors)

(Rev. 1, 09-11-02)

100.1 - Information That May Be Disclosed and Authorization Required

(Rev. 1, 09-11-02)

The contractor may disclose information on such matters as entitlement, benefit payment, or benefit utilization relating to an individual who has applied for benefits administered by the State welfare departments. It may make such disclosure without the authorization of the individual or the individual's legal representative (i.e., legal guardian appointed by a court or a parent or a minor) to any duly authorized officer or employee of a State agency administration of grants-in-aid programs under titles IV, V, or XIX or in the case of Puerto Rico, Guam, or the Virgin Islands, titles I, X, or XVI of the Social Security Act with the authorization of the individual or their authorized representative. Medical information beyond that shown on billing forms relating to an individual (and obtained in the administration of the Medicare program) may be disclosed to a State agency administering the grants-in-aid program if the State agency shows a special need for the information, and the file contains a written authorization by the beneficiary specifically consenting to disclosure of medical information.

Title XVIII, billing forms and completion instructions, provide for authorization by the beneficiary for the release of information to State agencies or their agents. Where the contractor, or a third party organization, is administering the medical payment program under a contract or underwriting arrangement with the State, it may release information to itself or to the third party organization on the same basis as it may release information to a State agency.

100.2 - Procedures for Supplying Title XVIII Information to State Agencies

(Rev. 1, 09-11-02)

The type of billing and payment information concerning individual beneficiaries that is needed by State agencies is determined by the nature of the State program and the extent to which it provides for payment of deductibles, coinsurance, and supplementary benefits. Subject to necessary payment of charges by the State in accordance with cost principles set by CMS, the contractor will arrange to release individual payment data such as copies, extracts, or summaries of Part A and Part B billing forms, explanation of benefits, denial letters, or equivalent special forms.

Because of the variations among State programs, contractors should arrange with the State Agency the procedures for release or use of information. Arrangements may include transfer of information on magnetic tape or other electronic means. For example, the contractor could match the updated State buy-in files or a current Medicaid master file or other files prepared by the State. The contractor

obtains approval for this procedure with appropriate cost allocation from the RO just as it must obtain RO approval for integrated processing. (See §90.1.)

100.3 - Disclosure to Title V and Title XIX Agencies of Information Indicating Unprofessional or Unethical Practices by Physicians and Other Practitioners

(Rev. 1, 09-11-02)

The release of information obtained in the administration of title XVIII indicating a course of unprofessional conduct or unethical practices by a physician or other practitioner will assist in the investigation of such conduct under titles V and XIX. The information to be released must have previously been released by a contractor in the administration of title XVIII to an official of a State licensing board (see §100.4 of this chapter), or to any officer, agency, establishment, or department of the Federal Government charged with the duty of conducting an investigation to determine whether there has been a violation of any provision of a Federal tax law (see §120 of this chapter).

100.4 - Disclosure to Title XIX Agencies of Physician Charge Information

(Rev. 1, 09-11-02)

CMS, which administers title XIX, requires a correlation between payments under that title and the fee schedule or the 75th percentile of the range of customary charges, if applicable, as determined for treatment rendered under Part B of title XVIII. This requirement assumes that title XVIII fee schedule and medical charge experience, where applicable, will be available to title XIX agencies.

Therefore, it is necessary that agencies administering title XIX be furnished with physician, other practitioner, and supplier identification numbers; and fee schedules and/or charges of physicians or other practitioners, or suppliers for services furnished to beneficiaries, to enable them to determine the amount of benefits payable for medical services furnished under Title XVIII and XIX.

As the individual title XIX programs of the several States may differ, some variation in the type of information required can be expected. Therefore, in order to obtain the information it needs, each State agency should consult with the title XVIII contractor(s) serving the various areas in each State concerning the availability and format of the necessary information. (CMS advises the appropriate State agencies of this arrangement.)

Contractor Involvement -

- The contractor will normally receive the request of the State title XIX agency through the RO.
- The contractor will determine the amount of information it can furnish in accordance with the particular request. (See §10ff of this chapter.)
- The contractor will determine the charge and discuss it with the State agency. (See Medicare Carrier Manual, Part 1, §4602.6.)
- If consultation or clarification of the data requested is needed, the contractor should make contact with the State agency through the RO or directly, as advised by the RO.
- When all details are completed, the contractor should submit the requested information in final form to the RO for furnishing to the State agency.

100.5 - Electronic Reporting of Crossover Claims

(Rev. 1, 09-11-02)

Contractors provide State agencies with a magnetic tape file of adjudicated claims for which Medicaid has some responsibility. The contractor provides this data in the record format shown on the CMS EDI web page at http://www.hcfa.gov/medicare/edi/edi3.htm.

If the State agency accepts crossover claims on magnetic tape, but requires additional data, or a different format, the contractor arranges to report these claims to the State agency. The State agency is required to absorb additional costs incurred by the contractor.

110 - Disclosure to Peer Review Organizations (PROs)

(Rev. 1, 09-11-02)

CMS and/or intermediaries may furnish PROs information on claims specified by CMS for PRO review.

120 - Disclosure to the Internal Revenue Service (Contractors)

(Rev. 1, 09-11-02)

The contractor may disclose information to the Internal Revenue Service when requested for investigation of a possible violation of the F.I.C.A., S.E.C.A., F.U.T.A., or any Federal income tax law. The contractor will bring to the attention of the RO any problem in complying with the IRS request (e.g., machine capability, cost, etc.).

130 - Disclosure to the Office for the Civilian Health and Medical Program of the Uniformed Services (TriCare) by Contractors

(Rev. 1, 09-11-02)

130.1 - General

(Rev. 1, 09-11-02)

CMS may disclose certain information to TriCare for use in administering the TriCare program. The type of information to be made available parallels that made available to non-complementary title V and XIX State agencies and contractors. (See §100 of this chapter.)

130.2 - Type of Information

(Rev. 1, 09-11-02)

We expect that most requests from TriCare for information will involve primarily Part B Medicare operations in such areas as identification of physicians and other medical practitioners who may be engaging in providing excess services or fraud. Before we release this information to a requesting TriCare contractor that is not also a Medicare contractor, a Medicare contractor must have previously released the information in the administration of Title XVIII to:

An official of a medical or other applicable professional society or

- An official of a State licensing board, or
- Any other officer, agency, establishment or department of the Federal Government charged with the duty of conducting an investigation to determine whether there has been a violation of any provision of Federal law (see §120 of this chapter).

We may also disclose to TriCare the types of data covered in §100.2 of this chapter, upon request when authorized by the beneficiary or the beneficiary's legal representative. In addition, we will, upon request, make available to TriCare Economic Index Data (as prescribed by §\$405.502(a) and 405.504(a) of Regulations No. 5) that might affect the limits on prevailing charge levels.

130.3 - Procedures for Release

(Rev. 1, 09-11-02)

TriCare fiscal agents desiring Medicare information will usually request it either through the TriCare headquarters in Denver or through the parallel RO. The TriCare headquarters will refer the request to the parallel RO. Should a Medicare contractor receive a request directly from TriCare, it will refer the request to the RO. The RO will meet with the TriCare fiscal agent and the Medicare contractor to determine exactly what is needed, impress upon the TriCare contractor that TriCare may not further disclose the information, and indicate the cost involved, if any. (Also note that payment may be estimated or waived if collection would interfere with efficient administration.) At the RO's discretion, the Medicare contractor and the TriCare fiscal agent may meet to discuss data and cost requirements without regional involvement. Upon agreement about data and cost requirements, the RO will authorize release of the information from the Medicare contractor to the TriCare fiscal agent.

140 - Disclosure of Information About Providers and Suppliers by CMS (Rev. 1, 09-11-02)

In keeping with the spirit of the Freedom of Information Act and growing consumer interest in health care facilities and suppliers, CMS has made available to the public various final reports and other information regarding providers and suppliers. Disclosure to the public by a contractor is limited, but the following guidelines indicate what information is available, to whom it may be released, and the source to which a requestor should be directed for information that the contractor is not authorized to release.

140.1 - Disclosure Necessary for Proper Administration of the Medicare Program (CMS) (Rev. 1, 09-11-02)

A - Disclosure of Survey Information of the Joint Commission on Accreditation for Hospitals (JCAH), American Osteopathic Association (AOA), or any other National Accreditation Organization

CMS may not disclose any accreditation survey made and released by the JCAH, AOA or other national accreditation organization except as prescribed by regulations. Accreditation letters and accompanying Recommendations and Comments prepared by the JCAH, AOA or other national accreditation organization for the provider's accreditation survey are confidential and exempted from public disclosure.

A copy of the provider's most recent accreditation survey may be disclosed with the provider's authorization to any authorized representative, employee or agent of CMS for official use only in connection with the Medicare sample validation or substantial allegation survey program.

B - Disclosure of Information About Hospitals, Skilled Nursing Facilities, Home Health Agencies, and Independent Laboratories

Information about a hospital, skilled nursing facility, or home health agency, as well as information about independent laboratories, providers of outpatient physical therapy (rehabilitation and public health agencies and clinics) and portable X-ray suppliers, may be disclosed without the authorization of the institution or organization when such disclosure is required for the proper performance of the duties of:

- An officer or employee of the Department;
- An officer or employee of a contractor; or
- An officer or employee of a State agency when necessary to carry out their duties under State law in the licensing or approving of hospitals, skilled nursing facilities, home health agencies, or independent laboratories, etc.

Information obtained in the provider certification process is not to be disclosed to those not included in the above three categories, except as indicated in §140.3 below.

However, information that a particular institution is participating in the program may be released; and information indicating the specialties for which title XVIII payment may be made for services of a particular independent laboratory may be released. This information is usually available from the Directory of Medical Providers and Suppliers of Services published by the Government Printing Office.

C. Contractors' Disclosure to Third Parties

Disclosure to third parties for program purposes of information about identifiable physicians and other suppliers who are natural persons is governed by the same restrictions and procedures as disclosure about named beneficiaries to third parties for program purposes. (See §50 of this chapter.)

Information about unidentifiable individuals or about providers or corporate entities (e.g., clinics or independent laboratories), is not exempt from disclosure by section 1106 and regulations at 42 CFR, Part 401. However, it is not to be disclosed by third parties to which information is disclosed for program purposes.

If the information the contractor discloses to third parties for program purposes is solely information that does not identify, or make identifiable, individuals (e.g., aggregate statistics or records on providers and other corporate entities) the following sample letter is to be used instead of the letter in §50B of this chapter.

Sample Letter from Contractor to Third Party

Dear

We are prepared to make available to your organization certain CMS records so that (state reason).

However, before making these records available to you, we must point out that by law all information derived in the administration of the Social Security Act is subject to the provisions of the Freedom of Information Act, which exempts from disclosure certain categories of records. In order to assure that no records held confidential under the Freedom of Information Act are disclosed, you must agree to meet the following conditions before we make any social security records available to you:

- 1. Any information we supply you will be used only for the specific purpose intended and for no other purpose.
- 2. All employees having access to this material will be instructed as to its confidential nature.
- 3. An official of your company will assume responsibility for ensuring that the information is not revealed to any other party.
- 4. You must return the material to us as soon as you have finished the job, and you may not retain copies.
- 5. We must have a letter from a responsible official in your office agreeing to these conditions and assuming responsibility for carrying them out.

Upon receipt of this letter from your organization, we will make available to you the CMS records you requested.

Sincerely yours,

140.2 - Disclosure of Medicare Reports

(Rev. 1, 09-11-02)

A. Provider Survey Report and Related Information

Information concerning survey reports of providers or facilities, as well as statements of deficiencies based on survey reports are available at the local social security office or the public health assistance office in the area where the facility is located. The following data may be released under this provision.

- The official Medicare survey report;
- Statement of deficiencies which have been conveyed to the provider following a survey;
- Plans of correction, and pertinent comments submitted by the provider relating to Medicare deficiencies cited following a survey.

B. Program Validation Review Reports and Other Formal Evaluation (CMS)

Upon written request, CMS makes available to the public official reports and other formal evaluations of the performance of providers. After CMS prepares the survey reports and other formal evaluations, it must provide the evaluated provider an opportunity (not to exceed 30 days) to review the report and submit comments on the accuracy of the findings and conclusions. CMS must incorporate the provider's pertinent comments in the report.

Generally, the RO serving the area in which the provider is located releases Program validation review reports. It will provide a copy of the report and the provider's comments to the contractors servicing the provider.

CMS may also make available to the public, subject to certain possible exemptions, informal reports and other evaluations of the performance of providers prepared by the contractor. The contractor refers requests for these reports to the Medicare regional office with a copy of the information requested.

C. Provider Cost Reports (Intermediaries)

1. Requests for Disclosure

Requests by the public either to inspect or to obtain a copy of a provider cost report must be in writing and must identify the provider or class of providers and specific cost reports requested.

NOTE: Personal salary information cannot be disclosed.

2. Ten Day Notice Requirement

Intermediaries must respond in writing within 10 working days after receipt of the written request, advising the requestor of the date the reports will be available. This date should be no earlier than 10 working days from the date of the contractor's response. (For exceptions, see Exceptions To 10-Day Notice Requirement immediately below.) A copy of this response must be sent simultaneously to the provider, thus putting it on notice that its report has been requested and by whom. (For exception, see §140.3 below.) If the request is for a report submitted by a former owner of a participating facility, copies of the contractor's response should go to both the present and former owners. If the request is for a report submitted by a provider no longer participating in the program, a copy of the contractor's response should be sent to the former provider. (For both former owners and former providers the copy of the response should be sent to the last known address of the party.)

If we cannot make the information available within a reasonable period of time, our response will include a brief explanation for the delay (e.g., because of the extent of searching, photocopying, or delay in securing reports from records retention centers). In addition to the provider's copy, the contractor sends a copy to the CMS regional office servicing the provider.

If a contractor receives a request for information concerning providers that it does not service, it will immediately forward such request to the CMS regional office servicing the provider and will inform the requester of its action. The regional office forwards the request to the servicing contractor.

3. Exceptions to 10 Day Notice Requirement

No 10-day delay in furnishing cost reports is necessary in the case of requests from:

• Federal or State agencies that need cost report information to carry out requirements of the Social Security Act (e.g., a State health planning agency under title XI (§1122), or a Medicaid State agency under title XIX).

The Congress - Requests from Congress are limited to those from the Congress as an official body (e.g., the Speaker of the House or the President of the Senate, or from the chairman of a committee or subcommittee of the House or Senate with jurisdiction related to the information

requested). Requests from individual members of the House or Senate are considered requests from the public.

4. Information That Can Be Disclosed

CMS's rules limit disclosure to cost report documents that providers are required by regulations and instructions to submit. In the case of a settled cost report, this includes the contractor's notice of program reimbursement. Cost report documents include the statistical page, the settlement pages, trial balance of expenses, cost finding schedules, balance sheet, statement of income and expenses, and other schedules or documents required as part of the regular cost report process. (Where a provider, after first obtaining program approval, submits equivalent documents in lieu of official program documents, such documents are subject to the same disclosure rules as apply to official forms.)

Information That May Not Be Disclosed

If a provider chooses to submit with its cost report additional information not specifically required by regulations or instructions, the contractor should not disclose such information unless it is contained within an official document or the equivalent thereof. For example, some providers may submit supplementary analyses of certain expenses, details of the professional component adjustment, financial statements (other than the statement of income and expenses and the balance sheet as required in accordance with cost reporting instructions), or income tax returns that are not required by the program. These would not be disclosable by the contractor.

Except where a provider has not submitted an acceptable cost report and supplements are required to complete the report, any additional documents or schedules that the contractor requires the provider to submit in support of its cost representations are also not to be disclosed. In addition, do not disclose audits, schedules, letters, notes, and comments; comments on results of desk reviews (including copies of the actual desk review documents); contractor notes and comments (including transmittal letters); audit adjustment summaries that contractors and auditors are required to prepare; and information pertaining to an individual patient. Contractors acknowledge such requests and refer them to the regional office.

NOTE: Cost report information that the contractor may not disclose may, nevertheless, be disclosed under the Freedom of Information Act by the CMS regional office, central office or, upon appeal of a denial, by the Administrator, CMS. Upon judicial review, a U. S. District Court may order Disclosure.

When an intermediary discloses a settled report, it may disclose schedules applicable to the settlement that have been reworked. The general rule is that if the contractor has reworked any of the schedules that the provider is required to submit with its original submission, these schedules become an integral part of the report for disclosure purposes. However, the intermediary may not disclose any details containing contractor or auditor comments concerning the settlement, details of specific adjustments, or supporting schedules applicable to the settlement of the provider's operation.

Prior to the release of cost report information to the requestor, the intermediary servicing the provider must screen each cost report and remove non-disclosable documents and information. If the intermediary is uncertain whether a particular document must be disclosed it should consult the regional office in the region in which the provider is located. Further, if the intermediary discovers as a result of review that the cost report reflects information that might result in adverse program publicity, it alerts the RO immediately. However, it does not delay disclosure.

The above instructions do not negate in any manner the requirement that the intermediary submit cost reports to CMS that contain full and complete information, including the details of all contractor and auditor adjustments, comments, and any supporting schedules that may be needed to verify the settlement.

5. Responding to Requests for Inspection of Cost Reports

The intermediary will make cost reports available for inspection by the requestor at the intermediary's office during regular business hours. In addition, arrangements will be made to make copies available upon specific request at any RO, or at central office. The intermediary must provide appropriate space for such inspections. It must establish procedures to ensure that all reports inspected are properly accounted for. Under no circumstances should it permit any requester to remove cost reports from the place of inspection.

If the requester has questions with respect to the interpretation or analysis of cost report information, the intermediary will advise the requester to submit such requests for more than routine explanations in writing to the regional office.

6. Disclosure of Cost Reports

If a request is received to inspect or to obtain a copy of a report that has not been settled; i.e., the final settlement notice of program reimbursement has not been sent, the intermediary will disclose a copy of the report as submitted by the provider. If settlement has been made, it will disclose the settled report. If a requester specifically asks for both the settled and unsettled cost reports of a provider, the intermediary will comply. When a report is made available for inspection or copying, it should be clearly marked with one of the following captions, as applicable.

Cost report as submitted:

- Settlement subject to audit; and
- Audited settlement.

Requests for reproduction of all or part of a provider's cost report may be subject to photocopy fees. If it is determined that a fee must be charged, the intermediary prepares the written response to the requester in accordance with §20 and §20.4 of this chapter.

7. Responding to Requests for Information

A requester may desire only selected cost report information of a provider or several providers. However, the DHHS Freedom of Information Regulation specifies that agencies are not required to create a record by compiling selected items from the files; such requests will be met by furnishing copies of specific documents which contain the information requested. In addition, when the contractor receives a request that requires selecting documents from various cost reports, the request should be honored. It will follow the written response procedure above even though the entire report is not being disclosed. It will advise the requester that a searching fee will be charged and give an approximate date when the documents will be ready.

D. Medicare Payment and Cost Data

The intermediary may release Medicare payment or Medicare cost data concerning a named provider without giving the provider the ten-day notice required in §10.2 or 30 above. However, it does not give

the requester enough information to calculate provider non-Medicare data as well. For example, intermediaries do not tell a requester that Medicare payments to the provider represent 30 percent of its total revenues.

The Medicare payment or Medicare cost data may be extracted from such documents as cost reports or Form CMS-3286, Monthly Actuarial Sample of Hospital Reimbursement. Any extraction of data is subject to the instructions above concerning creation of records.

E. Waiver of Liability Status

The waiver of liability status of a particular provider and the statistics used to determine that status may be disclosed.

140.3 - Disclosure of Information About Named Physicians and Other Suppliers of Service (RO's)

(Rev. 1, 09-11-02)

A. General

Generally, the contractor will not honor requests for information about a specific physician or other supplier of services, or for information which would identify such a physician or other supplier of services, except in situations described in §140.1 above. Disclosure to third parties for program purposes of information about individual physicians and other suppliers who are natural persons is governed by the same restrictions and procedures as disclosure about named beneficiaries to third parties for program purposes. (See §50B of this chapter.) The contractor acknowledges such requests and refers them to the RO for response.

If the information the contractor discloses is solely information that does not identify or make identifiable individuals (e.g., aggregate statistics or records on providers and other corporate entities), it uses the following language:

Dear:

We are prepared to make available to your organization certain social security records so that (state reason).

However, before making these records available to you, we must point out that by law all information derived in the administration of the Social Security Act is subject to the provisions of the Freedom of Information Act, which exempts from disclosure certain categories of records. In order to assure that no records held confidential under the Freedom of Information Act are disclosed, you must agree to meet the following conditions before we make any social security records available to you:

- 1. Any information which is turned over to you will be used only for the specific purpose intended and for no other purpose.
- 2. All employees having access to this material will be instructed as to its confidential nature.
- 3. An official of your company will assume responsibility for ensuring that the

information is not revealed to any other party.

- 4. The material must be returned to us as soon as you have finished the job.
- 5. We must have a letter from a responsible official in your office agreeing to these conditions and assuming responsibility for carrying them out.

Upon receipt of this letter from your organization, the social security records will be made available to you.

Sincerely yours,

If the contractor receives a written request for a physician's or supplier's customary charges or for amounts of program payments made to physicians outside of its jurisdiction, it acknowledges the request and sends it to the appropriate contractor, when known. If not known, it sends the request to the RO.

Disclosure of information about unidentifiable individuals, or about provider or corporate entities (e.g., clinics or independent laboratories) is not to be disclosed by third parties to which information is disclosed for program purposes.

B. Physician Fee and Supplier Charge Information

Customary charges may be disclosed to the public. However, requests from anyone (including a physician or supplier) for actual data used to determine the customary charge(s) should be referred to the RO.

Carriers may voluntarily specify a physician's or supplier's customary charges when explaining reimbursement or when furnishing denial or review notices.

The carrier may furnish a physician or supplier with a copy of their own customary charges free of charge; requests for customary charges from other sources will be subject to FOIA fees. However, anyone may inspect the customary charges free at the office of the carrier serving the locality for which the screens are used. Carriers produce a printout to be made available for inspection and/or photoopying in their office. If past experience with requests for customary charge screens from the public does not warrant the expense of producing a printout of all screens for inspection (i.e., there have been few request and no one has requested to inspect all the screens), printout copies of single screens or small numbers of screens may be furnished instead of photocopies; however, there will be no charge for the computer time for the printout nor shall the requesters be asked to pay more than they would if they were to request a photocopy of the screens. Requests specifically for printouts or tapes shall be furnished at cost to related Federal programs (for example, to TRICARE or Medicaid).

Carriers may release information to the public regarding the method used to determine Medicare allowances, e.g., that the median of the charges made by physician for a service is used as the customary charge, and that the 75th percentile of such customary charges in an area (weighted by frequency of service) is used to establish the prevailing charge.

Written requests for lists carriers do not have should be referred to the RO immediately by telephone so that RO can indicate whether the carrier should create a list. Such request might be for all the customary charge screens of internists in a given county or for a list of all physicians who charge a

given amount or less for a certain procedure. If requesters orally ask to see a list which does not exist, the carrier will offer them the opportunity to inspect the screens and create their own list.

C. Disclosure of Related Entrepreneurial Information

Information such as the specialty and business address of a physician or supplier may be disclosed to the public. The carrier will refer to the RO written requests for special lists, such as a list of five orthopedists in a given city. However, if the carrier has processed a list of all orthopedists in that city, it may, upon request, disclose the entire list. It may fulfill oral requests by offering the requester the opportunity to look at whatever disclosable records (e.g., customary charge screens) there are that contain the data (or to purchase photocopies of the data) or by referring the requester to the medical society or licensing board in the requester's area, if appropriate.

140.4 - Disclosure of Names of Providers, Physicians, and Suppliers of Services

(Rev. 1, 09-11-02)

140.4.1 - Names of Providers Terminated from Program Participation (Carriers)

(Rev. 1, 09-11-02)

Upon request, a carrier may disclose to the public the fact that a provider no longer participates in the program. It will refer to the RO inquiries as to why a provider no longer participates in the program.

140.4.2 - Names of Providers, Suppliers, and Physicians Found Guilty of Fraud (CMS) (Rev. 1, 09-11-02)

CMS publishes the names of providers, physicians, other suppliers of services, or any other persons who have been found guilty by a Federal court of submitting false claims in connection with title XVIII. Carriers will refer requests for such names to the RO for response.

140.4.3 - Names of Physicians on Second Opinion Lists (Contractors)

(Rev. 1, 09-11-02)

The DHHS Second Opinion Campaign has led to requests under the Freedom of Information Act for the list of physicians who have volunteered to accept referrals for second opinions. If a contractor has a list, it should disclose it upon request.

140.4.4 - Referral to State Licensing Boards, Medical Review Boards, and Professional Societies (Rev. 1, 09-11-02)

A. Referral of Suspended Practitioners

Section 1862(e)(2)(B) of the Social Security Act requires the Secretary to notify the appropriate State or local licensing authority (e.g., State licensing board or medical review board) whenever a physician or other practitioner has been suspended from participation in the Medicare program. Thus, whenever CMS suspends a practitioner from participation in the Medicare program because the practitioner has been convicted of a criminal offense related to participation in the title XVIII or XIX program, CMS

will promptly notify the appropriate State or local licensing authority(ies) to (1) make appropriate investigations, (2) invoke any sanctions available under State law which the authority(ies) deems appropriate, and (3) keep CMS and the Inspector General fully and currently informed of any action it takes.

B. Referral By Medicare Carriers

In addition to the referrals made by CMS under section 1862, Medicare carriers are authorized to refer title XVIII-related cases of apparent unethical practices or unprofessional conduct to medical or other professional societies, and State or local licensing authorities (licensing boards or medical review boards). (See the routine uses for the Medicare Carrier Claims Records system as described in the system notice published in the Federal Register. The web address for this information is: http://www.usdoj.gov/04foia/1974condis-3.htm.

When considering a case for referral, the carrier should assure itself that substantial basis for referral exists; that more than mere suspicion is involved. It need not compile evidence sufficient to prove misconduct before referral; it should ascertain the probability and severity of misconduct and leave further investigation, review, and disciplinary action to the appropriate society or board. Isolated instances of questionable practices or conduct should not normally be referred.

Further, referral of apparent unethical practices or a course of unprofessional conduct by a practitioner should be made only after proper professional advice has been obtained from the carrier's physician staff members, medical consultants, or other professional advisors.

Since State licensing boards and medical review boards are responsible for the licensing and sanctioning of practitioners, cases should be referred to those boards only where the apparent unethical practices or unprofessional conduct is of a severity to possibly warrant such sanctions; cases involving less severe improprieties should typically be referred only to professional or medical societies for action.

The following are examples of cases that should be referred:

- 1. Over-utilization This refers to a pattern of medical care which consists of providing more services than are medically necessary or which is not in accordance with acceptable medical practice (e.g., an inordinate number of office visits over an extended period of time for a chronic illness, a conspicuously high number of injections, excessive hospitalizations). Cases involving suspected over utilization should be reviewed by the carrier's physician staff. In cases where additional peer review is necessary, carriers may refer the case to a PRO. If the PRO servicing the area in question is not available to perform this type of review, the case should be reviewed by the medical society, medical consultants, or other professional advisors.
- 2. Mis-utilization This involves the rendering of services that are not medically acceptable according to the standards of the community concerned.
- 3. Overcharging This refers to the charging of fees by a physician, or other practitioner, that are not commensurate with the services rendered. Billing for amounts in addition to the deductible and coinsurance when assignment has been accepted also falls into this category.
- 4. Harmful Services or Pattern of Treatment This involves the furnishing of services or a pattern of treatment that is harmful to the patient, or of a quality which does not meet professionally-recognized standards of care.

- 5. Violation of Ethics These violations involve conduct of a physician or other practitioner that is contrary to the principles of ethics of the professional society to which the physician or other practitioner may belong, or which would make their practice a danger to the health and welfare of their patients or to the public.
- 6. Violation of the State's Professional Practice Statutes This involves the committing of acts that, under the applicable State law, would be grounds for suspension or revocation of the physician's or practitioner's license to practice.

These guidelines are not all inclusive; any activities by a physician, or other practitioner, in their treatment of program beneficiaries, which would warrant concern by professional societies or State or local licensing boards or medical review boards may be brought to their attention.

When a carrier refers a case to a professional society or State or local licensing board or medical review board because of apparent unethical practices or unprofessional conduct by practitioners furnishing services to beneficiaries, the PI staff in the regional office should be concurrently notified of the referral. Notification should include copies of all materials referred to the professional society or State or local board, and should be followed by reports of significant case developments. Since such cases may involve program abuse by the practitioner, the RO should be notified as quickly as possible to permit remedial or sanction action if deemed appropriate.

When a case is pending prosecution, or when a decision is pending on whether to proceed with prosecution, the contractor will delay referral to the professional society until: (1) the prosecution action is completed, (2) the decision is made not to prosecute, or (3) CMS authorizes the referral.

C. Requests for Assistance by State or Local Licensing Boards, or State or Local Medical Review Boards

While cooperation on the part of CMS ROs and carriers with State or local licensing/medical review boards is generally encouraged, there seems to be three distinct situations in which these boards might request assistance:

- 1. When a case has been referred to a State or local licensing/medical review board for investigation and possible sanctions (either by CMS as a result of the conviction and suspension of the subject practitioner, or by a carrier), it would seem appropriate for CMS and carriers to provide assistance to the board in its investigative activity, provided the demands on staff time and resources do not become burdensome or unreasonable. All requests for carrier assistance should, however be channeled through the servicing CMS RO for a determination regarding the reasonableness of the request. Appearance by CMS personnel before board meetings involving a case which has been previously referred would also be permitted.
- 2. When a State or local licensing board or medical review board requests information or assistance on a case which was self-initiated (i.e., not previously referred to the board by CMS or a Medicare carrier) as a result of a complaint, allegation, inquiry, etc., relative to a specific physician or practitioner's practices, this request should be treated as a Freedom of Information Act request. Therefore, any requested material should be screened for sensitive information, with the decision to release or withhold such information made on a case-by-case basis. Further, the board would be responsible for the costs involved in providing such information (searching costs, duplicating costs, etc.). In such instances, the appearance of CMS personnel at a board hearing would be discouraged.

3. When the State or local licensing board or medical review board's request is for general information pursuant to a study or investigation of physician or practitioner impropriety or abuse, and is not related to a complaint, allegation, etc., against a specific physician or practitioner; or when the request would represent a clearly unwarranted invasion of personal privacy, cooperation by CMS or contractor personnel would be discouraged. In such instances it would be appropriate to release general or statistical information that did not identify specific individuals; however, the request for identifying information would constitute an improper search for information.

140.4.5 – Disclosure of Medicare Statistics

(Rev. 1, 09-11-02)

Xref HHA 192.2, 192.3, SNF 192.2, 192.3, ESRD 126.2, 126.3, HSP 157.2, HO 192.2, 192.3

Numerous statistics on individual providers are available to the public. They include, but are not limited to, the following:

- 1. Waiver of liability statistics
- 2. Interim rate payment data
- 3. Amount of Medicare reimbursement
- 4. Overpayment data
- 5. Data from the Provider Monitor Listing
- 6. Information from the Directory of Medical Facilities and the Directory of Medicare Providers and Suppliers of Services
- 7. Medicare statistics (e.g., total visits, number of starts of care,)
- 8. Presumptive waiver of liability status
- 9. Information as to whether a provider participates in the Medicare program.
- 10. Medicare inpatient statistics for inpatient facilities (e.g., total inpatient days, number of admissions, average length of stay).

150 - Disclosure of Reports About Named Contractors

(Rev. 1, 09-11-02)

150.1 - Disclosure by ROs of Annual Reports of Contractors Performance (RCP)

(Rev. 1, 09-11-02)

RCPs are available to the public at the RO servicing the contractor. The contractor will refer requests for RCPs to the RO for response. It will refer requests for an RCP of a multi-State contractor to the RO servicing the home office of the contractor.

The contractor will refer requests for supporting documentation for the RCP contained in Contractor Performance Evaluation (CPE) reports to the RO.

150.2 - Disclosure of Medicare Audit Reports to the Press and Public (Audit Agency)

(Rev. 1, 09-11-02)

DHHS Audit Agency reports issued to Medicare contractors are available if requested, to members of the press and the public. The contractor will refer requests for DHHS Audit Agency reports directly to the Audit Agency, which is responsible for decisions regarding the release of these reports.

160 - Disclosure of Statistics (CMS or Contractors)

(Rev. 1, 09-11-02)

CMS and its agents may disclose statistical data and similar information that does not relate to any identifiable person or persons. However, they are not required to create records for requestors by compiling selected items from the files, nor are they required to create records to provide such data as ratios, proportions, percentages, per capitas, frequency distributions, trends, correlations or comparisons.

They may, however, create such records when efficient administration permits. Final authority to determine when "efficient administration permits" rests with CMS. The contractor will refer requests for information which can be created by hand to the RO unless the requestor is willing to accept copies of the pages containing the data needed to compile the requested information (i.e., the requester is willing to do the necessary compilation). The contractor will refer requests for information which must be created by machine (e.g., computer programs to manipulate data and print results) to the RO.

CMS and the contractors have a joint obligation to ensure that the program is clearly and accurately represented and that the public is given a complete picture of program performance.

There are certain problems in accurately representing program performance by means of statistical data alone, since some groupings of statistics are subject to inadvertent inaccuracies or are susceptible to misinterpretation. Careful judgment is, therefore, essential in presenting information on operating and payment activities.

160.1 - Information That the Contractor May Disclose

(Rev. 1, 09-11-02)

When considered appropriate, the contractor may release operating, payment, and cost data listed below without prior approval of the RO. The contractor will promptly forward to the RO a copy of information provided under this guideline.

In releasing any data under this guideline, the contractor must insure that the data are related specifically to the geographical area for which the contractor is responsible. For example, if a press inquiry is clearly for the purpose of presenting program operating data for a geographical area beyond the jurisdiction of the contractor, and CMS has not made such data available for public release, the contractor will tell the requestor that it does not have the authority to release information beyond its operational jurisdiction. It will refer the request to the RO. (See §160ff of this chapter.)

In releasing payment data, the contractor must indicate that the information refers only to the area it services and does not include payments to direct dealing providers, group practice prepayment plans, or railroad beneficiaries.

If it is not clear whether information can be released under this guide, the contractor will resolve doubt in favor of obtaining prior clearance from the RO.

The data released under this guideline should be based on the figures furnished to CMS in the monthly financial report and the monthly workload report. The administrative cost information should be the same as in the quarterly Operations Schedule-Cumulative Interim Expenditure Report.

The following data, relating only to the contractor's title XVIII workload, may be released:

- Benefit amounts paid by the contractor, including:
 - Aggregate amounts, or
 - Amounts broken out by category; e.g., inpatient hospital, outpatient hospital, etc., or
 - Amounts broken out by provider, including interim rates, and cost reports. (See also §140.1 of this chapter.)
- Total number of bills or claims paid by the contractor, either as an aggregate or broken out by category.
- Total number of bills processed.
- Admission notices, by category, i.e. inpatient hospital, skilled nursing facility, home health agency. (Extreme caution must be exercised to assure that this statistic is properly related to the items shown above.)
- Claims volume as an aggregate amount, or broken out by category.
- Median claims processing time, by category. Processing time is derived by subtracting date of receipt from date of approval for payment using the dates shown on bills.
- Any data determined from the data in items 1-6 above and related data available to the public (e.g., average payment per paid claims).
- Total administrative cost.
- Average (arithmetic mean) cost per bill or claim processed.
- Number or percentage of claims requiring contact with providers for further development.
- Contractor-completed CMS forms and reports listed chronologically by form number. Refer requests for CMS generated reports to RO.
 - Form SSA-1522 (Monthly Workload Report)
 - Form SSA-1523 (Estimate of Administrative Costs and Credits)
 - Form SSA-1566 (Intermediary Workload Report)
 - Form SSA-1566 Supplement (1972 Amendment Supplement to Intermediary Workload Report)

- Form SSA-1615 (Operations Schedule Final Administrative Cost Proposal)
- Form SSA-1822 (Monthly Provider Audit Activity Report-Schedule I)
- Form SSA-1822A (Final Summary of Audits Completed During Quarter-Schedule II)
- Form SSA-1822B (Costs Accrued During Quarter, Cumulative Costs by Contract and FY, Schedule III)
- Form SSAz-2580 (Cost Classification Report)
- Form CMS-2582 (Plan of Expenditures)
- Form CMS-2598 (Budget Distribution)
- Form CMS-3208 (Provider Audit Cost Report)
- Form CMS-1527 (Operations Schedule Cumulative Interim Expenditure Report and Budget)

Release of the last form should be accompanied by a qualifying statement such as the following:

"This report is based on preliminary data and is subject to adjustment by subsequent cumulative reports and a final cost proposal."

Absent prior RO clearance, the contractor may release only the data listed above. A request for data not included in the above list may, in the contractor's judgement, require an immediate response. If so, the contractor may furnish general comments but cannot provide statistics. No further action need be taken on these requests unless the requestor asks for additional information. When additional information is requested, the contractor refers the request to the RO.

160.2 - Information That May Be Released Upon Authorization by the Regional Office (Rev. 1, 09-11-02)

The CMS regional office may authorize contractors to release the following information:

- Total man-years (or man-quarters, etc.) employed per report period.
- Man-hours per claim or bill processed.
- Number of weeks' work on hand.
- Denial rates.
- Percent of claims or total payments involving reduction in charges.

Data which contract one contractor's performance with the performance of any other contractor, or which compare a contractor's performance with any set of national, regional, or other combined contractor performance data, may not be released by the contractor without prior RO approval. The contractor refers requests for such information to the RO. Generally, data of this type will be made available in periodic CMS releases when sufficient to permit valid statistical inferences. When the

release of this kind of information is considered valuable for contractor or program public relation, telephone request and approval for contractor release may be appropriate.

This section applies only to data compiled in the administration of the health insurance program. The contractor does not need prior RO approval to release, for example, compilations created by the Blue Cross Administration (BCA) or The National Association of Blue Shield Plans (NABSP).

Data pertaining to more than one contractor or which compare contractor performance may be released only in proper perspective. Essentially, cost information alone is not an accurate indication of a contractor's performance and, therefore, is not to be publicized without other relevant data accompanying it; e.g., information pertaining to workload volume and processing time. Moreover, release should always be accompanied by an interpretive statement indicating the limitations of the data and the known variables; e.g., wage differentials, etc.

Requests which include a provider population beyond the jurisdiction of the contractor may involve data which have not been made available by CMS for public release. Requests for such combined data will be handled by the RO, which, if it approves the release, will furnish the requestor with the appropriate combined data and alert the contractors involved regarding the release.

170 - Disclosure of Health Insurance Information by Providers

(Rev. 1, 09-11-02)

Records and information, acquired in the administration of the Medicare program, may be disclosed only under prescribed rules and regulations or under the authority of the Administrator of CMS. Information furnished specifically for purposes of a claim under the health insurance program is subject to these rules and regulations. These regulations apply to Governmental or private agencies that participate in program administration. These entities include the following:

- Institutions:
- Agencies;
- Person(s) providing services; and
- Providers of services.

The type of information includes, but is not limited to, the following:

- The individual's health insurance claim number (HICN);
- Facts regarding the individual's entitlement to health insurance benefits; and
- Medical and other information obtained from CMS, an intermediary, or a carrier (contractor).

Information not subject to these rules and regulations includes information in the provider's own records, such as the following:

- Name;
- Date of Birth:

- Sex:
- Marital status; and
- Address.

A provider's own records are, however, subject to requirements listed in the "Conditions of Participation", that "Patient's records be kept confidential (20 CFR Part 405.126). These records may also be subject to State or local laws governing disclosure.

Providers are also responsible for following conditions for coverage. A provider or supplier that receives a request for disclosure of information about a Medicare beneficiary, Medicare claim, or related information that it may not disclose, should refer the requestor to the appropriate contractor for further consideration of the request.

170.1 - Disclosure of Health Insurance Information to a Beneficiary or in Connection With a Claim

(Rev. 1, 09-11-02)

Information such as Medicare entitlement or eligibility data may be disclosed to a beneficiary or their authorized representative (this includes the beneficiary's representative payee).

170.2 - Disclosure to Contractors

(Rev. 1, 09-11-02)

Providers and suppliers may not forward medical information to contractors on a confidential basis, expressed, or implied, since, under the Privacy Act, any medical information obtained by a contractor is subject to disclosure to the individual to whom the information pertains or to another person authorized by the individual to have access to it.

Some providers and suppliers document findings on medical forms pre-printed "confidential" or routinely stamp all records "confidential" whether or not such records are ever intended for disclosure to a contractor. The contractor will accept such records only if the provider or supplier accompanies them with a signed statement indicating the following:

- That the provider or supplier understands the information is subject to disclosure to the patient under the Privacy Act, and
- That any words or statements that the transmitted records are confidential may be disregarded if the patient or the patient's representative requests them from the contractor or from CMS.

170.3 - Disclosure to Third parties for Proper Administration of the Health Insurance Program (Rev. 1, 09-11-02)

Disclosure by the provider or supplier to persons other than the individual or the individual's authorized representative of any records, reports, or other information about the individual is authorized without the individual's or their representative's consent under the following circumstances:

The disclosure is required in connection with any claim or other proceeding under the Social Security Act for the proper performance of the duties of:

- Any officer or employee of the Department; or
- Any officer or employee of a State agency, intermediary, provider of services, or other agency or
 organization participating in the administration of the program, by contract or agreement, in
 carrying out such contract or agreement.

These limitations apply whether or not the individual to whom the information pertains authorizes further disclosure to third parties (e.g., to a private medical plan).

170.4 - Disclosure to Third Parties for Other Than Program Purposes

(Rev. 1, 09-11-02)

Information obtained from CMS or its contractor is confidential and may be disclosed only under conditions prescribed in rules and regulations or on the express authorization of the Administrator of CMS. However, certain limited information about a beneficiary's Medicare eligibility status and related claims information may be released to third party payers with the beneficiary's express authorization

The following information may be released subject to necessary authorization:

- Beneficiary HICN;
- Coinsurance and deductible status;
- Dates of entitlement to Medicare:
- Copies of Medicare claims forms;
- Medicare report of eligibility; and
- Medicare Summary Notice (MSN).

Providers should refer requests for other information to the contractor. Contractors refer requests to the CMS regional office.

The provider or supplier will adhere to the following authorization guidelines to ensure that information is not released without the required authorization. Authorization must:

- Be in writing;
- Be signed and dated by the individual or someone authorized to act on the individual's behalf;
- Specify the name of the provider authorized to disclose information;
- Specify what information the individual is authorizing the provider to disclose;
- Specify the names of the third party payers to whom the information is being released;
- Specify the purpose for which the information is being released;

- Specify an expiration date for the authorization that should not exceed 2 years from the date it was signed; and
- Specify that it may be revoked at any time.

170.5 - Disclosure of Claims Payment Information in Alcohol and Drug Abuse Cases (Rev. 1, 09-11-02)

The law requires providers to observe more stringent rules when disclosing medical information for claims processing purposes from the records of alcohol and drug abuse patients. Since the standard consent statement on the provider billing form is not sufficient authority, under the law, to permit the provider to release information from the records of alcohol or drug abuse patients, more explicit consent statements are required.

Providers that participate in Medicare and alcohol and drug abuse prevention and treatment programs must obtain written consent in each alcohol or drug abuse case from beneficiaries to release medical information. This written consent, which allows the provider to disclose the records of the patient, must include all of the following:

- The name of the organization (e.g., hospital name) that is to make the disclosure;
- The name or title of the person or organization to which disclosure is to be made (e.g., CMS, including the appropriate intermediary or carrier, specified by name);
- The name of the patient;
- The purpose or need for the information to be disclosed (e.g., for processing a claim for Medicare payment and for such evaluation of the treatment program as is legally and administratively required in the overall conduct of the Medicare program);
- The specific extent or nature of information to be disclosed (e.g., all medical records regarding the beneficiary's treatment, hospitalization, and/or outpatient care including treatment for drug abuse or alcoholism);
- A statement that the beneficiary may revoke their consent at any time to prohibit disclosures on or after date or revocation;
- A statement specifying a date (not to exceed 2 years), event, or condition upon which consent expires without revocation;
- The date on which the consent is signed; and
- The signature of the patient or the signature of their authorized or legal representative.

If the beneficiary wishes, the consent statement may be expanded to permit disclosure by the provider to any other person, organization, or program (e.g., PRO), as appropriate. Providers may also give authorization to CMS and its contractors to re-disclose specific information to third party payers for complementary insurance purposes.

The provider keeps the consent statement with the patient's medical and other records.

The duration of the consent statement is not to exceed 2 years after which it must be renewed by the beneficiary if further disclosures are necessary.

170.6 - Disclosure of Itemized Statement to an Individual for Any Item or Service Provided

A. General

(Rev. 1, 09-11-02)

Section 4311 of the Balanced Budget Act of 1997 requires that if a Medicare beneficiary submits a written request to a health services provider for an itemized statement for any Medicare item or service provided to that beneficiary, the provider must furnish this statement within 30 days of the request. The law also states that a health services provider not furnishing this itemized statement may be subject to a civil monetary penalty of up to \$100 for each unfulfilled request. Since most institutional health practices have established an itemized billing system for internal accounting procedures as well as for billing other payers, the furnishing of an itemized statement should not pose any significant additional burden

B. 30-Day Period to Furnish Statement

The provider will furnish to the individual described above, or duly authorized representative, no later than 30 days after receipt of the request, an itemized statement describing each item or service provided to the individual requesting the itemized statement.

C. Suggested Contents of Itemized Statement

Although §4311 of the Balanced Budget Act of 1997 does not specify the contents of an itemized statement, suggestions for the types of information that might be helpful for a beneficiary to receive on any statement include: beneficiary name, date(s) of service, description of item or service furnished, number of units furnished, provider charges, and an internal reference or tracking number. If Medicare has adjudicated the claim, additional information the provider can include are: amounts paid by Medicare, beneficiary responsibility for co-insurance, and Medicare claim number. The statement should also include a name and telephone number for the beneficiary to call if there are further questions.

D. Penalty

A knowing failure to furnish the itemized statement shall be subject to a civil monetary penalty of up to \$100 for each such failure.

180 - Cost to a Provider That Requests Information Available to the Public

(Rev. 1, 09-11-02)

Providers are required to pay appropriate fees for information they request pertaining to other providers, Medicare contractors, or State agencies. A provider may claim such fees as allowable costs only if it demonstrates to the contractor the information is necessary in developing and maintaining the operations of patient care facilities and activities.

Exhibit A

CENTERS FOR MEDICARE & MEDICAID SERVICES FREEDOM OF INFORMATION ACT REQUEST

1. Case #:				
2. Date Received: 3. Due D	ate: 4. P	esponse Date:	5. Processir	ng Days:
6. Requester: 7. Affiliation/Address:				
8. Subject:			***	
9. Referred To:				
10. Category of Requester C	ommercial			
	ducational/Scientific of	or News Media		
C	other			
11. IS THERE PROGRAM CONCERN AE	BOUT DISCLOSING 1	HESE RECORDS? _	Yes	No
Ongoing Deliberations		nvasion of Privacy	Circ	cumvention of
Decision-making Process	S I	Pending Litigation	Agency Rules	
Proprietary Information		Open Investigation		
Other (Specify)				
12 ACTIONS: Direct Reply	No Re	ecorde Found		Dogwoot Withdrawn
12. ACTIONS: Direct Reply Not FOIA	Recor	ds Not Reasonably De	scribed ———	Request Withdrawn
Fee Related Clos	sure Refer	al to Next Review Levi	el	
				_
4.0-				
13. ACTUAL PROCESSING COSTS:	Hours	PONDING TO REQUI	EST Total	17. Invoiceable Fees
Reading/Interpreting/Logging	Hours	Hourry wage	iotai	
				XXXXXXXXXXXXXXX
Clarifying/Negotiating/Consultation				xxxxxxxxxxxxx
Searching for Records				\$
Review/Edit/Delete (DFOI Only)				\$
Compose/Type Response				XXXXXXXXXXXXXX
Other (specify)				\$
			·	
14. COPYING COSTS - @ \$.10 per page	: No. of Pages	No. of Sets	Total	
Pages Located/Copied		1 x \$.10 per page		XXXXXXXXXXXXXX
No. of Pages Released to Requester		1 x \$.10 per page	xxxxxxxxxxx	\$
No. of Pages Sent to Next Review Level		1	xxxxxxxxxxx	xxxxxxxxxxxxxx
15. MAILING COSTS: Postage				XXXXXXXXXXXXXX
Special Handling				
				<u> </u>
		16. Total Actual Cost:		
		10 To	tal Invaisantle Fe	
		16. 10	tal Invoiceable Fee	es:
19. Fees Charged:				
20. Fee Waived:				
21. Name(s), Phone Number(s) and Com	nonent(s) of Person	s) Who Searched For	and Compiled Th	acca Pasardo
and com	ponentaj or reison	S, THIO SCAICHEU FOR	and Compiled II	iese neculus:

See reverse side for instructions on completing this form. If you have questions, call the Division of Freedom of Information at (410) 786-5353. Form CMS-632-FOI (9/00) (Formerly HCFA-632-FOI)

Exhibit B

INTERMEDIARY/CARRIER/CENTRAL OFFICE COMPONENT

SUMMARY SHEET FOR THE CMS MONTHLY FOIA REPORTS

Data on this Summary Sheet is a summation of the data from all Forms CMS-632-FOI completed during the month.

INSTRUCTIONS

- 1. Enter the month and year that this summary sheet covers.
- 2. Enter, in the applicable space, appropriate identifying information concerning your unit.
- 3. Enter the total number of 632s completed during the identified month.
- 4. Enter the number of times the 632s showed a request was closed by the cited action.
- 5. Enter the total of all staff hours used during the identified month. Use only whole numbers and fractions or decimals, e.g., 102 1/2 or 102.25, not 102 hours and fifteen minutes.
- 6. Enter the total of actual cost figures, based on the salaries of the staff involved.
- 7. Enter the total of copying charges.
- 8. Enter the mailing/postage costs.
- 9. Enter the total dollar amount of fees charged. This figure is not the same as actual costs.
- 10. Enter the total of all fees waived. (Unless the FOI Office, CMS has granted a waiver, this figure is a compilation of all fees which were waived because they were lower than CMS' \$15.00 threshold).
- 11. Enter the name, office and address of the person who completed this summary sheet.
- 12. Enter the date the summary sheet was completed.

PLEASE FILL OUT COMPLETELY

1.	Month:	Year:		
2.	INTERMED	IARY CARRIER:		
	CENTRAL (OFFICE COMPONENT:		
3.	Total 632s: _			
4.	Actions:	Direct Reply (Records Sent)	No Records Found	
	-	Request Withdrawn	Not FOIA	
	-	Records Not Reasonably Described	Subpoena Denial	
	-	Fee Related Closure	Other	
5.	Staff Hours:			
6.	Staff Charge:	s:		
7.	Copy Charge	es:		
8.	Postage:			
9.	. Fees Charged:			
10.	Fees Waived	: <u></u>		
11.	Name/Phone	:		
	Office:			
12.	Date:			

☆ U.S. GOVERNMENT PRINTING OFFICE 2001: 671-118

	INVOICE OF FE	ES FOR FOIA S			
CA	SE NUMBER	DATE			
IA7	FERIAL REQUESTED				
	NAME OF REQUESTOR				
2	ORGANIZATION				
CHARGE	STREET ADDRESS				
_	CITY	STATE	ZIP CODE		
		NUMBER	CHARGE		
₹E	PRODUCTION				
	EACH PAGE 10 ¢				
_	OTHER (e.g. COMPUTER PRINTOUT)				
SE	ARCH FEES; Per hour (Based on	Salary of Searcher as p	per 45 CFR 5.43)		
	Level 1				
	Level 2				
	Level 3				
RE	VIEW FEES; Per hour (Based on	Salary of Reviewer as p	er 45 CFR 5.43)		
	Level 1				
	Level 2				
	Level 3				
SP	ECIAL SERVICES;				
	CERTIFICATION (\$10.00)				
	RETURN RECEIPT (\$1.35)				
	OTHER				
		PAY TOTAL OF	\$		
Q	uestions regarding enclosed ma	aterial or charges, call:			
N.A.	AKE CHECK OR MONEY ORDER EDICAID SERVICES AND REMIT	PAYABLE TO: CENTERS WITH A COPY OF THIS I MEDICARE & MEDICAID	NVOICE TO:		
_	P.O. BOX 7520 BALTIMORE, MI				
*	PLEASE INCLUDE THE CASE NUMBI	ER ON YOUR CHECK OR I	NONEY ORDER		
_	nclosed is payment of \$	by check	money order		